ED 294 604 JC 870 513

TITLE Agreement between the Massachusetts Board of Regents

of Higher Education for the Massachusetts Regional Community Colleges and the Massachusetts Community College Council/Massachusettes Teachers Association, an Affiliate of the National Education Association for Academic Years 1983-84 to 1985-86 and 1986-87 to

1988-89.

INSTITUTION Massachusetts State Board of Regents of Higher

Education, Boston.; National Education Association,

Washington, D.C.

PUB DATE [86] NOTE 140p.

PUB TYPE Legal/Legislative/Regulatory Materials (090)

EDRS PRICE MF01/PC06 Plus Postage.

DESCRIPTORS \*Collective Bargaining: \*College Faculty: Community

Colleges; \*Contracts; Department Heads; \*Employment Practices; Faculty Promotion; Faculty Workload; Fringe Benefits; Grievance Procedures; Job Layoff; Leaves of Absence; Personnel Policy; Released Time;

Teacher Dismissal; Teacher Retirement; Teacher Salaries; Tenure; Two Year Colleges; Unions

IDENTIFIERS \*Massachusetts; Massachusetts Board of Regional

Community Colleges; National Education Association;

\*NEA Contracts

#### **ABSTRACT**

Two consecutive collective bargaining agreements between the Massachusetts Board of Regents of Higher Education and the Massachusetts Community College Council/Massachusetts Teachers Association are presented, covering the years 1983 through 1986 and 1986 through 1989. The 26 articles in the agreements set forth rights and provisions related to: (1) unit recognition and definitions; (2) the relationship between the association and the board, and the formation of a special joint study committee to resolve problems related to the agreement; (3) the use of board facilities; (4) the rights and responsibilities of the board, and the role of unit members in college governance; (5) maintenance of records; (6) deduction of dues and agency fees; (7) academic freedom and responsibility; (8) affirmative action; (9) supplemental benefits; (10) grievance procedures; (11) appointment, reappointment, and tenure; (12) workload, work assignment, and working conditions; (13) faculty evaluation; (14) promotion and change of rank of faculty and professional staff; (15) termination, dismissal, discipline, and resignation; (16) filling of vacancies; (17) transfer; (18) written notices; (19) retrenchment; (20) department chairs; (21) salary adjustments; (22) the establishment of the Management Association Committee on Employee Relations; (23) no strike or lockout pledge; (24) a savings clause; (25) holdover; and (26) the duration and successorship of the co tract. Forms for applying for sabbaticals, complaints, workload computation, instructional evaluation, student advisement, promotion and professional development are appended to the 1986-89 agreement. (MDB)



# AGREEMENT

# BETWEEN

The Massachusetts Board of Regents
of Higher Education
For The
Massachusetts Regional Community Colleges

# AND

The Massachusetts Community College Council/
Massachusetts Teachers Association
An Affiliate Of
The National Education Association

JULY 1, 1983 through June 30, 1986

July !, 1986 through June 30, 1989

"DEDMICCION TO DESCRIPTION
"PERMISSION TO REPRODUCE THIS
MATERIAL HAS BEEN GRANTED BY
MAICHIAL HAS BEEN GRANTED BY

S. Edgar

TO THE EDUCATIONAL HESOURCES INFORMATION CENTER (ERIC) "

# IJ S DEPARTMENT OF EDUCATION Office of Educations's Research and Improvement

EDUCATIONAL RESOURCES INFORMATION CENTER (ERIC)

- This document has been reproduced as received from the person or organization originating it.
- Minor changes have been made to improve reproduction quality
- Points of view or opinions stated in this document do not necessarily represent official OERI position or policy



# **AGREEMENT**

# **BETWEEN**

The Massachusetts Board of Regents
of Higher Education
For The
Massachusetts Regional Community Colleges

# AND

The Massachusetts Community College Council/
Massachusetts Teachers Association
An Affiliate Of
The National Education Association

For Academic Years

1983-1984 • 1984-1985 • 1985-1986

July 1, 1983 through June 30, 1986



# TABLE OF CONTENTS

Negotiating Teams' Page		4	ARTICLE IX	
Preamble		5	Supplemental Benefits	1
			9 01 Authorized Leaves with Pay 1	1
ARTICLE I			9 02 Unpaid Leaves of Absence 1	7
Recogn	nition and Definitions	5	9 03 Association Leave 19	9
1.01	Recognition	5	9.04 Insurance and Other Benefits/	
1.02	Definitions	5	Health and Welfare Trust 1	9
1.03	New Positions Added	6		
1.04	Jurisdiction	6	ARTICLE X	
	Appendix A	7	Grievance Procedure 2	1
ARTIC	CLE II		ARTICLE XI	
Relatio	enship Between		Appointment and	
Associ	ation and Board	7	Reappointment - Tenure 2	4
2.01	Fair Practices	7		
2.02	Complaints	7	ARTICLE XII	
2.03	Safety	7	Workload, Work Assignment	_
2.04	Individual Contracts	7	and Working Conditions	
2.05	Association Rrepresentatives	7	12.01 General Provision 2	
2.06	Relevant Information	7	12.02 Work Assignment 2	
			12.03 Workload of Faculty 2	
ARTIC	CLE II-A		12.04 Workload of Professional Staff 2	
Specia	Joint Study Committee	7	12.05 Worklend Reduction 2	
•	•		12.06 Workload Grievance Panel 2	9
	CLE III		ABTICLE VIII	
Use of	Board Facilities	8	ARTICLE XIII	^
3.01		8		0
3.02	Bulletin Boards	8	•	0
3.03		8		0
3.04	Unit Facilities and Services	8	13.03 Evaluation of Professional	_
3.05	Access to College Facilities	8	Strff Members	0
ARTIC	CLE IV		•	0
	ights and Responsibilities		13.05 Written Reasons	0
	Board	8		
			ARTICLE XIV	
ARTIC	CLE IV-A		Criteria and Process for <del>Promotion</del>	
Colleg	e Governance	9	of Faculty and Professional	,,
				33
ARTIC	CLE V			33
Maint	enance of Records	9		33
			14.03 Announcement of	33
ARTI	CLE VI			35
	tion of Dues and Agency Fees	9		35
6.01	Dues Deductions	9	14.05 140n-State Appropriated Funding	, .
6.02	Agency Service Fee	10	ADTICLE 30/	
vive regulary dervice tee		••	ARTICLE XV	
ARTICLE VII			Termination, Dismissal,	3:
ARTICLE VII Academic Freedom & Responsibility		10	• • •	
ACRUE	mic Freedom of Responsionity	10		3
ABT	CI E 1911			3(
ARTICLE VIII Affirmative Action		11		3( 3(
<b>₩</b>	Buve Action	11	15.04 Resignation	) د



ARTI	CLE XVI		20.07 Procedure for the Selection of	
Filling	of Vacancies in the		Department Chairperson	40
Bargai	ning Unit	36	20.08 Evaluation of Department	70
			Chairperson	40
	CLE XVII		20.09 Recall	40
Transf	er	37	20.10 Workload Reduction	40
			20.11 Compensation	40
ARTIC	CLE XVIII			
Notice	<b>8</b>	37	ARTICLE XXI	
			Salary Adjustments	42
ARTIC	CLE XIX			
Retren	chment	37	ARTICLE XXII	
19.01	Basis for Retrenchment	37	Management Association Committee	
19.02	Consultation	38	on Employee Relations	45
19.03	Reassignment	38	22 01 College Level	45
19.04	Retrenchment Procedure	39	De of Conege Devel	43
19.05	Seniority	39	ARTICLE XXIII	
19.06	Lay-Off Status	39	No Strike or Lock Out Pledge	48
19.07	Recall	39	No Suike of Lock Out Fledge	45
19.08	Retraining	39	ARTOLD SOUN	
			ARTICLE XXIV	
ARTIC	CLE XX		Savings Clause	45
	ment Chairpersons	39		
20.01	Department Chairpersons/Curriculum	• ,	ARTICLE XXV	
20.0.	Coordinators/Program		Holdover	46
	Coordinators	39		
20.02	Appointment	40	ARTICLE XXVI	
20.03	Withdrawal from Duties	40	Duration and Successorship	46
20.04	Jurisdiction	40		
20.05	Department Chairperson/Curriculum	40	Signature Page	47
-4.45	Coordinator (Work Area)	40	Calogo Cabadalaa	•
20.06	Department Chairperson/Curriculum	70	Salary Schedules	X
	Coordinator (Program)	40	Forme	x
	(Liopeni)	<del></del> -	Forms	Λ.



# NEGOTIATING TEAM PAGE

# For the Board

# CARLETON H. LA PORTE

Vice Chancellor & Director For Employee Relations Chief Negotiator

JUDITH A. WONG Community College Counsel

CAROLYN R. YOUNG
Assistant Community College Counsel

SHARON Y. LOWE

Contract Administrator/Hearing Officer

Board of Regents

# PATRICIA CHISHOLM

Dean of Student Services
Bunker Hill Community College

# **KEVIN KIERNAN**

Dean of Administrative Services
Quinsigamond Community College

# ROBERT MC DONALD

Dean of Academic Affairs
Northern Essex Community College

# CARL SCHILLING

Dean of Academic Affairs
Middlesex Community College

# For the Association

R. MICHAEL MC SWEENEY
Chairperson, Bunker Hill Community College

HELEN GOOLISHIAN
Vice Chairperson

Cape Cod Community College

**DENNIS FITZGERALD** 

Team Secretary

Massasoit Community College

LEO ARNFELD

Bristol Community College

JACK ARONSON

MCCC Board of Directors
Northern Essex Community College

JAMES RICE

MCCC President
Quinsigamond Community College

THOMAS PARSONS

MCCC Vice President

JACK CARPENTER

MTA Consultant

JON BUTLER

Research Coordinator, Resource Consultant North Shore Community College

RAYMOND LEMIEUX

MCCC Treasurer, Resource Consultant Springfield Technical Community College

#### **PREAMBLE**

This Agreement is entered into by and between the Board of Regents of Higher Education (hereinafter the Board), and the Massachusetts Teachers Association/Massachusetts Community College Council (hereinafter the Association) as the exclusive bargaining representative for all employees in the bargaining unit described in Article I. Both parties to this Agreement recognize the unique contributions of the community colleges to education in this Commonwealth. Further, the parties recognize the need to strengthen and secure community college education. This Agreement has as its purpose the promotion of harmonious relations between the Board and the Association. The parties declare their commitment to maintaining and improving the quality of educational services offered by the Community Colleges. To this end the parties recognize their statutory obligations, pursuant to the provisions of General Laws. Chapter 150E, and the rules and regulations promulgated thereunder, to negotiate in good faith with respect to wages, hours, standards of productivity and performance, and other terms and conditions of employment.

Both parties recognize and declare that providing quality higher education and services to the citizenry of the Commonwealth with the broadest accessibility is their mutual goal. The education of our students is our primary purpose.

In recognition of these obligations it is hereby agreed as follows:

# ARTICLE I - RECOGNITION AND DEFINITIONS

#### 1.01 Recognitio.

The Board hereby recognizes the Association as the exclusive bargaining representative with respect to conditions of employment for all regular full-time employees occupying the positions delineated in Appendix A.

The Board agrees to apply the applicable provisions of this Agreement to those employees whose source remuneration is derived from non-state appropriated funds and who perform the functions of any of those positions delineated in Appendix A, to the extent that the terms of their respective grants or non-state appropriated funding source and the level of funding thereunder allow, as determined by the President of the College or his/her designee. It is understood that the following Articles of this Agreement shall not apply to those individuals occupying said positions, except as otherwise provided in this Agreement:

Article XI Appointment and Reappointment

Article XIX Retrenchment

#### 1.02 Definitions

Academic Year – a period of time encompassing two (2) semesters, beginning no earlier than September 1 and ending no later than May 31, exclusive of Commencement.

Administration – all professional employees of the Board who are not members of the bargaining unit as described in Appendix A.

Administrator - a non-unit professional employee.

Association – the Massachusetts Community College Council/Massachusetts Teachers Association, an affiliate of the National Education Association.

Association Representative—a member of the Association who has been designated as a representative by the Association President in writing to the President of the College; also, a representative of the Massachusetts Teachers Association or the National Education Association as designated by the Association President to the President of the College.

Board—the Board of Regents of Higher Education or any College Board of Trustees as defined in General Laws, Chapter 15A, whichever the case may be as provided in Article XXVI.

Chancellor—the Chancellor of the Board or an individual acting in that capacity as duly appointed by the Board, in accordance with General Laws, Chapter 15A.

Cause-shall mean just cause.

College President – the President of a Community College or an individual acting in that capacity as duly appointed by the Board in accordance with General Laws, Chapter 15A.



Colleges—all facilities and properties of \*Community College now or hereinafter established by the Board pursuant to General Laws, Chapter 15A.

Complaint—a written statement setting forth a grievance as hereinafter defined, which includes a statement of all the known facts pertaining to the alleged breach on which the grievance is based, including but not limited to the date(s) when the breach allegedly occurred, the specific contractual provisions alleged to have been breached, the remedy requested.

Curriculum Coordinator – a unit member who functions in a similar manner as a department chairperson but is responsible for a smaller work area.

Department Chairperson (Work Area) - a unit member performing the duties delineated in Article 20.05.

Department Chairperson (Program) – a unit member performing the duties delineated in Article 20.06 and who may be assigned responsibility to assist in the implementation of a College program and/or curriculum, defined as a series of courses customarily leading to a certificate or associate degree and which accreditation, licensure or other external governing agencies require a member defined within the unit to perform supervisory functions for approval, maintenance and continuance of the program.

Dismissal- the discharge of any unit member for just cause prior to the expiration of his/her appointment.

Faculty-unit members occupying full-time positions as instructor, assistant professor, associate professor or professor.

Grievance—an allegation by a unit member(s) or by the Association that a specific provision of the Agreement has been breached in its application to him/her or it.

Grievant-a unit member(s) or the Association who seeks a resolution of a grievance pursuant to Article X.

Lay Off Status-the status of a unit member who has been retrenched pursuant to Article XIX.

Professional Judgment—every decision to renew or fail to renew a professional appointment, to terminate any such appointment, to grant or refuse to grant academic tenure, promotion, professional leave, sabbatical leave, educational needs/professional/development awards, performance-based salary adjustments, and evaluation results shall be deemed to have been made pursuant to an exercise of professional judgment; and every grievance that, explicitly or by implication, questions the merits of any such decision, but of no other decision, shall be deemed to be a grievance that questions an exercise of professional judgment. In matters of professional judgment, said decision is limited by the specific terms of this Agreement.

Professional Staff Member - a unit member whose primary duties are other than teaching.

Promotion—an appointment of a faculty member to the next higher faculty rank; or appointment of a professional staff member to the next higher classification.

Recall—the removal for cause of a person from his/her position as department chairperson but not the separation of said person from the service of the College.

Retrenchment—the discontinued employment of a unit member prior to the expiration of his/her term of appointment through no fault or delinquency on his/her behalf, pursuant to Article XIX.

Seniority—continuous full-time service by a unit member at his/her College, which is deemed to commence with the unit member's effective date of employment as a unit member.

Temporary Employee—one employed as a unit member to substitute for a unit member holding a regular, multiple year, or tenured appointment for a period of one (1) year or less for the purpose of replacing a unit member on leave or whose employment ended prior to the completion of the fiscal or academic year.

Tenure—an appointment which vests a right to continued employment at a College without limit of time subject to dismissal for just cause in accordance with Article XV and retrenchment in accordance with Article XIX.

Termination—the non-renewal of a contract of a non-tenured unit member.

Transfer—the appointment by the Board of a unit member to a faculty or professional staff position at the same rank or classification in the same or another College in the Community College System.

Unit Member-an employee of the Board occupy a position as defined in Article I and as delineated in Appendix A.

Work Area-those specific work areas designated by the administration as work areas.

Vacancy-a position for which funding is available and which the administration intends to fill.

# 1.03 New Positions Added

The Board recognizes its obligation under the provisions of General Laws, Chapter 150E, to bargain in good faith over the addition of any professional positions within the bargaining unit in excess of the total number of existing bargaining unit positions at a College.

#### 1.04 Jurisdiction

During the term of this Agreement, the Board agrees not to negotiate with any unit member(s) or with any employee organization other than the Association.



#### APPENDIX A

The bargaining unit contains the following positions as described in Article 1.01:

Position	Pavroll Title Code	Position Payroll	Title Code
Instructor	90-156	Coordinator of Student Activities	90-754
Assistant Professor	90-155	Placement Officer 90-3	75, 90-717
Associate Professor	90-154	Health Care Counselor	90-771
Professor	90-153	Counselor	90-712
Librarian (in those cases Librarian is under the the Director of the Lea Center or its equivalen	supervision of rning Resource (1) 90-702	AV/TV Coordinator Director of Instructional Media (in those cases where the Director is under the supervision of the Direct	
Assistant Librarian Cataloguer	90-703 90-724	of the Learning Resource Center of the equivalent)	90-772
C=10800.	30-124	Director of Cooperative Education	90-766

# ARTICLE II - RELATIONSHIP BETWEEN ASSOCIATION & BOARD

#### 2.01 Fair Practices

The Board and the Association recognize and affirm their commitment to the policy of non-discrimination with regard to race, color, religious creed, national origin, sex, sexual preference, age, marital status, family relationship and handicap status.

#### 2.02 Complaints

The administration shall within fourteen (14) calendar days send or communicate to a unit member any written complaint or material which the administration believes would adversely affect his/her employment status.

## 2.03 Safety

In accordance with applicable state or federal law, unit members shall not be required to work under unsafe conditions whenever such conditions have been brought to the attention of the President of the College or his/her designee by the unit member(s) and the College has failed to exercise reasonable efforts to redress the complaint.

#### 2.04 Individual Contracts

All rights, benefits, duties and obligations of unit members as set forth in the Agreement shall during its term be expressly incorporated by reference into and made part of any contract of employment that has been or shall be entered into between the Board and a unit member and no such contract shall be contrary, in whole or in part, to the terms and conditions as set forth herein.

## 2.05 Association Representatives

The President of the Association shall furnish the Chancellor or his/her designee with a written list by September 15 of each year of officers and representatives of the Association and their terms of office. It is clearly understood that in the implementation of this section, there shall not be more than one (1) Association representative from each College; provided, however, that if a designated representative is unavailable, the President of the Association shall notify the Chancellor or his/her designee as to who shall be that representative's alternate.

#### 2.06 Relevant Information

The Board shall provide to the Association in accordance with Chapter 150E, such information as is necessary for the proper discharge of its duties as the exclusive bargaining agent. Such information shall include, but not be limited to, a copy of the payroll at each College, the number of vacant and filled positions at each College, and the enrollment figures by program at each College; provided, however, that such information shall be provided to the local chapter president by October 15 of each year; provided further that this shall not require the College to compile such information in the form requested unless already compiled in that form.

# ARTICLE II-A - SPECIAL JOINT STUDY COMMITTEE

There shall be established under this Agreement a Special Joint Study Committee according to the following provisions:

- 2A.01 There shall be appointed eight (8) members: the Board of Regents of Higher Education shall appoint three (3) Community College Presidents and the Vice Chancellor and Director of Employee Relations, or in his absence, the Associate Vice Chancellor of Employee Relations; the President of the Massachusetts Community College Council shall appoint three (3) officers of the Council and a representative or officer of the Massachusetts Teachers Association.
- 2A.02 The Committee shall meet no less than six (6) times per year, unless mutually agreed otherwise.



- 2A.03 The Committee may ask a neutral to join them on occasion as a useful catalyst in creating a problem-solving atmosphere.
- 2A.04 The Committee shall make rules of order within which to operate.
- 2A.05 All fees and expenses of the Committee (including those necessary to retain a neutral) shall be divided equally between the parties to this Agreement.

The Committee during the life of this Agreement shall:

- 1. Resolve matters of mutual concern between the parties to this Agreement.
- 2. Address issues which require in-depth study and discussion.
- 3. Resolve contract issues referred to it in the recent fact-finder's report and recommendations.
- 4. Explore and identify root causes of current problems between the parties with a view of resolving all such problems.
- 5. Make recommendations to the parties to this Collective Bargaining Agreement during the life of the new contract as recommended by the Fact-Finder.

All reports, findings and recommendations of the Special Study Committee shall be expeditiously given to the Chancellor of the Board of Regents of Higher Education and the President of the Massachusetts Community College Council.

# ARTICLE III - USE OF BOARD FACILITIES

# 3.01 Facility Use

Upon request in writing made to the President of a College or to his/her designee, the Association or any College-based chapter thereof shall have the right to nieet at such College, if appropriate facilities are available. All requests must be received at least twenty-four (24) hours prior to the time requested for the meeting.

#### 3.02 Bulletin Boards

The President of each College or his/her designee shall assign at least two (2) bulletin boards for the exclusive use of the Association for the purpose of posting Association notices concerning the administration of the provisions of this Agreement. All notices so posted shall bear the signature of the President of the Association or his/her designee and shall indicate the date on which they are posted.

#### 3.03 Intra-College Mail

The Association shall be permitted to use the intra-college mail system for the distribution of Association communications. All notices so placed shall bear the signature of the President of the Association.

#### 3.04 Unit Facilities and Services

The Board shall make reasonable efforts to maintain at each College:

- 1. Office space currently being used or office space that may become available for the Association for on-campus contract administration;
- 2. Existing furnished employee lounges, restroom and eating facilities;
- 3. Existing assigned space and necessary equipment to carry out their assigned duties;
- 4. Existing parking facilities;
- 5. Existing telephones where currently operating;
- 6. Existing custodial, secretarial and technical assistance.

The President of the College or his/her designee shall consult with the Chapter President of the Association or his/her designee prior to altering the level of existing facilities or services listed herein and currently in use by unit members.

# 3.05 Access to College Facilities

The Board agrees to make available to unit member access to their work areas and offices on the same basis and to the same extent as such facilities are made available to other College employees, subject, however, to applicable Board policy, rules and regulations governing access to College facilities.

# ARTICLE IV - THE RIGHTS & RESPONSIBILITIES OF THE BOARD

- 4.01 All management's rights and functions, except those which are clearly and explicitly abridged by the specific terms of this Agreement, shall remain vested within the Board. These exclusive rights include, but are not limited to the following:
  - 1. To the executive management and administrative control of its Colleges and their properties and facilities;
  - 2. To hire all employees, to determine their qualifications and the conditions for their continued employment or their dismissal or demotion, and to promote and transfer all such employees;



- 3. To establish courses of instruction, including special programs and to provide for athletic, repreational and social events for students, all as deemed necessary or advisable by the Board.
- 4. To decide upon the means and methods of instruction; the duties, responsibilities, and assignment of teachers and other employees with respect to administrative and non-teaching activities, and the terms and conditions of employment;
- 5. To establish the standards of productivity of its employees; and
- 6. To establish policies, rules and regulations and practices in carrying out its responsibilities.
- 4.02 The exercise of the foregoing powers, rights, authority, duties and responsibilities by the Board, the adoption of policies, rules, regulations and practices in furtherance thereof, and the use of judgment and discretion in connection therewith shall be limited only to the specific and express terms of this Agreement, and then only to the extent such specific and express terms hereof are in conformance with the Constitution and laws of the United States. Nothing contained in this Agreement shall be deemed or construed to impair or limit the powers and duties of the Board under the laws of the Commonwealth.
- 4.03 It is understood that the matters contained in this Article are not subject to the grie rance and arbitration procedures in this Agreement, except as to the limitation stated in this Agreement or unless it can be shown that in the exercise of these rights the Board acted unreasonably and to the detriment of employee rights.

# ARTICLE IV-A - COLLEGE GOVERNANCE

- 4.01 The Board recognizes the importance of the advisory role of unit members in matters of College governance, including the improvement and development of academic programs and resources. The Board and the Association recognize that advisory organizations currently exist within the Community College System and that the structures and procedures established by such bodies vary among the Colleges in response to differing conditions, interests and needs of each College. Such advisory organizations or similar organizations shall be maintained or created at each College, after consultation with the Association, to insure advisory comment from unit members and other constituencies of the College. A governance structure shall provide for an open forum for discussion and information sharing for the purpose of providing the President of the College with advisory input prior to the promulgation of College policy.
- 4.02 The Board recognizes the importance of the role of unit members in the selection process of unit members; provided, however, that unit members shall also participate in the selection process of administrators where they are currently allowed to do so pursuant to an established and continuous college-wide past practice or written College rule; provided further that the President of the College shall select unit members to serve on selection committees for these purposes after first consulting with the members of the appropriate division/department/work area.

# ARTICLE V - MAINTENANCE OF RECORDS

5.01 Each Community College shall maintain an official personnel file for each unit member, which shall be the personnel file consulted when making all personnel decisions and recommendations. Any and all material contained in this personnel file concerning the unit member shall be open to the unit member with a right to copy at his/her expense, upon written request and by appointment during regular business hours.

The unit member shall be responsible for supplying the administration with all of the necessary documents requested by the administration in order to complete the unit member's files. The cost of supplying any materials after the unit member has complied with the aforementioned request shall be paid by the Board.

If any additional material is included within a unit member's files after the effective date of this Agreement, the unit member shall be sent a copy of such material within seven (7) calendar days thereafter.

The unit member shall have the right to file a statement in response to any written documents placed in his/her files.

- 5.02 Each College shall maintain a grievance file separate from the official personnel file.
- 5.03 The Colleges shall maintain the confidentiality of these files in accordance with state and federal law.
- 5.04 Whenever any individual or individuals inspect the official personnel file of a unit member, the date and name of the individual or individuals conducting such investigation shall be noted in the file.

# ARTICLE VI - DEDUCTION OF DUES AND AGENCY FEES

6.01 Dues Deductions

A Payroll deductions for membership dues of the Association shall be authorized pursuant to applicable laws of the Commonwealth.



B. The Association shall, at least thirty (30) days prior to the beginning of the academic year or within thirty (30) days of the effective date of this Agreement, give written notice to the Board of the amount of dues which are to be deducted on behalf of the Association and the names of the treasurers of the local affiliates of the Association.

## 6.02 Agency Service Fee

As a condition of employment during the term of this Agreement, every member of the bargaining unit who is not also a member of the Association shall pay or by payroll deduction shall have paid to the said Association an agency service fee that shall be one hundred per centum (100%) of the then current rate of dues payable by a member of the said Association; provided, however, that no such payment or deduction shall include any amount that represents a cost not related by the provisions of Section 3 of Article IX of the Rules and Regulations Relating to the Administration of Chapter 150E of the General Laws as such rules and regulations have been promulgated, and as they may be amended from time to time by the Labor Relations Commission. Such fee so required to be paid shall be payable on or after the thirtieth (30th) day next following the beginning of employment of such member of the bargaining unit or on or after the thirtieth (30th) day next following the effective sate of this Agreement, whichever shall be later.

Such fee may be paid by payroll deduction if so authorized pursuant to an Agency Service Fee Deduction Authorization: provided, however, that such authorization shall be deemed to have effect only with respect to such sum as is herein provided.

Any other provisions of this Agreement to the contrary notwithstanding every unit member who shall have failed to furill the condition of employment as is herein prescribed shall be subject to immediate dismissal and shall be so dismissed by the Board with effect no later than the end of the semester during which the Board shall have acted to dismiss him/her in accordance with the provisions of this Article; provided, however, that such dismissal shall be effected by notice promptly issued by the Board or its designee to such unit member after the Association shall have notified the President that such unit member has not fulfilled the condition herein prescribed. The said notice shall be sent by registered mail, return receipt requested, and shall give such unit member fourteen (14) days from the date of its receipt to fulfill the said condition. Within the said fourteen (14) days, the Board or its designees shall grant such unit member such opportunity to respond to the said notice as the Board may from time to time prescribe for the purposes of this provision.

Whenever such unit member shall not have fulfilled the condition herein prescribed within the said fourteen (14) days, the Board shall act to dismiss him/her at its meeting next following the expiration of the said fourteen (14) days; provided, however, that the Board need not so act if such unit member fulfills the said condition prior to the date of such meeting.

The Association shall reimburse the Board for any expenses incurred as a result of being ordered to reinstate an employee terminated at the request of the Association for not paying the agency fee. The Association shall intervene in and defend any administrative or court litigation with respect to any unit member's refusal or failure to pay the agency service fee, including, but not limited to, actions or claims arising from an employee's termination or dismissal. In such litigation, the Board shall have no obligation to defend the Association and the Association shall indemnify and hold the Board harmless from any loss occasioned by such litigation.

Any unit member who pays an agency service fee may on request obtain a rebate of part of his/her agency service fee in accordance with General Laws, Chapter 150E, Section 12, and Association procedures.

#### ARTICLE VII - ACADEMIC FREEDOM & RESPONSIBILITY

#### A. Academic Freedom

The Board and the Association endorse the principles and standards of academic freedom and academic responsibility as generally and traditionally accepted in institutions of higher education. The parties agree to promote public understanding and support of academic freedom and agreement upon procedures to assure academic freedom in Colleges and universities. Institutions of higher education are conducted for the common good and not to further the interest of either the individual teacher or the institution as a whole. The common good depends upon the free search for truth and its free exposition. Academic freedom is essential to these purposes and applies to both teaching and research.

Academic freedom is the right of scholars in institutions of higher education free!y to study, discuss, investigate, teach, exhibit, perform and publish. Freedom in research is fundamental to the advancement of truth. Academic freedom in its teaching aspect is fundamental for the protection of the rights of the teacher in teaching and of the student to freedom in learning.

The teacher is entitled to full freedom in research and in the exhibition, performance and publication of the results of his/her research, to full freedom in the classroom in discussing his/her subject, and most specifically in the selection of his/her classroom materials, including the selection of texts. The teacher is entitled to discuss controversial issues. As both a teacher and scholar, he/she recognizes his/her professional obligation to present various scholarly opinions and to avoid presenting totally unrelated materials, that being fundamental to the advancement of truth.

A faculty nuember has the right to determine the amount and character of the work and other activities he/she pursues outside the College; provided such work and other activities do not interfere with the discharge of his/her re-



sponsibilities under the terms of this Agreement. Unit members are entitled to freedom of expression of political belief or affiliation.

## B. Academic Responsibilities

Academic freedom carries with it correlative responsibilities.

The faculty member has the responsibility to his/her colleagues and the College community to preserve intellectual honesty in his/her teaching and his/her research. He/she respects the free inquiry of his/her assignates and avoids interference in their work.

The parties recognize that adherence to the complementary concepts of academic freedom and academic responwill most nearly ensure that the greatest contributions to the several Colleges will be made by their most ande resource, the faculty. The College or university teacher is a citizen and a member of a learned profession affiliated with an educational institution. When he/she speaks or writes as a citizen, he/she should be free from institutional censorship or discipline, but his/her special position in the community in: poses special obligations. As a person of learning, affiliated with an educational institution, he/she should remember that the public may judge his/her profession and his/her utterances. Hence, he/she should at all times be accurate, should exercise appropriate restraint, should show respect for the opinion of others, and should make every effort to indicate when he/she is not an institutional spokesperson.

Institutions of higher education are committed to the search for truth and knowledge and to contributing to the solution of problems and controversies by the method of rational discussion.

# ARITCLE VIII - AFFIRMATIVE ACTION

- 8.01. The Board and the Association recognize and affirm their commitment to the policy of non-discrimination, equal opportunity and affirmative action in all aspects of employment, including, but not limited to, recruitment, selection, placement, tenure, wages, training, retenchment, promotion and termination. The Board and the Association agree to cooperate in the administration of the Board's Affirmative Action Policy and in the implementation of any Affirmative Action Policy as may be adopted by the Board based on or mandated by federal or state law. The Board and the Association recognize that the realization of equal opportunity shall be based on their mutually cooperative good faith efforts to achieve full and prompt utilization of wome and minorities through the Board's Affirmative Action Policy.
- 8.02 The Board and the Association agree that the Joint Study Committee shall establish a subcommittee to study and report on the administration of the Board's Affirmative Action Policy.

# ARTICLE IX - SUPPLEMENTAL BENEFITS

# 9.01 Authorized Leaves with Pay

#### A Sick Leave

#### 1. Entitlement

All faculty members shall be entitled to ten (10) days of sick leave for each academic year of service. All professional staff members shall be entitled to fifteen (15) days of sick leave for each year of service. Sick leave credit shall begin with the first (1st) full month of employment and accumulate monthly as follows:

- a. Faculty members shall accumulate at the rate of one and one-ninth(1 1/9) days of sick leave for each full month of employment.
- b. Professional staff members shall accumulate at the rate of one and one-quarter (1 1/4) days of sick leave for each full month of employment.

Credits for periods of .acs than one (1) full month's employment shall not be allowed. Sick leave not used in any year may be accumulated. No person shall be entitled to a leave of absence with pay on account of sickness in ex ess of the accumulated sick leave then due, except as provided under Section 9.01 A4 – Sick Leave Bank.

# 2. Reinstatement

Unit members who are reinstated shall be credited with sick leave credits as have accrued at the termination of their previous service. No credit for previous service may be allowed where reinstatement occurs after an absence of three (3) years or more from the date of termination of their previous service unless approval of the Board is secured for any of the following reasons:

- 1. Illness of such person and not because of illness of his/her immediate family;
- 2. Dismissal through no fault or delinquency attributable to such person; or
- 3. Injury while in the service of the Commonwealth in the line of his/her duties for which such person would be entitled to receive Workmen's Compensation benefits.

#### 3. Use of Sick Leave

Sick leave shall be granted at the sole discretion of the President of the College under the following conditions:

Ţ

- When a unit member cannot perform his/her duties because he/she is incapacitated by personal illness or injury;
- b. When, through exposure to contagious disease, the presence of the person at his/her post of duty would jeopardize the health of others;
- c. In case of serious illness of husband, wife, child, parent of either spouse or of any other person subject to these rules, or of any person living in the immediate household of a person subject to these rules, he/she may be granted sick leave with pay not to exceed seven (7) working days within the fiscal year.
- d. A unit member shall be entitled to sick leave with pay, not to exceed ten (10) consecutive working days, in the event his wife gives birth.

Notification of absences shall be given as early as possible on the first (1st) day of absence. If such notification is not made, such absence may, at the discretion of the College President or his/her designee, be applied to absence without pay. For any period of absence on account of sickness, the College President or his/her designee may require a physician's certificate proving the necessity of such absence. If such certificate is not filed within seven (7) calendar days after a request therefor, such absence may be applied at the discretion of the College President or his/her designee to absence without pay.

#### 4. Sick Leave Bank

- a. Upon the date of execution of this Agreement, there shall be established, or continued in the case of those Colleges having previously so established, a Sick Leave Bank.
- b. During the term of this Agreement, a unit member who is not a member of the Sick Leave Bank may become so during the month of October by giving written notice to the President of the College or his/her designee that he/she is assigning to the Board one (1) day of his/her personal sick leave accumulation.
- c. The President of the College or his! her designee shall maintain a register of the membership and the number of sick leave days accumulated in the Bank.
- d. Five (5) working days after the exhaustion of his/her personal sick leave accumulation, every member of the Sick Leave Bank shall be entitled to draw upon the Sick Leave Bank, effective thereafter upon notice to the President of the College. The granting of such sick leave shall be subject to the same criteria as regular sick leave days and shall be in all other respects consistent with Board policy; provided, however, that such sick leave shall be available only for the illness of the employee and not for the illness of the family.
- e. Whenever the accumulation of sick leave days in the Sick Leave Bank shall have fallen to fifty (50) days, the President of the College or his/her designee shall notify all members. Thereafter, one (1) personal sick day from each member's accumulated sick leave shall be assigned to the Bank unless a member notifies the President of the College or his/her designee in writing within five (5) days after receipt of said notice that he/she does not wish to remain a member; provided, however, that any member of the Sick Leave Bank wishing to remain a member thereof and who shall have exhausted his/her personal sick leave accumulation on the date of the giving of such notice, shall assign such additional days within fifteen (15) days after the date on which such member is entitled to personal sick leave and shall retain all his/her rights in the Bank until such period of assigning an additional day shall have expired.

No unit member may draw upon the Sick Leave Bank in excess of the number of days to which he/ she is entitled as determined by the administration.

A unit member who receives disability compensation provided by statute and who is entitled to any personal sick leave allowance may take so much of his/her personal sick leave allowance payment which, when added to the amount of disability compensation provided by statute, shall result in the payment to him/her of his/her full salary. The Sick Leave Bank shall not be used for this purpose.

# 5. Certification and Notification of Sick Leave Status

The President of the College or his/her designee shall notify every unit member on or before September 15th of each year of the number of sick days accumulated to his/her credit as of September 1.

Unit members shall be paid twenty (20) percent of the value of their unused accrued sick leave at the time of their retirement; provided, however, that such payment shall not enlarge or diminish those pension benefits for which a unit member would otherwise be entitled.

Whenever because of prolonged illness, the accumulated sick leave of a unit member in expended, the unit member may request from the Board an entension of sick leave benefits for the purposes provided in this Article.

# B. Bereavement Leave

Upon notification to the President of the College or his/her designee of the death of the husband, wife, child, parent of either spouse, brother or sister of a person subject to this Agreement, or of a person living in the immediate household of a person subject to this Agreement, leave of absence with pay may be granted for a period not exceeding four (4) consecutive calendar days.

# C. Military Leave

1. Every unit member shall be entitled, during the term of his/her service in the armed forces of the Commonwealth, under Sections 38, 40, 41 or 60 of Chapter 33 of the General Laws, or during his/her annual tour of



duty, not exceeding seventeen (17) days as a member of a reserve component of the armed forces of the United States, to receive pay therefor, without loss of his/her ordinary remuneration and shall in addition be entitled to all leaves of absence provided under this Agreement.

- 2. Any unit member who is a member of a reserve component of the armed forces of the United States and who is called for duty other than the annual tour of duty not exceeding seventeen (17) days shall be subject to the provisions of Chapter 708 of the Acts of 1941, as amended, or of Chapter 805 of the Acts of 1950, as amended.
- 2. Any unit member, who on or after January 1, 1980 shall have tendered his/her resignation or otherwise terminated his/her employment for the purpose of service in the military or naval forces of the United States, and who does or did serve or was or shall be rejected for such service, shall, except as otherwise provided by Chapter 708 of the Acts of 1941, as amended, be deemed to be or to have been on military leave, and no such unit member shall be deemed to have resigned or to have terminated his/her employment until the expiration of two (2) years from the termination of said military or naval service by him/her.

#### D. Court Leave

- 1. Unit members who are called for jury duty or are summoned to appear as witnesses on behalf of any town, city, county, state or federal government shall be granted court leave. Notice of service shall be filed with the President of the College or his/her designee upon receipt of summons.
- 2. If jury or witness fees received by a unit member amount to more than his/her rate of compensation, he/she may retain the excess of such fees and shall submit the regular rate of compensation together with a court certificate of service to the Board or its designee, and shall be deemed to be on leave of absence with pay. If the jury or witness fees amount to less than the unit member's regular rate of compensation, he/she shall be deemed to be on leave of absence with pay and he/she shall remit said fees to the Board or its designee with a court certificate of service.
- 3. Expenses submitted by the court for travel, meals and room hire shall be retained by the unit member and shall not be considered part of the jury or witness fees.
- 4. Whenever a unit member is called for jury duty or summoned to appear as a witness and such jury duty or appearance occurs during his/her vacation, there shall be no necessity to account to the Board or the College for any fees received during such period.
- 5. When a unit member has been granted court leave for jury or witness service, and is excused by proper court authority, he/she shall report back to his/her official place of duty whenever the interruption in jury or witness service will permit four (4) or more consecutive hours of employment.
- 6. Court leave shall not affect employment rights of unit members.
- 7. Court leave shall not be granted when a unit member is the defendant or is engaged in personal litigation.

# E. Other Leaves

Such other leaves as are herein authorized shall also be available to unit members, provided that whenever the granting of any such leave is discretionary, such discretion shall be exercised by the President of the College or his/her designee. Leave of absence with pay may be granted for the following reasons:

- 1. To permit unit members who are veterans to pay tribute at the funeral in Massachusetts of dead veterans. The President of the College or his/her designee shall grant leave of absence with pay to veterans who are members of firing squads, color details, pall bearers, buglers or escorts participating in such service.
- 2. Unit members shall be entitled to leave of absence with pay to allow for loss of timedue to prophylactic inoculation required as a result of their employment. If such absence with pay exceeds one (1) week, the Board or its designee shall, if legally appropriate and as soon as practicable, initiate a Workmen's Compensation claim and further payments because of such prophylactic inoculation shall cease.
- 3. Unit members shall be entitled to leave of absence with pay to allow for the period of absence due to quarantine because of exposure to contagious disease in the regular performance of duty.
- 4. Red Cross blood donations.
- 5. Oral, written and physical examination for state service conducted by the Division of Civil Service or the Division of Registration; and physical examination for state retirement.
- 6. To attend hearings in Industrial Accident cases as the injured person or as a witness therein. Any witness fees received by such injured person or witness shall be paid to the Board.
- 7. To allow attendance by unit members who are delegates or alternates to state or national conventions of the following veterans' organizations: American Legion, AMVETS or World War II, Disabled American Veterans, Legion of Valor, Marine Corps League, Order of the Purple Heart, United Spanish War Veterans, Veterans of Foreign Wars, Reserve Officers Association of the United States, Veterans of World War I of the U.S.A., Inc.
- 8. Leave of absence with pay not to exceed two (2) hours shall be granted to any unit member, if he/she makes application therefore to permit him/her to vote in the voting precinct, ward or town in which he/she is entitled to



vote: provided that the hour of opening and the hour of closing of the polls at such place would otherwise preclude his/her traveling to or from the polls.

9. Leave of absence with pay may be granted for officers, delegates or alternates of employee organizations for the purpose of attending conventions of their organizations.

# F. Vacation Leave for Professional Staff Members

#### 1. Entitlement

- a. All professional staff members who work a twelve (12) month year shall be entitled to an annual vacation leave of twenty-two (22) days.
- b. A professional staff member who has completed seven (7) full years of service but less than seventeen (17) full years of service on June 30 of any vacation year shall be granted annual vacation leave of twenty-three (23) days.
- c. A professional staff member who has completed seventeen (17) full years of service but less than twenty-six (26) full years of service on June 30 of any vacation year shall be granted annual vacation leave of twenty-five (25) days.
- d. A professional staff member who has completed twenty-five (25) years or more of service on June 30 of any vacation year shall be granted annual vacation leave of thirty (30) days.

## 2. Accrual of Vacation Days

Professional staff members shall accrue vacation days subject to the following provisions:

- a. The vacation year shall be for the period of July 1 to June 30 inclusive.
- b. All professional staff members shall be credited as of the first day of each month with vacation leave with pay accrued prior to that date not to exceed the following:
  - (1) For less than one (1) year of service beginning July 1 and ending on June 30, vacation leave of two (2) days for each full month of service not to exceed twenty-two (22) days.
  - (2) For one (1) full year's service beginning on the first (1st) working day of July and ending on June 30, vacation leave of 1.83 days for each full calendar month.
  - (3) Professional staff members having an aggregate or more than one (1) day of leave without pay and/or absence without pay in any calendar month shall not receive a vacation credit for that month.
  - (4) Professional staff members who are granted vacation leave in addition to twenty-two(22) days shall be credited with any additional days on July 1 following their anniversary date.

## 3. Scheduling of Vacation Days

Vacation leave shall normally be scheduled and taken during the vacation year in which it becomes available; provided, however, that a professional staff member may, at his/her discretion, carry over up to sixty-four (64) vacation days from year to year; provided, further that in no event shall vacation credits in excess of sixty-four (64) days be carried over for more than one (1) year unless otherwise approved by the Board. Unused vacation days in excess of sixty-four (64) days shall be converted to sick leave at the end of each fiscal year.

Persons who are eligible for vacation under these rules, whose services are terminated by dismissal through no fault or delinquency of their own, by retirement, or by entrance into the armed forces, shall be paid an amount equal to the vacation allowable as earned but not granted in the vacation year prior to such dismissal, retirement, or entrance into the armed forces, and in addition that portion of the vacation allowance earned in the vacation year during which such dismissal, retirement, or entrance into the armed forces occurred, up to the time of separation; provided that no monetary or other allowance had already been made therefor.

# 4. Vacation Status Report

The President of the College or his/her designee shall notify each professional staff member on or before July 15 of the number of vacation days accumulated to his/her credit as of July 1 preceding.

#### G. Holiday Pay

All unit members during the term of this Agreement shall be entitled to the following holidays:

New Year's Day
Martin Luther King Day
Washington's Birthday
Evacuation Day
Patriots' Day
Memorial Day
Bunker Hill Day

Independence Day
Labor Day
Columbus Day
Veterans' Day
Thanksgiving Day
Christmas

#### H. Personal Dava

#### 1. Entitlement

Each faculty member shall be entitled to two (2) personal days per academic year. A faculty member with an initial employment date after December 31 shall be allotted one (1) personal day for the remainder of the academic year.



Each professional staff member shall be entitled to three (3) personal days per fiscal year. Those professional staff members with initial employment dates between July 1 and September 30 shall be allotted three (3) days, those with initial employment dates between October 1 and December 31 shall be allotted two (2) days, and those with initial employment dates after January 1 shall be alloted one (1) personal day for the remainder of the fiscal year.

#### 2. Usage

A unit member who wishes to use a personal day shall, whenever practicable, submit a request no later than three (3) cays in advance. Such request shall not be unreasonably denied. Each faculty member using a personal leave day shall obtain coverage for his/her classes or, where appropriate, assign a self-directed learning experience.

Personal leave days may not be accumulated beyond the termination of the academic year for which they are allotted.

#### L Sabbatical Leave

#### 1. Purpose

The purpose of a sabbatical leave shall be for professional growth, research or study which may involve travel and which benefits the unit member and the College.

#### 2. Criteria for Sabbatical Leave

- 4. The following criteria shall be considered in determining who should be granted sabbatical leave.
  - (1) That the objectives of the sabbatical leave, if attained, would substantially contribute to the professional growth of the unit member.
  - (2) That the objectives of the sabbatical leave, if attained, would assist the unit member in substantially contributing to institutional needs and attainment of institutional purposes.
  - (3) That the unit member has the ability to achieve the goals of the project or plan based on his/her past experience and formal educational background.
  - (4) That the attainment of the objectives of sabbatical leave as proposed are realistic in terms of time, costs and other related variables.
  - (5) There exists independent financial support from other funding sources concerned with the proposed plan or project where College funding sources are otherwise unavailable.

#### 3. Eligibility

- a. Unit members must have completed six(6) years of continuous full-time service to the College to be eligible for consideration for sabbatical leave. Time granted for other types of professional leaves shall not apply to the six(6) year period required for sabbatical leave eligibility. In addition, a unit member must have received a summary evaluation of other than unsatisfactory on his/her most recent evaluation to be considered eligible.
- b. A unit member who may have previously received a sabbatical leave shall be eligible for consideration of a subsequent leave after six (6) years of continuous service upon returning from the original sabbatical leave.

#### 4. Limitation

Sabbatical leave is not an inherent right of the unit member and there is no obligation on the part of the Board to approve any and all sabbatical leave requests made by unit members. No more than five (5) percent of the full-time unit members at a given College shall be granted sabbatical leave in any academic year unless otherwise approved by the Board.

# 5. Length of Sabbatical

A unit member granted sabbatical leave may be granted one-half (1/2) year of leave at full or half salary, one (1) full year at half salary, or one-half (1/2) year or full year at half workload at full or half salary. A year is defined by the duration of the unit member's contract in the year prior to sabbatical.

# 6. Limitations on Sabbatical Stipend/Outside Employment

The Board encourages unit members to seek additional funds which would enhance directly the purpose of the sabbatical leave.

- a. Non-salary funds (travel, materials, and such other types of expenses) obtained from outside sources directly related to the purpose of the sabbatical leave shall not reduce the sabbatical leave stipend.
- b. Salary funds obtained from outside sources which directly relate to the attainment of the specific objectives of the sabbatical leave shall not reduce the sabbatical leave stipend unless the total amount of outside salary funds and sabbatical stipend exceed the current annual salary of the unit member or its prorated equivalent in the case of a half- year sabbatical leave. In no instance shall any activity required to earn or receive this outside salary interfere directly or indirectly with the purpose of time available for which the sabbatical leave was granted.



- c. Salary funds derived from outside source(s) which in addition to the sabbatical stipend exceed the current salary or its proration and which interfere with the purpose of the sabbatical leave shall result in the sabbatical stipend being reduced an equal amount so as to result in an amount equal to the current salary or its prorated equivalent.
- d. Failure to report additional outside salary shall result in the immediate reduction in a like amount of the sabbatical stipend if the leave is still in process. Should funds remaining in the sabbatical stipend be inadequate to liquidate the additional unreported remuneration or should the sabbatical leave have been completed, the unit member shall return a like amount to the College within six(6) months of the discovery of the unreported forms of remuneration above.

# 7. Post-Sabbatical Service

Prior to the granting of such leave, said unit member shall enter into a written agreement with the Board that upon the termination of such leave he/she shall return to the service of the Board and serve as a unit member within the System for a period of one (1) year, and that in default of completing such service, he/she shall refund to the Commonwealth, unless excused therefrom by the Board for reasons satisfactory to it, an amount equal to such proportion of the salary as the amount of service not actually rendered as agreed bears to the whole amount of services agreed to be rendered.

#### 8. Process

- a. The applicants for sabbatical leave shall submit their proposed plan concurrently to their immediate supervisor and the standing Sabbatical Leave Committee which shall be appointed annually by the President of the College. Applications must be submitted no later than seven (7) months prior to the starting date of the leave.
- b. The President of the College shall annually invite recommendations for appointment to the Sabbatical Leave Committee. In determining the membership of the Sabbatical Leave Committee, the President of the College shall appoint, insofar as possible, unit members who have successfully completed sabbatical leaves and who represent the unit membership. Unit membership on said Committee shall be proportionally representative of the faculty and professional staff members eligible to apply for sabbatical leave. The Committee size should be of an uneven number not to exceed seven (7) nor be less than five (5). A unit member shall not serve on the Committee during a year in which he/she applies for sabbatical.
- c. The Committee shall rank each eligible candidate who has applied for sabbatical leave in terms of the criteria stated for sabbatical leave. The recommendations of the Sabbatical Leave Committee in rank order shall be concurrently submitted to the Dean(s) and the applicants no later than six (6) months prior to the starting date of the leave. The immediate supervisor shall forward his/her recommendation to the appropriate Dean with a plan for coverage of the sabbaticant's classes, services, or other professional responsibilities by other unit members and, where appropriate, an estimate of the cost of replacement relative to existing budgeted salaries. Said recommendations shall be forwarded no later than six (6) months prior to the starting date of the leave.
- d. The Dean(s), at ting as a committee, shall consider the recommendations of the Sabbatical Leave Committee and the immediate supervisor and recommend to the President of the College candidates for sabbatical leave with a plan for coverage of the sabbaticant's classes, services, or other professional responsibilities by unit members and, where appropriate, a final estimate of the replacement costs. Copies shall be transmitted to the applicant, the applicant's immediate supervisor and the Chairperson of the Sabbatical Leave Committee.
- e The President of the College shall evaluate the recommendations of the Sabhatical Leave Committee and the Dean and recommend sabbatical leave for those unit members who he/she deems rank highest on the criteria; provided, however, that if the President's recommendation differs from that of the Sabbatical Leave Committee, the President of the College shall in addition attach his/her recommendation and a written statement of reasons therefore which shall be based on applicable criteria established for the granting of sabbatical leaves. Said recommendations shall be forwarded to the Board no later than five (5) months prior to the start of the leave.
- f. The Board shall notify an applicant, where practicable, no later than three (3) months prior to the start of the leave whether his/her application has been approved.

#### 9. Funding

A unit member shall, if possible, provide coverage for his/her classes, services or other professional responsibilities for the period during which he/she is on sabbatical leave; provided, however, that if in the determination of the President of the College or his/her designee it is necessary to hire replacement(s) for the sabbaticant's classes, services or other professional responsibilities, such replacement shall to the extent practicable be provided for within the College's existing salary funds by reducing services, classes and/or by hiring part-time replacements.

#### 10. Report of Activity and Accomplishment

A unit member who is granted sabbatical leave shall submit a report of the results of the sabbatical leave within



ninety (90) days of the expiration of said leave on such form as may be promulgated by the Board. The report shall include:

- a. An account of activities during the leave, including travel, itineraries, institutions visited and persons consulted.
- b. A statement of progress made on the sabbatical leave as proposed in the application and an explanation of any significant changes made in the program
- c. An appraisal of the relationship between the results obtained and those anticipated in the sabbatical leave program statement.
- d. A final account of all sources and kinds of salary support, other than the sabbatical stipend, with a description of the sabbaticant's activity, if any, necessary to receive each such type of fund.

# 11. Impact of Sabbatical on Conditions of Employment

- Unit members shall fully participate in the following fringe benefits during the sabbatical year and continue to contribute to those requiring employee contributions.
  - 1. sick leave accumulation
  - 2. leave for death or illness in immediate family
  - 3. term insurance
  - 4. medical insurance
  - 5. retirement based on actual salary paid for year
  - 6. membership in the Health and Welfare Fund when such is implemented.
- b. Unit members shall participate in vacation day accumulation on a prorated basis, determined by their non-sabbatical period of employment during the contractual year.
- c. Unit members may use their sabbatical year as a year of service toward eligibility requirements and may be considered for evaluation purposes in promotion and tenure decisions; provided, however, that a unit member shall not be promoted during the year he/she is on a sabbatical leave.
- d. This section shall not be applicable to non-state appropriated funded unit members; provided, however, that service rendered by such unit members shall be counted on a one (1) year for two (2) year basis in determining eligibility for sabbatical leave after such unit member becomes an "01" employee.

# 9.02 Unpaid Leaves of Absence

Any unit member granted an unpaid leave of absence shall retain those benefits accrued prior to the period of his/her leave, including seniority; provided, however, that such unit member shall be placed upon the salary schedule at the step and rank he/ she held prior to the leave, as adjusted by the provisions of any collective bargaining agreements between the Board and the Association. There shall be no promotion nor shall there be any entitlement to any performance based awards that might have become available during the leave of absence. This clause shall apply to all unpaid leaves of absence subject to the exception cited in Section (d.) below, unless as otherwise provided herein. Unpaid leaves shall not be unrearonably denied.

#### A. Professional Leave

#### 1. Purpose

Upon the application of a unit member and a recommendation of the President of the College, the Board or its designee may grant to such unit member leave without pay for up to three (3) years for professional reasons as provided herein. The purpose for which a unit member may submit his/her application for such unpaid leave shall include, but shall not be limited to:

- a. advanced study;
- b. participation in a program of exchange teaching in accordance with the conditions set forth under the provisions of 9.02 D;
- c. participation in a program related to his/her professional responsibilities;
- d service as an officer or staff member of any recognized professional organization;
- e. service in a public office to which he/she has been elected or appointed and for such other purposes as may be allowed under the laws of the Commonwealth.

#### 2. Eligibility

Unit members shall be eligible for such leave after six (6) years of full-time service. In addition, the unit member must have received a summary evaluation of other than unsatisfactory on his/her most recent evaluation to be considered eligible.

# 3. Conditions

The granting of a professional leave shall be subject to the following conditions:

a. the applicant's professional duties permit his/her absence for the period of time requested; and/or



b. the leave is of value to the individual and to the College as determined by the President of the College or his/her designee.

#### 4. Procedure

Applicants shall prepare a proposal which describes the prospective activity and indicates the contribution it will make to the individual concerned and the College. The proposal shall be submitted to the appropriate Dean six (6) months in advance of the requested leave; provided, however, that the President of the College or his/her designee may waive the six (6) month application period. Prior to making his/her recommendation to the President of the College, the Dean upon request shall first inform the applicant of his/her intended recommendation. The Dean shall forward his/her recommendations to the President of the College, with a copy to the unit member within thirty (30) days of receipt.

The President of the College shall notify the unit member and the appropriate Dean of his/her decision and shall forward his/her recommendation to the Board within sixty (60) days of receipt.

#### 5. Continuation

Any unit member requesting a continuation of said leave for any succeeding period beyond that initially granted shall do so at least one (1) semester prior to the expiration of said leave.

# **B.** Maternity Leave

# 1. Entitlement

A female unit member who is employed by the Board and who has given notice, when possible, at least three (3) months prior to her anticipated date of departure is entitled to be absent from such employment for a period certified by her physician due to disabilities caused or contributed to by pregnancy and recovery therefrom. Sick leave may be utilized for any period of disability; provided that the President of the College or his/her designee may require that her physician certify that any period in excess of five (5) days to which sick leave is applied is medically necessary as a result of or to recover from said disability.

#### 2. Conditions

Disabilities caused or contributed to by pregnancy, abortion, miscarriage, childbirth, and recovery therefrom shall be treated like any other temporary disability. After all employee-designated sick and vacation leave have been exhausted, a maternity leave of up to twelve (12) months without salary shall be granted, except that adjustments in the duration of the leave may be made by the President of the College or his/her designee to insure that such leave is least disruptive of the instructional progress of students. Time on maternity leave shall be considered as continuous service for the purposes of sabbatical leave but shall not accrue as accumulated time required to qualify for sabbatical leave, award of tenure or promotion.

#### 3. Reinstatement

The unit member shall be restored, subject to retrenchment and any other provisions resulting in faculty or professional staff member reductions, to the same or substantially similar position with the same salary and fringe benefits which she had attained at the time maternity leave was granted, plus any adjustments made as a result of collective bargaining; provided that the unit member returns to full-time service within twelve(12) months from the beginning of the leave or within twelve(12) months from the beginning of any additional leave granted by the President of the College or his/her designee. The President of the College or his/her designee may allow a unit member up to twelve(12) months additional leave; provided, however, that such decision shall not be grievable. For up to twelve(12) months from the beginning of maternity leave, part-time return to service may be arranged by approval of the President of the College or his/her designee when such can be reasonably accommodated; provided, however, that in the case of an employee holding a terminal appointment, a leave shall not extend beyond and provisions for re-employment shall not apply beyond the termination date of the appointment unless the employer agrees in writing to such extension; provided further that such decision to allow part-time return shall not be grevable.

#### C. Family Leave

#### 1. Entitlement

Upon written application to the President of the College, including a statement of reasons, any full-time unit member who has been employed at least one (1) year and who has given notice at least one (1) semester prior to his/her anticipated date of departure, unless otherwise approved by the President of the College or his/her designee, shall be granted parental or adoptive leave without pay from such employment for period not exceeding one (1) academic year. The President of the College or his/her designee may grant a unit member a half-time leave with full benefit:, provided, however, that such decision shall not be grievable.

# 2. Purposes

The purpose for which a unit member may submit his/her application for such unpaid leave shall be the need to care for or to make arrangement for the care of an unemancipated minor child of the unit member, spouse of the unit member or parent of the unit member.

#### 3. Conditions

Time used for family leave shall be considered as continuous service for the purpose of sabbatic al leave but shall



not be included as accumulated time required to qualify for sabbatical leave or award of tenure or promotion.

#### 4. Reinstatement

A unit member shall be restored, subject to retrenchment and any other provisions resulting in faculty or professional staff member reductions, to the same or a substantially similar position with the same salary and fringe benefits which he/she attained at the time family leave was granted, plus any adjustments made as a result of collective bargaining; provided that the unit member returns within one (1) academic year from the beginning of the leave or within one (1) year of the beginning of any additional leave granted by the President of the College or his/her designee. The President of the College or his/her designee may allow a unit member up to one (1) academic year additional leave; provided, however, that such decision shall not be grievable.

#### D. Exchange Teaching

- I. Notwithstanding any other provisions of this Article to the contrary, a leave of absence of up to two (2) years may be granted to any faculty member upon application for the purpose of participation in exchange teaching programs in other states, territories, countries, or an educational or cultural program related to his/her professional responsibilities. On return from such leave, a faculty member shall be placed at the salary held by him/her at the time the leave commenced, except as provided in (2) below.
- 2. In the event there is in the judgment of the President of the College or his/her designee a valid performance evaluation conducted by the host College, the substance of that evaluation shall be considered as evidence of performance qualifying the faculty member for possible award of a performance based award.
- 3. Each faculty member shall submit a certified statement from the appropriate academic administrator of the host college before the first (1st) semester of his/her return describing in detail the exchange teaching responsibility and/or educational/cultural experience.

#### E. Professional Staff Leave

Professional staff members may, upon mutual agreement between the professional staff member and the President of the College or his/her designee, be granted professional leave without pay for a continuous period of not less than two(2) weeks nor more than eight (8) weeks during the months of June. July and August; provided, however, that unit members granted such leave shall not be denied eligibility for promotion, sabbatical or professional leave, tenure or performance based awards.

# 9.03 Association Leave

- A All meetings referred to in this section shall be scheduled so as to provide the least disruption of classes.
- B. When it is necessary, pursuant to the Grievance Procedure in Article X of this Agreement, for a representative designated by the Association to attend a hearing during a work day, he/she shall upon notice to his/her immediate supervisor be released without loss of pay as necessary in order to participate in the foregoing activities. Any person whose ap, parance in such hearings as a witness is necessary shall, when possible, obtain coverage for his/her classes satisfactory to the President of the College.
- C. When the Board of Directors' meetings of the Association are scheduled during normal working hours of a work day, unit member representatives to such meetings shall be relieved from all regular duties without loss of pay; provided, however, that such release from duties shall be limited to no more than ten (10) meetings. The President of the Association shall notify the Chancellor or his/her designee and the Presidents of the affected Colleges of the names of the Board of Directors and the dates of the ten (10) scheduled Board of Directors' meetings prior to September 1. Whenever it becomes necessary to schedule additional meetings of the Association's Board of Directors, such members may be released; provided, however, that they first obtain coverage for their classes satisfactory to the President of the College.

#### 9.04 Insurance and Other Benefits

#### A. Health and Accident Insurance

Unit members shall continue to be covered under the State's Group Health and Accident Insurance plan currently in effect pursuant to the provisions of Chapter 32 A of the General Laws as amended or as such plan may be made available under applicable law of the Commonwealth.

#### **B.** Pension

Unit members shall continue to be covered under the State's Retirement Plan pursuant to the provisions of General Laws, Chapter 32, or such plan as may be made available under applicable laws of the Commonwealth.

#### C. Workmen's Compensation

Unit members shall continue to be covered under the provisions of the State Workmen's Compensation Act, pursuant to General Laws, Chapter 152.

#### D. Tax- Sheltered Annuities

The Board shall continue to provide for the purchase of tax-sheltered annuities by unit members pursuant to the provisions of General Laws, Chapter 15, Section 18A, or Chapter 15A.



#### E. Tuition Waiver

The spouse and/or child or children, including any adopted or stepchild or children of any unit member, who after the date of execution of this Agreement, shall have been admitted as a student in the regular day program at any College in the Community College System shall be entitled to matriculate as a student in such program without payment of any tuition save as is provided in Massachusetts General Laws; previded, however, said tuition free enrollment within the Community College System shall be limited to the day division only.

In addition, effective for the semester following the date of execution of this Agreement, each unit member and the spouse and dependent child or children shall be eligible for system-wide tuition remission as follows:

- A. For enrollment in any state-supported course or program at the undergradua 2 or graduate level at any Community College, State College, or University excluding the M.D. Program at the University of Massachusetts Medical School, full tuition remission shall apply;
- B. For enrollment in any non-state supported course or program offered through continuing education, including any community service course or program at any Community College, State College or University, fifty percent (50%) tuition remission shall apply;
- C. Tuition remission shall apply to non-credit as well as credit-bearing courses;

in accordance with all the provisions and conditions of the System-wide Tuition Remission Policy for Higher Education Employees issued by the Chancellor on May 21, 1984. This policy shall remain in effect for the duration of this agreement.

# F. Travel and Conference Expenses

- Travel expenses, including reasonable charges for hotel rooms and registration fees to all approved conferences, meetings, workshops, student activities and any other business of the College which is required and authorized by the President of the College or his/her designee in his/her absence, shall be paid by the Board.
- 2. Whenever use of an employce's private car is necessary and has been authorized by the appropriate administrator, the approved mileage rate of twenty-two(22) cents per mile shall be allowed unless a higher rate is otherwise authorized by law and approved by the Board. This approved rate covers all charges, including by way of example garage; parking and toll charges. Whenever a unit member has been assigned, in writing, a second work location other than his/her principal work location by the President of the College or his/her designee, travel shall be paid either for the distance from his/her home to the second work location or from his/her principal work location to the second work location, whichever is nearer.
- Whenever use of any other mode of transportation is necessary and has been so authorized, the cost of all fares
  less federal taxes shall be allowed; provided that such receipted bills shall be first submitted for such
  charges.
- 4. Reimbursement shall not be made for expenses incurred for the sole benefit of the traveler, such as by way of example valet service, entertainment and laundry service.
- 5. When an employee on full travel status is engaged in travel, reimbursement shall be allowed for actual meal expenses incurred; provided that said expenses shall not exceed the following meals allowance schedule:

Meal	Maximum Allowance	Applicable Period
Breakfast	<b>\$</b> 2.50	3:01 a.m. to 9:00 a.m.
Lunch	4.00	9:01 a.m. to 3:00 p.m.
Supper	7.00	3:01 p.m. to 9:00 p.m.

#### G. Health and Welfare Fund

- (a.) The Board and the Association agree to establish a Health and Welfare Fund or Funds under an Agreement and Declaration of Trust. The initial design of such trust fund shall be negotiated by the parties and once established the fund shall be jointly administered by an equal number of trustee members. The Agreement and Declaration of Trust shall be executed no later than ninety (30) days after the effective date of this Agreement. The trustees' task, among other things, will be the selection of any additional benefits or the increasing of existing benefits dependent upon the size of the fund.
- (b.) Effective January 1, 1984, the Board agrees to contribute to the Health and Welfare Fund on behalf of each full-time equivalent bargaining unit member the sum of two dollars (\$2.00) each calendar week.

  Effective December 30, 1984, the Board agrees to contribute to the Health and Welfare Fund an additional two dollars (\$2.00) each calendar week for each full-time equivalent bargaining unit member.

  The contributions made by the Board to the Health and Welfare Fund shall be used for health and welfare benefits to be determined by the trustees of the Trust Agreement and to pay the operating and administering expenses of the Fund. The contributions shall be made by the Board in an aggregate sum within forty-five (45) days following the end of the calendar month during which contributions were collected.
- (c.) No dispute over a claim for any benefits extended by this Health and Welfare Fund shall be subject to Article X Grievance Procedure.



(d.) It is expressly agreed and understood that the Board does not accept, nor is the Board to be charged with hereby, any responsibility in any manner connected with the determination of liability to any bargaining unit member claiming under any of the benefits extended by the Health and Welfare Fund; such liability shall be limited to the contributions indicated under subsection 9.04(G) above

# ARTICLE X - GRIEVANCE PROCEDURE

- 10.01 It is the intent of the parties to this Agreement to use their best efforts to encourage the informal and prompt settlement of grievances which may arise between the Association or a member or members of the bargaining unit and the Board In recognition of this intent, the parties agree that they shall use the procedure set forth in this Article for the resolution, strictly pursuant to the terms of this Agreement, of all disputes involving the application of this Agreement. The Association further agrees that it shall not initiate proceedings in any other forum in respect of any matter that is or may become the subject of a grievance as hereinbefore defined until it shall have first exhausted the procedures provided herein.
- A Any member(s) of the bargaining unit may initiate and pursue a grievance through the first (1 st) two (2) steps of the grievance procedure without intervention of the Association, provided that a representative of the Association shall be afforded the opportunity to be present at any conferences held; and provided further that any disposition made of any grievance under this Article shall not be inconsistent with the terms of this Agreement.

  Any member of the bargaining unit may be represented by the Association at any step of the grievance procedure.
  - B. Failure of a grievant(s) to comply with any of the provisions of this Article shall be deemed to be a waiver of the right to seek resolution of the grievance under the terms of this Article. In determining whether there has been any such failure to comply with any of the provisions of this Article, time shall be of the essence and any failure of the grievant to comply with any of the time limits prescribed herein shall be deemed to be a waiver of the right to seek resolution of the grievance under this Article; provided, however, that the time limits prescribed herein may be extended in any specific instance by mutual agreement of the parties or by oral agreement confirmed in writing.
  - C. The grievant(s) may file a complaint with the Chancellor or his/her designee in the event that the administration fails to comply with any of the provisions of this Article, including time limits. The Chancellor or his/her designee shall determine within fourteen (14) calendar days whether or not the administration has failed to comply with the provisions of this Article; provided that the administrator who is the subject of a complaint under this paragraph shall not be designated to determine that complaint. If the Chancellor or his/her designee determines that the administration has failed to so comply, then he/she shall state in writing the remedy he/she deems appropriate. Copies of the Chancellor's or his/her designee's decision shall be sent to the Association and the appropriate administrators. The decision of the Chancellor or his/her designee shall be final.
  - D. The resolution of a grievance at any such shall not be deemed to be an admission by the Board or the Association of any violation or breach of the terms of this Agreement, or that such grievance is judicially cognizable or legally sufficient pursuant to any applicable provisions of the laws of the Commonweal, hor shall it establish any precedent nor grant to the Association or any unit member standing to initiate proceedings or pursue a remedy in any other forum.
  - E. If any member or members of the bargaining unit shall initiate in any other forum or under the College's or Board's Affirmative Action Plan any proceeding that relates to any matter that is the subject of a pending or disposed of gnevance or arbitration in respect of which such member or members is or are the grievant(s), such grievance proceeding shall terminate as of the date of the initiation of such other proceeding and the grievance and arbitration procedures delineated in this Article shall in inapplicable to such grievance; provided that such unit member(s) shall have no further legal recourse under the provisions of this Article.
  - F. A grievance may be withdrawn without prejudice at any level.
  - G. The Board agrees not to interfere, restrain, or coerce any unit member because of his/ner filing a grievance and/or his/her participation in any of the grievance proceedings.
  - H. Whenever possible grievance meetings shall be scheduled so as not to interfere with professional responsibilities of individuals involved. If it is necessary to meet with the employer during working hours, the grievant, one (1) Association representative who is a member of the bargaining unit, and necessary witnesses may attend without loss of time or compensation for such meetings.

#### 10.03 Disposition of Grievances

A. Informal Adjustments - Immediate Supervisor

Whenever possible, unit member(s) and the Association shall first attempt in good faith to adjust their grievances with the immediate supervisor or within the College's administrative structure up to the level of the President of the College or his/her designee.

**B. Formal Adjustments** 

All complaints shall be filed on a standard form. If the grievances involves an action by the Board or a matter of





general system-wide applicability, the complaint shall be filed at Step Two within thirty (30) calendar days after the grievant knows, or should have known, of the alleged acts or condition on which the grievance is based; provided, however, that the President of the Association or his/her designee must first certify in writing that the grievance is of general system-wide applicability. Within ten (10) calendar days of receipt of said certification, the Chancellor or his/her designee shall determine whether or not the grievance is of system-wide applicability or applies to the Board. If the Chanceller or his/her designee determines that the grievance is of system-wide applicability or applies to the Board, he/she shall meet and confer with the grievant for the purpose of resolving the grievance and shall render a written decision within fifty (50) calendar days after receipt of certification. If determined not to be of system-wide applicability or applicable to the Board, the unit member(s) or the Association shall have fourteen (14) called a day from receipt of the hearing officer's determination in which to file the

If determined not to be of system-wide applicability or applicable to the Board, the unit member(s) or the Association shall have fourteen (14) called a days from receipt of the hearing officer's determination in which to file the grievance at Step One. The Clancellor or his/her designee may upon mutual consent consolidate at Step Two multiple grievances which involve the same issue. All other grievances shall be processed in accordance with the following procedures:

### 10.04 Step One - President of the College or His/Her Designee

Within thirty (30) calendar days after the grievant knows or should have known of the alleged act or omission on which the grievance is based, the grievant shall present the complaint and all evidence upon which he/she relies or intends to rely as supporting his/her claim for relief to the President of the College or his/her designee. The Complaint shall state all issues and contract violations upon which the grievance is based. No further issues or contract violations may be added subsequent to filing the Step One complaint; provided, however, that the grievant shall "ave an opportunity to provide rebuttal evidence. The President of the College or his/her designee shall meet with the grievant for the purpose of resolving the grievance and shall, within thirty (30) calendar days of receipt of the written complaint and evidence, render his/her decision and reasons therefor in writing to the grievant.

### 10.05 Step Two - Chancellor

A. If the grievance is not resolved at Step One or the written decision of the President of the College or his/her designee is not rendered within the time specified, the grievant may then appeal the decision in writing to the Chancellor or his/her designee (on a standard form) with a copy to the President of the College or his/her designee. The appeal shall be filed within ten (10) calendar days of receipt of the President's or his/her designee's decision or the end of the time specified in Step One for said decision, whichever is sooner. Upon receipt of the appeal, the President of the College or his/her designee shall submit to the Chancellor or his/her designee the grievance form filed at Step One, all evidence introduced to date, and copies of all decisions.

Failure to so file with the Chancellor or his/her designee within the time specified shall be deemed to be acceptance of the decision rendered at Step One. The Chancellor or his/her designee shall meet with the grievant for the purpose of conducting a hearing and shall render a written decision within forty (40) calendar days of receipt of the appeal.

# 10.06 Step Three - Arbitration

- A. Within forty (40) calendardays after receipt of the decision of the Chancellor or his/her designee or where no decision has been issued within forty (40) calendar days after it was due, arbitration of a grievance may be initiated in accordance with the following provisions:
  - The Association shall have the exclusive right to initiate arbitration of a grievance. The decision or award of the
    arb trator shall be final and binding for the Association, the employee and the Board in accordance with applicable provisions of state law.
  - 2. The Association may initiate arbitration of a grievance only if the resolution of the grievance has been duly authorized by the Association and so certified by the President of the Association or his/her designee.
  - 3. The Association shall initiate arbitration by filing a demand for arbitration with the American Arbitration Association and with the Chancellor or his/her designee within forty (40) calendar days of receipt of the decision of the President of the College or where no decision has been issued within forty (40) calendar days after it was due.
  - 4. Such arbitration shall be conducted in accordance with the rules and regulations of the American Arbitration Association in effect on the date of said submission, unless otherwise provided herein; provided, however, that the jurisdiction of the arbitrator to inquire into any issue or to render any award is governed solely by the provisions of this Agreement.

#### B. Jurisdiction of the Artitrator

# Limit of the Arbitrator's Jurisdiction

Subject to the provisions of this Agreement, the arbitrator shall have no authority or jurisdiction to:

- 1. Arbitrate such portion of any grievance which is removed from the jurisdiction of the Board by the express terms of this Agreement.
- 2. Add to, alter, or amend any terms or conditions of this Agreement.
- 3. To inquire into or arbitrate any issue not presented by the original complaint.



# C. Authority of the Arbitrator

Unless otherwise provided in this Agreement, the arbitrator shall have the authority to make a final and binding award on any dispute concerning the interpretation or application of this Agreement. The arbitrator's authority in matters which are arbitrable is limited to a determination as to whether the provisions set forth in this Agreement were violated; provided that in matters of professional judgment, the arbitrator shall determine whether the application of such to the grievant has been arbitrary, capricious or unreasonable. Beyond such determination, the arbitrator shall be without power, right or authority to make a decision or to substitute his/her judgment for that of the Board or its representatives, except as otherwise provided in this Agreement.

The arbitrator shall have no authority to arbitrate:

- 1. Any incident which occurred or failed to occur prior to the ratification date of this Agreement.
- 2. Workload and working conditions.
- 3. The failure or refusal by the Board to renew the contract of or to reappoint a unit member in the first (1st) three (3) years of regular appointments.
- 4. Affirmative action.
- 5. Basis for retrenchment.

#### D. Award of Arbitrator

If the arbitrator determines no express provision of this Agreement has been breached in its application to the grievant as claimed, he/she shall dismiss the grievance. If the arbitrator determines that his Agreement has been so breached, he/she may, subject to the provisions of this Article and except as hereinafter provided, provide an appropriate remedy for the breach; provided, however, that in making any monetary award, the arbitrator shall only provide compensation for actual damages directly attributable to such breach, and shall in no event make any award by way of penal damages.

#### E. Dismissal and Retrenchment

If a unit member's employment is discontinued as a result of dismissal or retrenchment, and the arbitrator determines based on clear and convincing evidence that the decision of the President of the College or his/her designee was arbitrary, capricious or unreasonable, he/she shall remand the matter for reassessment by the President of the College or his/her designee of the original decision but shall not have the result or effect of granting any binding award. After remand, the President of the College or his/her designee shall have thirty (30) days to render a new decision. If the grievant believes the decision of the President of the College or his/her designee on remand is arbitrary, capricious or unreasonable, he/she may file an appeal to the original arbitrator in accordance with the provisions of this Article. On appeal, if the arbitrator determines based on clear and convincing evidence, that the decision of the President of the College or his/her designee is arbitrary, capricious or unreasonable, he/she shall have the power to make the grievants whole; provided, however, that any monetary award shall be reduced by way of mitigation by an amount equal to the total income received by the grievant during the period for which mor ary damages are so awarded. If the arbitrator determines that the Step One or Two decision was not issued within the applicable time limits contained in this Article, he/she may in his/her first decision provide a binding award in accordance with the foregoing procedures.

#### F. Promotion

- 1. It is expressly understood that decisions regarding promotion and initial placement are made pursuant to the exercise of professional judgment. Whenever a unit member files a grievance under this Article, the act or decision grieved shall be reviewable by an arbitrator as an exercise of professional judgment and whether the application to the grievant was arbitrary, capicious or unreasonable.
- 2. A unit member shall submit a complaint at Step One to the President of the College or his/her designee within thirty (30) calendar days after he/she knew or should have known of the act or conditions upon which the grievance is based. The President of the College or his/her designee shall meet with the grievant and shall render his/her decision and the reasons therefor in writing to the grievant within thirty (30) calendar days of receipt of the complaint.
  - If the grievance is not resolved at Step One or if the written decision of the President of the College is not rendered within the time specified, the grievant may appeal to Step Two. The appeal shall be filed within ten (10) calendar days from the date it was due. The Chancellor or his/her designee shall meet with the grievant and shall render a written decision within forty (40) calendar days of receipt of the appeal.
- 3. If the grievance is not resolved at Step Two or if the written decision of the Board is not rendered within the time specified, arbitration may be initiated in accordance with Article X. If arbitration is so initiated and the arbitrator determines based on clear and convincing evidence that the decision of the President of the College or his/her designee was arbitrary, capricious or unreasonable, he/she shall remand the matter for reassessment by the President of the College or his/her designee of the original decision but shall not have the result or effect of granting any binding award. After remand, the President of the College shall have thirty (30) days to render a decision. If the grievant believes the decision of the President of the College or his/her designee on remand is arbitrary, capricious or unreasonable, he/she may file an appeal to the original arbitrator in accordance with the provisions of this Article. On appeal, if the arbitrator determines based on clear and convincing evidence that



23

the decision of the President of the College or his/her designee is arbitrary, capricious or unreasonable, he/she may provide an affirmative and binding award; provided, however, that if so awarded, the grievant shall be promoted to the next available position at appropriate rank as determined by the President of the College; provided, however, that such determination shall not be grievable or arbitrable and provided further that any monetary award shall begin to accrue one (1) year after the date of issuance of the first (1 st) arbitration award. If the arbitrator determines that the Step One or Step Two decision was not issued within the applicable time limits contained in this Article, he/she may in his/her first (1 st) decision provide a binding award in accordance with the foregoing conditions.

- 4. A unit member shall have the right to appeal to the President of the College an alleged violation of the procedural requirements by the Division Chairperson and/or the appropriate Dean, provided that such an appeal is filed in writing with the President of the College within seven (7) calendar days of the date of recommendation of the Division Chairperson and/or the appropriate Dean. The President of the College or his/her designee shall investigate any such allegation, and the President of the College shall, if he/she finds a procedural violation prejudicial to the individual, take remedial action or give weight to the violation in making his/her recommendation to the Board. These provisions shall be the exclusive remedy for an alleged violation of the contractual procedures by the appropriate Dean or the appropriate Division Chairperson. Any such allegation shall not be subject to grievance unless the President of the College has failed to take remedial action or give weight to the violation.
- G. The granting or failure to grant tenure shall be arbitrable but any award is not binding.
- HL Notwithstanding any rule of the American Arbitration Association to the contrary, in making his/her decision, the arbitrator shall apply the express provisions of this Agreement and shall not alter, amend, extend or revise any term or condition hereof.
- I. All fees and expenses of the arbitrator shall be divided equally between the parties. Each party shall bear the cost of preparing and presenting its own case.

#### ARTICLE XI - APPOINTMENT AND REAPPOINTMENT - TENURE

# 11.01 Appointments

# A. General

Except as expressly provided for by Board policy, the authority to offer and make effective appointments and reappointments rests with the Board and no employee or agent of the Board or the Colleges may expressly or by implication make effective any appointment or reappointment.

- 1. Each unit member's reappointment to a bargaining unit position shall be in accordance with the provisions of this Article.
- 2. All appointments and reappointments to unit positions at a College shall be made by the Board upon recommendation of the President of the College. Notification of such appointment shall be in writing as provided in Article XXI.
- Unit members who are temporarily shifted to other than state appropriated funds shall, during and upon their return to state appropriated funds, retain all rights and benefits they would have had if they had not been transferred.

### a. Types of Appointment

- 1. A regular appointment shall be one which creates an interest in employment for a specified term of one (1) year or less. All employees holding regular appointments small be subject to non-reappointment without cause.
- 2. A tenured appointment may be offered only to those faculty members in the ranks of Assistant Professor, Associate Professor and Professor and to professional staff members in the classification of Professional Staff II, III and IV.

#### 11.02 Reappointment and Non-Reappointment of Regular Appointments

- A. The non-reappointment of a faculty or professional staff member shall be in accordance with the following procedures:
  - 1. Termination at the end of the first (1st) through third (3rd) years of service; notice provided by February 15 of that year of service;
  - 2. Termination at the end of the fourth (4th) year of service or later; notice provided by October 15 of the year of termination.
- B. Non-reappointment of a unit member on a regular appointment in the first (1st) three (3) years shall be without cause, except for written notice requirements required in Section 11.02. There shall be no necessity to provide reasons, nor shall the decision be subject to the grievance procedure. For the fourth (4th) year of regular appointment, the notice of non-reappointment of non-tenured unit members shall be subject to the exercise of professional judgment. Such notice shall contain a statement of reasons in writing. Termination in the fourth (4th) year or later of unit members whos source remuneration is paid from non-state appropriated funds shall be accompanied by a statement of reasons; provided, however, that no statement shall be required if non-reappointment is due to insuffi-



cient funds or to the terms and conditions of the non-state appropriated funding source and provided furth in that said non-reappointment shall be grievable to step two and shall be subject but not arbitrable to the exercise of professional judgment and whether application to the grievant was arbitrary, capricious or unreasonable

- C. Termination of a unit member in his/her fifth (5th) regular appointment or later shall be for just cause.
- D. Notice shall be in writing given by the President of the College or his/her designee. Any unit member holding a regular appointment who does not receive such notice shall be entitled to inquire of the President of the College as to the failure to give notice. The President of the College or his/her designee shall respond to the inquiry within fourteen (14) calendar days.
- E. Failure to give notice of reappointment shall not prohibit reappointment of the unit member. Failure to give proper notice shall constitute reappointment of the unit member for one (1) academic or calendar year, whichever is applicable, but shall not thereby entitle the unit member concerned to academic tenure, to any further appointment with or without academic tenure, or to further notice of non-reappointment.
- F. Each unit member holding a regular appointment shall notify the President of the College in writing of his/her intent to accept or reject a reappointment within thirty (30) days after receipt of notice of reappointment. Failure by the unit member to notify the President of the College of acceptance shall constitute a rejection of reappointment.

## 11.03 Reappointment of Unit Member

A. Tenure may be granted by the Board on recommendation of the President of the College and shall relate only to the specific College and not the entire Community College System.

# B. Eligibility - Tenure Contracts

- 1. The unit member must have served at least six (6) full years as a unit nember, at least three (3) of which have been in his/her current job function.
- 2. The unit member must have received other than unsatisfactory on his/her most recent summary evaluation.

#### C. Procedures

- 1. There shall be established by November 21 of each year a Unit Personnel Practices Committee elected by members of the unit. The size and composition of this Committee shall be determined by the President of the College or his/her designee and shall reflect, whenever possible, the ratio of teaching faculty to professional staff in the unit; provided, however, that a unit member shall not serve on the Committee during the year he/she is a candidate for tenure. The Committee shall elect a chairperson.
- 2. The appropriate Dean(s) shall notify all eligible unit members by October 1 of their sixth (6th) or later regular appointment that they may be considered for tenure during the academic year if otherwise eligible; provided that copies shall be forwarded to the immediate supervisor and to the Unit Personnel Practices Committee.
- 3. The immediate supervisor and the Unit Personnel Practices Committee shall review all relevant material within the individual's official personnel file and shall forward recommendations for either tenure or a one (1) year terminal appointment to the appropriate Dean(s) by March 15th.
- 4. The appropriate Dean(s) shall review the recommendations of the immediate supervisor and the Unit Personnel Practices Committee; and whenever practicable consult with the immediate supervisor and the Chair of said Committee; and thereafter shall forward his/her recommendations to the President of the College by April 15.
- 5. The President of the College shall review the recommendations of the appropriate Dean(s) and shall forward his/her recommendation to the Board by May 1. The unit member shall be notified of the President's recommendation within twenty (20) calendar days thereafter.
- 6. A recommendation for a one (1) year terminal contract shall be accompanied by a state nent of reasons.

# 11.04 Protection of Bargaining Unit Upon Appointment of Returning Administrators

Any unit member who assumes administrative duties outside the bargaining unit and subsequently returns to faculty or professional staff unit member status shall, at the option of the President of the College or his/her designee either:

1) retain all previously accrued unit seniority and be placed upon the salary schedule at the step and rank he/she held prior to assuming administrative appointment, as adjusted by the provisions of any collective bargaining agreement between the Board and the Association: or 2) shall relinquish all prior accrued unit seniority and be placed upon the salary schedule at a step and rank to be determined by the President of the College or his/her designee; provided, however, that a unit member who accepts an acting administrative appointment shall upon return to the unit retain all prior accrued unit seniority and be placed upon the salary schedule at the rank and step he/she held prior to assuming administrative appointment as adjusted by the provisions of any collective bargaining agreement between the Board and the Association. It is agreed that the return of an administrator to the unit under either of the options stated above will have no adverse effect on present unit members.

11.05 Subsequent to July 31, 1981, the faculty lines of Professor, Associate Professor, and Professional Staff Classifications III and IV shall be occupied only by full-time faculty and/or professional staff members; provided, however, that the foregoing shall not require displacement from an upper faculty position of any incumbent administrator who occupies such position prior to August 1, 1981. Nothing contained in this Article shall be construed to



impair the rights of the parties with regard to those challenges to returning administrators pending on the date of ratification and approval of this Agreement.

# ARTICLE XII - WORKLOAD. WORK ASSIGNMENT AND WORKING CONDITIONS

#### 12.01 General Provision

Unit members shall be given a contract stating the subjects or functions to which the unit member is to be assigned. There shall be included a job description which shall define the duties to be performed.

#### 12.02 Work Assignment

- A. The President of the College or his/her designee shall consider as advisory written notice from each faculty member and/or department chairperson/curriculum coordinator as to the preferred courses to be taught. Such written notice must be received by the President of the College or his/her designee at least one (1) month prior to the date the College must notify each faculty member as contained herein. The President of the College or his/her designee shall notify in writing each faculty member of his/her tentative courses and number of sections to be taught by the end of the April payroll period for the Fall semester and by the end of the November payroll period for the Spring semester.
- B. The President of the College or his/her designee shall consider as advisory, written notices from each faculty member and/or department chairperson/curriculum coordinator as to the preferred class schedule. Such written notice must be received at least one (1) month prior to the date the College must notify each faculty member as contained herein. The President of the College or his/her designee shall notify each faculty member in writing of his/her tentative class schedule by the end of the May and December payroll period for the succeeding Fall and Spring terms, respectively. The class schedule indicates the days and times each week that the faculty member has contact responsibility with students and does not refer to the academic calendar defined elsewhere.
- C. Whenever changes in the tentative class schedule or subject matter are necessary, the President of the College or his/her designee shall notify the affected faculty member(s) in writing at his/her official residence or a temporary address supplied by the faculty member on leave. The change in the tentative class schedule or subject matter preparation shall be consistent with Section 12.01.

#### 12.03 Workload of Faculty

- A. Faculty workload shall consist of:
  - 1. Instructional workload, which includes:
    - a. teaching in both traditional and non-traditional learning modes;
    - b. instructional preparation;
    - c. assessment of student performance.
  - 2. Non-instructional workload shall consist of:
    - a. student assistance/advisement:
    - b. college service, which includes:
      - 1. serving as advisor to student activities;
      - 2. serving on governance, ad hoc, college standing committees, system- wide task forces or committees, or labor-management committees;
      - preparing grant proposals;
      - 4. participating in college, division, department or other related college meetings and/or activities;
      - participation in the improvement and development of academic programs and resources, including recruitment.
    - c. college recognized community service, provided that such service is not compensated by an outside funding source:
    - d. professional development activities, which include:
      - 1. related graduate study;
      - 2. related in-service training;
      - 3. research and other College recognized contributions to a faculty member's area of competence;
      - participation in College recognized professional associations, including teachers' organizations.

#### **B.** Instructional Workload

- 1. The standard faculty instructional workload shall be twenty-four (24) units of instruction per year with two (2) preparations per semester. This standard shall be achieved by the end of the 1983-86 collective bargaining agreement. A unit of instruction shall mean per week:
  - s. one (1) fifty (50) minute lecture or seminar;
  - b. one and one half (1 1/2) hours of laboratory instruction;



- c. two (2) hours of clinical instruction:
- d. two (2) hours of individualized instruction or other non-traditional modes of instruction:
- e. where team teaching involves multiple faculty, each faculty member shall be allotted instructional workload credit in accordance with B.1. a, b, c and d above, except that it shall be based on one-half(½) the course units as described above, regardless of the number of faculty involved in teaching, during the initial implementation of a team structure for a course. This shall also apply to an individual teacher who joins an existing team for the first time. Thereafter, credit shall be based in direct proportion to the percentage of total course time that he/she teaches.

The President of the College or his/her designee shall adjust the ratios in b, c, d and e if in his/her judgment, after consultation with the unit member the difficulty in instruction, the difficulty and duration of preparation, or the provision of assistance in instruction or such other related instructional, clinical or programmatic factors ment such adjustment; provided, however, that if a unit member after such consultation continues to believe he/she is aggrieved by the decision of the President of the College or his/her designee, he/she may appeal to the Workload Grievance Panel in accordance with the provisions governing said Panel. The President of the College or his/her designee may upon mutual agreement assign a faculty member non-instructional activities such as course, program or curriculum development, professional development activities or administrative duties; provided that he/she is qualified to perform such activities in lieu of a proportional number of instructional units.

2. Faculty members shall not be expected to generate more than 420 student credit hours per semester within the applicable workload requirement or more than 300 student credit hours per semester for instruction of English Composition, English as a Second Language, and remedial and/or developmental courses; to be determined by the number of students enrolled at the end of add/drop period; provided that if the total number of student credit hours generated exceeds these limits, there shall be a proportional adjustment in the succeeding semesters, consistent with the provisions of 12.03B.

It is expressly agreed that during the term of this Agreement, the instructional and non-instructional workload of a faculty member shall be consistent with the established and continuous past practice of the applicable department, program or work area; provided, however, that a faculty member's workload shall not be increased or decreased beyond his/her 1979/80 workload until the implementation of the new standard. It is agreed by the parties that if a faculty member agrees to an increase in the number of course preparations beyond his/her 1979/80 course preparations, the President of the College or his/her designee and the faculty member shall mutually agree as to the corresponding decrease in advisees, and/or office hours and/or college service. New faculty hired shall be assigned an instructional workload consistent with the existing workload assignments of other faculty members within their respective academic department, work area or division. It is agreed by the parties that whenever a dispute in the instructional and non-instructional workload exists, the standard to be used shall be the workload article of the 1977-80 Collective Bargaining Agreement.

#### C. Non-Instructional Workload

#### 1. Student Assistance/Advancement

A faculty member shall be responsible for advising 20-25 students assigned to him/her by the President of the College or his/her designee.

It is the agreement of the parties that it is not the intent of this Article to allow a unit member to schedule in advance student advisement time during his/her office hour schedule. It is agreed, however, that if a student advisee appears of his/her own volition during a unit member's office hours, a unit member may utilize that time for student advisement responsibilities.

It is further understood that the administration has the obligation to assist the unit member in contacting student advisees and that the administration cannot determine the scheduling of this advisement time.

Each faculty member shall hold a group meeting with his/her advisees at the beginning of the academic year and shall be available to meet with each advisee at least once each semester and at such other reasonable times as requested by the advisee.

If in the judgment of the President of the College or his/her designee after consultation with the faculty member the number of students should be more than or fewer than 20-25, the faculty member's non-instructional workload may be adjusted.

# 2. Office Hours

During the academic year, faculty members shall maintain at least five (5) posted office hours per week on at least four (4) days. Office hours shall be scheduled in accordance with the following procedure

- a. A faculty member shall submit a preferred office hours' schedule to the President of the College or his/her designee prior to the commencement of the academic year.
- b. The President of the College or his/her designee shall either approve or disapprove the schedule.
- c. In the event the President of the College or his/her designee shall disapprove the schedule, the faculty member shall be given an opportunity to review and discuss their respective positions with the President of the College or his/her designee.



d. In the event the faculty member believes he/she is aggreed by the decision of the President of the College or his/her designee, he/she may appeal to the Workload Grievance Panel in accordance with the procedures outlined in Article 12.06.

Such schedule shall be posted by faculty members by the end of the first (1st) week of classes.

#### 3. College Service

In addition to posted office hours, a faculty member shall be on campus and available on any day during the academin year at such times as required by the President of the College or his/her designee to participate in scheduled activities consistent with 12.03 A2b.

#### 4. Community Service and Professional Development

A faculty member shall be responsible for such community service and professional development activities as mutually agreed by the President of the College or his/her designee and the faculty member.

#### D. Professional Days

The President of the College or his/her designee may assign to any faculty member up to seven (7) days during the academ. year for scheduled orientation and registration programs, commencement and convocation activities and such other College-sponsored activities he/she deems appropriate. Professional days shall be assigned no earlier than three(3) days prior to the first (1st) day of classes in the Fall semester and no earlier than four (4) days prior to the first (1st) day of classes in the Spring semester. Such duties and responsibilities may be assigned to individual faculty members in blocks of one-half (1/2) day or more; provided that a faculty member assigned more than one-half (1/2) day shall be credited with one (1) full day; and provided further that a faculty member shall not be required to discharge these responsibilities over more than fourteen (14) days during the academic year.

#### E. Academic Year

The academic calendar shall be a period of time encompassing two (2) semesters, beginning no earlier than September 1 and ending no later than May 31, exclusive of Commencement. Classes shall begin no earlier than the day after Labor Day and end no later than December 24 for the Fall semester; provided, however, that the President of the College and the President of the local Association Chapter may by mutual agreement set other dates for the academic calendar, including beginning classes prior to the day after Labor Day. When first (1st) semester classes begin after September 5, the President of the College may, after consultation with the local MACER, increase the class time from fifty (50) minutes to fifty-five (55) minutes for that semester consistent with the established past practice. There shall be at least a four (4) week recess between semesters.

# F. Outside Employment

Full-time employment by the Board shall be considered the primary employment of each faculty member. Outside employment is work for which compensation is received and which is not the normal duties and responsibilities assigned to a faculty member as an employee of the Board. Outside employment which requires a faculty member's absence from the College during his/her normally scheduled working hours is presumed to interfere with the performance of the duties and responsibilities of that faculty member and is hereby prohibited.

#### 12.04 Workload of Professional Staff

- A. Professional staff workload shall consist of:
  - 1. professional duties as contained in a letter of appointment and position description;
  - 2. student advisement, if assigned;
  - 3. college service, which includes:
    - a. service as advisor for college-approved student activities;
    - b. service on governance, ad hoc, college standing committees, system-wide task forces or committees; or labor-management committees;
    - c. preparing grant proposals;
    - d. participating in college, division, department or other related college meetings and/or activities as the President of the College or his/her designee may deem appropriate;
    - e. participation in the improvement and development of academic programs and resources, including recruitment.
  - 4. College recognized community service, provided that such service is not compensated by an outside funding source, and professional development; provided that such community service and professional development activities are mutually agreed to by the President of the College or his/her designee and the professional staff member.
  - 5. Assigned instructional responsibilities and related preparation; provided that customarily professional staff members shall not be assigned traditional academic discipline responsibilities; and provided further that no professional staff member shall be required to teach subject matter which is beyond the scope of his/her academic and/or professional competency and experience as determined by the President of the College or his/her designee.



#### B. Work Year

All professional staff employees shall work a twelve (12) month year.

#### C. Work Schedule

The President of the College or his/her designee shall consider as advisory written notice from the professional staff member as to his/her preferred work assignment if received on or before June 1 st prior to the fiscal [cart s] covered by this Agreement. The President of the College or his/her designee shall notify the professional staff member in writing of his/her work assignment no later than July 1. Such work assignment shall be consistent with the needs of the College.

- 1. The workload of each professional staff member shall consist of such duties and responsibilities in accordance with his/her respective work schedules as may be assigned by the President of the College or his/her designee.
- 2. During the term of this Agreement, the normal work week for professional staff members shall be 37 1/2 hours per week.
- 3. During the term of this Agreement, the normal work day shall not exceed 7 1/2 hours, excluding lunch
- 4. Whenever a professional staff member is required to work more than 37 1/2 hours in a particular week, he/she shall receive compensatory time on an hour-to-hour basis within the next sixty (60) working days, subject to mutual agreement between the professional staff member and the President of the College or his/her designee.
- 5. The customary work week for professional staff members shall be Monday through Friday, within the hours of 8:00 a.m. to 5:00 p.m. but in no case shall a professional staff member be require. I to work more than five (5) days in any seven (7) consecutive day period; provided that a professional staff member who is assigned to teach may request a revised work schedule.
- 6. Upon request of a professional staff member, the President of the College or his/her designee may assign up to three (3) days during the fiscal year in half-day segments or greater for participation in off-campus activities outside those assigned under the provisions of 12.04; one of these days shall be granted for the day following the Thanksgiving holiday. Such requests shall not be unreasonably denied.
- 7. All time spent in student advisement (if assigned), in attendance at conferences, meetings and student activities when such attendance is required during the work day shall be counted in the professional staff member's weekly and daily hours.
- 8. Upon request of a professional staff member the President of the College or his/her designee may, where practicable, grant a flexible schedule; provided, however, that the President's decision shall be final and nongrievable.
- 9. Upon request of a professional staff member, the President of the College or his/her designee may, where practicable, grant a ten(10) month year option at 10/12 of pay and benefits; provided, however, that the President's decision shall be nongrievable.

#### 12.05 Workload Reduction

- A. The President of the Association shall notify the Chancellor or his/her designee and the Presidents of the affected Colleges or their designees of the names of unit members who have been certified to have been approved by the President of the Association for a workload reduction. The Board shall provide fifteen (15) course sections per semester without cost to the Association. The Association shall pay for an additional ten (10) course sections per semester at the usual rate received by part-time faculty at each affected College.
  - Unit members who receive a workload reduction shall teach at least one (1) course per semester. No College shall be required to provide more than three (3) sections of unpaid workload reduction in any semester.
- B. A department chairperson/curriculum coordinator (program and work area/discipline) shall be granted a reduction in instructional workload; provided that the reduction shall not exceed eighty (80) percent nor be less than twenty (20) percent. The amount of workload reduction shall be dependent upon the duties performed by the department chairperson/curriculum coordinator and shall be solely determined by the President of the College or his/her designee. Such reduction shall not be made in an arbitrary, capricious or unreasonable manner.

#### 12.06 Workload Grievance Panel

- A. If a unit member believes he/she is aggrieved by the application or interpretation of this Article, he/she may file a grievance in accordance with the provisions of Article X within thirty (30) calendar days after he/she should have known of the act or condition upon which the grievance is based.
- B. The President of the College or his/her designee shall investigate the grievance and respond in writing within thirty (30) calendar days after receipt of the grievance. No hearing is required. The decision of the President of the College shall be final except as hereinafter provided.
- C. If the unit member believes the decision of the President of the College is arbitrary and capricious and unreasonable, he/she may appeal to the Workload Grievance Panel, which shall be composed of three (3) members selected in the following manner:



- 1. a President designated by the Chairperson of the Presidents' Council;
- 2. the President of the Association or his/her designee;
- 3. the Chancellor or his/her designee;
- 4. neither the President of the College whose decision is being appealed nor the affected faculty member shall be eligible to serve on the grievance panel
- D. The appeal to the Panel shall be filed with the Chairperson designated by the Board within ten (10) calendar days after receipt of the Step One decision.
- E. Hearings shall be at the call of the Panel. The Panel shall respond to the appeal within forty (40) calendar days after receipt of the appeal. The decision of the Panel shall be final and binding upon all parties.
- F. All time limitations may be waived by agreement of the interested parties.

# ARTICLE XIII - EVALUATION

# 13.01 Evaluation Objectives

The evaluation of unit members is directed to the following objectives:

- 1. Assessment of the professional performance of the unit member.
- 2. Improvement of performance and quality of instruction.
- 3. To provide a basis upon which decisions shall be made concerning the reappointment, promotion, performance-based salary adjustments, tenure, sabbatical and professional leaves, and termination, dismissal and discipline of a unit member.

# 13.02 Evaluation of Faculty Members

# A. Faculty Evaluation Criteria

Evaluation of faculty members shall be uniformly applied and based upon total performance with primary emphasis on teaching effectiveness, including consideration of the following criteria:

- 1. Development and improvement of instructional methodology;
- 2. Establishment of course objectives, course content and instructional activities;
- 3. Establishment of appropriate and fair procedures and instruments for student evaluation;
- 4. Student advising competency and accessibility; including the observance of established office hours;
- 5. College service, including
  - a serving as advisor to student activities; and/or
  - b. serving on governance, ad hoc, college standing committees, system-wide task forces or committees, and labor/management committees; and/or
  - c. preparing grant proposals; and/or
  - d. participating in college, division, department or other related College meetings and/or activities; and/or
  - e. participation in the improvement and development of academic programs and resources, including recruitment.

#### **B.** Procedure

The procedure for evaluating faculty members shall consist of five (5) processes: (1) student evaluation; (2) course materials evaluation; (3) classroom observation evaluation; (4) student advisement and college service evaluation; and (5) summary evaluation.

# 1. Student Evaluation

- a. The President of the College or his/her designee shall be responsible for administering the student evaluation process.
- b. Student evaluation packets for each class containing instruments and instructions shall be distributed to each faculty member by November 15.
- c. It is expressly agreed that the faculty member being evaluated shall not be present in the classroom when the student evaluation is being administered and that all instruction to students with regard to such student evaluation shall be included in writing on the instrument, provided further that the designated unit or non-unit professional shall return the student evaluation directly to the President of the College or his/her designee. The administering of the student evaluation shall be the responsibility of the President of the College or his/her designee who shall determine who among unit or non-unit professionals shall administer such student evaluation. Student evaluations shall be valid only if signed by the student; provided, however, that faculty members shall not be entitled to the identity of the student responding unless such student evaluation is used as a basis for dismissal or other disciplinary action and such will be communicated to the students.
- d. The data from the student evaluation shall be tabulated and copies sent to the President of the College or his/



her designee. The raw data shall be retained by the College for a period of one (1) year during which time the faculty member shall have access thereto upon written request.

- e. The President of the College or his/her designee shall review the tabulated data and shall prepare a student evaluation summary and forward a copy to the faculty member by December 15.
- f. The faculty member shall have seven (7) working days in which to respond to such summary evaluation.

#### 2. Course Materials

- a. The faculty member shall forward to the Presider tof the College or his/her designee no later than the end of the second (2nd) week of classes in each semester the following materials for each course:
  - (1) Course outline:
  - (2) Instructional objectives and requirements students must meet to complete the course; and
  - (3) Description of procedures to be used in evaluating students.

The confidentiality of these materials shall be maintained.

- b. Upon request of the affected unit member the President of the College or his/her designee shall consider advisory input from the appropriate Department Chairperson/Curriculum Coordinator relating to the unit member's course materials.
- c. The President of the College or his/her designee shall review the course materials and evaluate them. In reviewing said materials the President of the College or his/her designee may seek additional information from other sources he/she may deem appropriate, including but not limited to Department Chairpersons.
- d. The President of the College or his/her designee shall return all course materials to the faculty member and a copy of the evaluation of the course materials by the end of the fifth (5th) week of classes in each semester and, if requested, shall meet and confer with the faculty member to discuss the reasons therefor.
- e. The faculty member shall have seven (7) working days in which to respond to the evaluation.

#### 3. Classroom Observation

- a. The President of the College or his/her designee shall conduct at least one (1) classroom observation per academic year.
- b. Each faculty member shall be advised of a two (2) week period during which the President of the College or his/her designee shall conduct classroom observation, provided that each faculty member shall be given at least twenty-four (24) hours' notice of the date of classroom observation; provided, however, that a one (1) time postponement may be requested by a faculty member, if requested twenty-four (24) hours in advance. Each faculty member may submit to the President of the College or his/her designee supplementary course materials regarding the planned classroom activities during said two (2) week period.
- c. Pre- and Post-Observation Conferences: For each classroom observation conducted by the President of the College or his/her designee there shall upon request of the affected unit member occur a pre-observation and post-observation conference between the unit member and the President of the College or his/her designee. The post-observation conference shall occur not later than two (2) weeks after the classroom observation unless otherwise mutually agreed.
- d. The President of the College or his/her designee shall within fourteen (14) days of observation prepare an evaluation of the classroom observation in accordance with the criteria set forth in 13.02A and shall forward a copy to the faculty member.
- e. The faculty member shall have seven (7) working days in which to respond to the evaluation.

# 4. Student Advisement and College Service

- a. The President of the College or his/her designee shall evaluate the performance of the unit member's assigned duties and responsibilities consistent with the criteria established in this Article. The procedure for evaluating performance shall be as hereinafter described.
- b. The faculty member shall submit documentation relating to performance of his/her duties and responsibilities, including but not limited to a log of student advisement and college service activities to the President of the College or his/her designee by the last day of classes in each semester.
- c. The President of the College or his/her designee shall review the document(s) and may seek additional information from other sources he/she deems appropriate. The President of the College or his/her designee shall prepare an evaluation of student advisement and college service performed since the last such evaluation and shall forward a copy of the evaluation and, if requested, shall meet and confer with the faculty member to discuss the reasons therefor.
- d. The faculty member shall have seven (7) working days in which to respond to the evaluation.

# 5. Summary Evaluation

The President of the College or his/her designee shall after receipt of all the foregoing evaluation components and prior to February 1 develop a summary evaluation of each faculty member and shall consider only the foregoing evaluation components. The results of the components shall be applied in a uniform manner and shall be assigned the following weights:



Student Evaluation	30%
Course Materials Evaluation	25%
Classroom Observation Evaluation	25%
Student Advisement Evaluation	10%
College Service Evaluation	10%

A copy of the summary evaluation shall be forwarded to the faculty member by February 1, upon receipt of which the faculty member shall have seven (7, working days to respond to the summary evaluation.

# 13 03 Evaluation of Professional Staff Members

#### A. Professional Staff Evaluation Criteria

Evaluation of professional staff shall be uniformly applied and based on total job performance, including the following criteria:

# 1. Professional performance including

- a. conformance with assigned workload as established by the appropriate administrator in accordance with Article XII; and
- b. the rendering of effective assistance to students, faculty and staff of the College, individually and/or collectively in accordance with Article XII.
- 2. Student advising competency and accessibility, if appropriate.
- 3. College service, including
  - a. serving as advisor to student activities; and/or
  - b. serving on governance, ad hoc, college standing committees, system-wide task forces or committees, or labor/management committees; and/or
  - c. preparing grant proposals; and/or
  - d. participating in college, division, department or other related college meetings and/or activities as the President of the College or his/her designee may deem appropriate; and/or
  - e. participation in the improvement and development of academic programs and resources, including recruitment.
- 4. Assigned instructional responsibilities and related preparation in accordance with Article XII

#### B. Procedure

The procedure for evaluating professional staff members shall consist of three (3) processes: (1) work performance evaluation, (2) student advisement and college service, and (3) summary evaluation.

#### 1. Evaluation of Work Performance

#### a. Pre-Evaluation Conference

By July 1 or the beginning of a professional staff member's appointment, the President of the College or his/her designee shall upon request of the affected professional staff member conduct a pre-evaluation conference for the purpose of developing his/her position description, which shall contain a list of duties and responsibilities and may include mutually agreeable work objectives, if appropriate. The President of the College or his/her designee shall upon request meet with a professional staff member during the evaluation year to discuss his/her progress,

b. The President of the College or his/her designee shall complete the work performance evaluation by January 15. The professional staff member shall receive a copy of the evaluation and shall have seven (7) working days in which to respond to the evaluation.

#### c. Post-Evaluation Conference

The President of the College or his/her designee shall upon request meet and confer with the professional staff member to discuss the reasons for his/her evaluation within fourteen (14) days following the completion of saic evaluation.

# 2. Evaluation of Student Advisement and College Service

- a. The professional staff member shall submit to the President of the College or his/her designee documentation relating to performance of his/her dutics and responsibilities, including but not limited to a log of student advisement, if appropriate, and college service activities by December 30 and May 30.
- b. The President of the College or his/her designee shall review the document(s) and prepare an evaluation of student advisement and college service. He/she may seek additional information from other sources. The President of the College or his/her designee shall forward a copy of the evaluation to the professional staff member by January 15 and shall if requested meet and confer with the professional staff member to discuss the reasons therefor.
- c. The professional staff member shall have seven (7) working days in which to respond to the evaluation.

#### 3. Summary Evaluation

The President of the College or his/her designee shall after receipt of all the foregoing evaluation components



and prior to February! develop a summary evaluation of each professional staff member and shall consider only the foregoing evaluation components. The results of the components shall be applied in a uniform manner and shall be assigned the following weights:

Work Performance 90% College Service 10%

# 13.04 Reallocation of Weights Assigned to Evaluation Components

The weights assigned to evaluation components shall be reallocated proportionally as determined by the President of the College or his/her designee, if the workload assignments of a unit member have been adjusted by assigning non-instructional activities to a faculty member pursuant to Article 12.03.

# 13.05 Written Reasons

Upon request, a unit member shall be provided with written reasons for his/her evaluation (s).

# ARTICLE XIV - CRITERIA AND PROCESS FOR PROMOTION OF FACULTY AND PROFESSIONAL STAFF UNIT MEMBERS

#### 14.01 Rank and Classification System

- A. There shall be four (4) ranks for faculty members:
  - 1. Instructor
  - 2. Assistant Professor
  - 3. Associate Professor
  - 4. Professor
- B. There shall be four (4) classifications for professional staff unit members:
  - 1. Professional Staff I
  - 2. Professional Staff II
  - 3. Professional Staff III
  - 4. Professional Staff IV

# 14.02 Eligibility for Promotion

- A. Faculty and professional staff unit members who meet the following minimum qualifications shall be considered for promotion to a higher faculty rank or professional staff classification.
  - 1. The unit member must have served at least two(2) years in his/her present faculty rank at his/her College as of September 15 or professional staff classification as of July 15.
  - 2. The unit member must have been deemed other than unsatisfactory on his/her most recent summary evaluation.
  - 3. Qualifications for promotion hereinafter set forth are not to be construed to limit the right of the recommending authorities to specify additional criteria when such criteria are customarily required for specialized or professional areas.

# Minimum Qualifications for Consideration for Promotion

#### **Faculty**

acuity			
Rank	Degree*	Time in Rank	Total Experience
Assistant Professor	Doctorate	2 years	4 years
	Master's +		
	15-30 gr. cr. hrs. Master's or	2 years	5 years
	equivalent	2 years	6 years
Associate Professor	Doctorate	2 years	6 years
	Master's +		
	15-30 gr. cr. hrs. Master's or	2 years	7 years
	equivalent	2 years	8 years
Professor	Doctorate Master's +	2 years	8 years
	15-30 gr. cr. hrs. Master's or	2 years	9 years
	equivalent	2 years	10 years
*Degrees must be from n	economic pathography and a second	<del>-</del>	20 ,02.0

\*Degrees must be from regionally accredited institutions



#### Computation of Total Experience

For purposes of determining total experience, prior experience shall be credited as follows:

- 1. One (1) year of credit for each year of teaching experience within the Community Colleges or at a community college outside the System;
- 2. One (1) year of credit for each year of prior full-time post-secondary teaching at other than a community college;
- One-half (1/2) year of credit for each year of prior related work experience to a maximum of ten (10)
  years of credit;
- One-half (1/2) year of credit for each year of full-time directly related professional staff responsibilities at the college level;
- 5. One-half (1/2) year of credit for each year of full-time elementary or secondary school directly related teaching experience to a maximum of ten (10) years of experience.
- A Master's Degree plus fifteen (15) thirty (30) graduate credit hours shall be the equivalent of one (1) year of experience.
- 7. A Doctorate shall be the equivalent of two (2) years of experience.

#### Professional Staff

Class.	Degree*	Time in Class.	Total Experience
Prof. Staff	Doctorate	2 years	4 years
II	Master's + 15-30 gr. cr. hrs. Master's or	2 years	5 years
	equivalent	2 years	6 years
Prof. Staff	Doctorate	2 years	6 years
Ш	Master's + 15-30 gr. cr. hrs. Master's or	2 years	7 years
	equivalent	2 years	8 years
Prof. Staff	Doctorate	2 years	8 years
IV	Master's + 15-30 gr. cr. hrs. Master's or	2 years	9 years
	equivalent	2 years	10 years

<sup>\*</sup>Degrees must be from regionally accredited institutions

#### Computation of Total Experience

- 1. Une (1) year of credit for each year of experience in a directly related profestional staff function within the Community College System or at a community college outside the System;
- 2. One (1) year of credit for each year of prior post-secondary experience in a directly related professional staff function at other than a community college;
- 3. One-half (1/2) year of credit for each year of prior related work experience to a maximum of ten (10) years of credit:
- 4. One-half (1/2) year of credit for each year of full-time college teaching experience;
- 5. One-half (1/2) year of credit for each year of full-time elementary or secondary school directly related professional staff function to a maximum of ten (10) years of credit;
- 6. A Master's Degree plus fifteen (15) thirty (30) graduate credit hours shall be the equivalent of one (1) year of experience.
- 7. A Doctorate shall be the equivalent of two (2) years of experience.

## **Conditions**

- Part-time work and/or teaching experience shall be credited proportionally on the aforementioned criteria. No
  fractional total shall be credited.
- b. No credit shall be given for summer session or evening division employment.
- c. The Board may in its discretion waive the degree and/or experience criteria.
- d. Degree requirements shall not apply to any unit member presently employed who does not have a Master's Degree.



- B. Fulfillment of the minimum formal requirements for promotion to a higher rank or classification and/or recommendations by the President of the College creates no right to promotion by implication and such authority remains within the sole discretion of the Board.
- C. The President of the College may waive the time in rank or classification and/or education and experience criteria requirement.

## 14.03 Announcement of Promotional Opportunities

The President of the College shall by October 1 each year declare the number of potential promotions subject to the availability of funds and positions at appropriate rank levels as determined by the President of the College that he/she intends to utilize for qualified faculty. The number of promotions available to faculty members each year shall not be reduced by an administrator who is appointed to a faculty position. The President of the College or his/her designee shall make reasonable efforts, where financially practicable, to promote unit members at each College each year at least at a rate consistent with the past practice at each College during the term of the 1977-80 Collective Bargaining Agreement; provided, however, that the President of the College or his/her designee does not impose or use any quotas of any kind by rank or classification, college, department, division, or work area and no such quota shall govern the eligibility for promotion of any unit member.

#### 14.04 Procedures

- A Each unit member who is eligible for promotion pursuant to the criteria specified in thir Agreement may apply for promotion to his/her immediate supervisor with a copy to the appropriate Dean prior to October 15.
- B. The immediate supervisor shall apply the following criteria uniformly and shall make a recommendation for or against promotion of those unit members eligible to be considered for promotion under the provisions of this Article. In making his/her recommendations, the immediate supervisor shall consider only the following in priority order:
  - 1. The unit member's performance, including all relevant material in his/her file since his/her last promotion or appointment to the College, whichever is more recent;
  - 2. The unit member's seniority; provided that for those unit members at the same rank or classification being recommended for promotion, the unit member with the most College unit seniority shall be given preference; for those unit members at different ranks or classifications, the unit member with the most time in rank or classification shall be given preference.

The immediate supervisor shall apply the criteria in B above uniformly within his/her area of supervision.

- C. For those unit members being recommended for promotion, a list within rank or classification up to the number of promotional opportunities at each rank or classification shall be established and forwarded to the appropriate dean and on request to eligible unit members prior to March 1. The immediate supervisor shall, on request, provide a written statement of reasons to a unit member placed lower on the priority list.
- D. The appropriate dean(s) shall review the recommendations and shall submit a prioritized list up to the number of promotional opportunities at each rank or classification to the President of the College and on request to eligible unit members prior to April 1. The Dean shall consider only the cr. ceria listed in 14 04B as they apoly to his/her area of supervision. If the Dean disagrees with the immediate supervisor and changes the rank order of priority, the Dean shall give a statement of reasons in writing to the affected unit member.
- E. The President of the College shall review the recommendations of the immediate supervisor and the appropriate dean(s) and shall consider only the criteria listed in 14.04B as they relate to the college in making his/her recommendations to the Board, if the President of the College disagrees with the Dean and changes the rank order of priority, he/she shall give a statement of reasons in writing to the affected unit member.
- F. Promotions shall become effective on the first (1st) day of the September pay period for faculty members and on the first (1st) day of the July pay period for professional staff.
- G. It is expressly understood that the provisions of this Article shall be applied in a manner which is not arbitrary, capricious, discriminatory, or unreasonable.

## 14.05 Non-State Appropriated Funding

Service provided by a unit member while on non-state appropriated funding shall be counted for the purposes of this Article in the event such unit member becomes an "01" employee; provided, however, that at the time of making application for a grant the President of the College or his/her designee shall when otherwise financially practicable and in conformance with law make reasonable efforts to request sufficient funds to allow for the promotion process for non-state appropriated funded unit members. It is understood by the parties that eligibility and criteria for promotion for non-state appropriated funded unit members shall be the same as for state-funded members.

## ARTICLE XV - TERMINATION, DISMISSAL, DISCIPLINE & RESIGNATION

#### 15.01 Termination

Due notice of intention of the Board to terminate a unit member must be provided in accordance with Article XI.



#### 15.02 Dismissal

Dismissal is defined as the discharging of a unit member for just cause prior to the expiration of his/her appointment and shall not be invoked except through due process.

- Charges relating to dismissal must be filed with the President of the College and may be filed only by the appropriate administrator to whom the individual is responsible, except that charges may also be made by the President of the College.
- 2. The President of the College shall notify the unit member that he/she shall be recommended for dismissal and, if appropriate request is made, the dismissal shall be preceded by a discussion with the President of the College. The matter may be concluded by mutual consent of the parties at this point.
- 3. a. If the matter is not concluded, the President of the College shall frame with reasonable particularity a formal statement of charges and shall serve the statement of charges upon the unit member. The unit member may be represented by the Association.
  - b. Within ten (10) days of the service of the statement of charges upon the unit member, he/she shall indicate in writing whether he/she requests a formal hearing. If no such request is received within (10) days, it shall be presumed that the unit member waives his/her right to a hearing.
- 4. The President of the College shall conduct the hearing, if held.
- 5. During the hearing, the unit member shall be entitled to have counsel of his/her choice which may include representation by the Association.
- 6. A verbatim record of the hearing shall be taken and a copy shall be made available to the unit member at his/her request for inspection and copying; provided that the Board and the Association shall share the cost of providing a copy of the transcript to the unit member.
- 7. a. The unit member shall be afforded an opportunity to obtain necessary witnesses and documentary evidence and the President of the College shall to the extent practicable secure the cooperation of such witnesses and make available necessary documents and other evidence within his/her control.
  - b. The unit member and the College shall have the right to confront and cross-examine all adverse witnesses. When a witness cannot or will not appear, but the President of the College determines that the interests of fairness require admission of his/her affidavit, the President of the College shall identify the witness, disclose his/her affidavit, and give such statement appropriate probative weight in view of the parties' inability to cross-examine.
- 8. If the decision is for dismissal, the affected unit member may appeal to the Chancellor or his/her designee within ten (10) days after receipt of the decision. The entire record so established, including the transcript, shall form the grievance record and shall be forwarded to the Chancellor or his/her designee upon receipt of the Step One Appeal. Said appeal shall be construed as a grievance at Step Two of the Grievance Procedure, Article X, but the following expedited procedure shall apply:
  - a. The Chancellor or his/her designee shall meet with the affected unit member for the purpose of conducting a hearing and shall render a written decision within fifteen (15) calendar days of receipt of the appeal.
  - b. Within fifteen (15) calendar days after receipt of the decision of the Chancellor or his/her designee or within fifteen (15) days after it was due, the Association may initiate arbitration of a grievance involving dismissal of a unit member in accordance with the rules of the American Arbitration Association.

#### 15.03 Discipline

Nothing in this Article shall preclude the Board or its representatives from disciplining unit members by means less than discharge, including but not limited to suspension with or without pay, provided that such discipline shall be for just cause; and provided further that a unit member who is suspended without pay shall upon written request be entitled to a hearing within fourteen (14) calendar days after receipt of such request and to back pay in the event the suspension is reversed.

#### 15.04 Resignation

A unit member shall give notice of resignation at least thirty (30) calendar days prior to the effective date of resignation, unless otherwise mutually agreed by the unit member and the President of the College or his/her designee.

15.05 Unless otherwise specifically modified herein, the provision of Article X shall be applicable to this Article.

### ARTICLE XVI - FILLING OF VACANCIES IN THE BARGAINING UNIT

16.01 A notice of vacancy in the bargaining unit shall include a statement of duties, salary range for the position, closing date of application and all qualifications for said position as determined by the President of the College or his/her designee.

All notices of said vacancies shall be posted on bulletin boards customarily used to notify unit members. In addition, the Board shall administer Article XVI in a manner consistent with Article VIII and the Board's Affirmative Action Policy.



The vacancy list prepared by the President of the College or his/her designee on a monthly basis shall be sent to the President of the Association.

- 16.02 Vacancies as defined shall be filled by unit members within the College at which the vacancy occurs when in the professional judgment of the President of the College or his/her designee such unit members are the best qualified applicants. If the President of the College or his/her designee determines that two(2) or more applicants are equally best qualified, priority of consideration shall be given in the following order:
  - 1. to the retrenched unit member within the Community College System;
  - 2. to the unit member within the department where the vacancy occurs;
  - 3. to the unit member at the College where the vacancy occurs;
  - 4. to the unit member employed at another college in the Community College System.

Any appointments to a vacancy shall be consistent with the affirmative action goals and in accordance with the Board's Affirmative Action Policy.

## ARTICLE XVII - TRANSFER

#### A. Transfer Procedures

Transfer within the Community College System or within a Community College may be initiated by the Board or requested by a unit member, provided, however, a unit member shall not be transferred without his/her consent.

In the event the transfer is initiated by the unit member, application for transfer shall be made directly to the President of the College or his/her designee to which or within which transfer is desired; provided, however, that applicants shall inform the President of the College from which transfer is desired of any such application at the time application is made. Candidates for transfer are subject to the same selection procedures as any other application for an available position. It is expressly understood that transfer is not automatic and is subject to the discretion of the receiving College; provided that if the President of the receiving College or his/her designee determines that two (2) or more applicants are equally best qualified, priority of consideration shall be given in the following order:

- I. to the unit member within the department where the vacancy occurs;
- 2. to the unit member at the College where the vacancy occurs;
- 3. to the unit member employed at another College in the Community College System.

#### B. Tenure

Tenure appointments shall be transferable from one (1) College to another only at the option of the President of the receiving College or his/her designee; provided, however, that the transferred unit member is otherwise eligible for said tenure appointment pursuant to the provisions of this Agreement.

## **ARTICLE XVIII - NOTICES**

## A. Written Notices, Communications, Etc.

All notices, recommendations, reports and official communications required by this Agreement shall be in writing and shall be deemed to be given if delivered by hand or if mailed certified mail, return receipt requested, or first class mail, and addressed to the person concerned at his/her address as shown on the records of the College or if to administrators of the College or the Board to their regular place of official business.

#### B. Notice of Class Cancellations

If classes are cancelled all faculty and professional staff shall be notified in accordance with the notification procedure established at each College that they need not report to work. Such procedure shall provide for either radio or telephone notice or both to be given at least one (1) hour prior to the beginning of the first (1st) scheduled class. In emergency situations, notice shall be given as soon as is practicable thereafter.

## ARTICLE XIX - RETRENCHMENT

## 19.01 Basis for Retrenchment

- A. The Board may from time to time retrench one (1) or more members of the bargaining unit whenever in the exercise of its sole discretion it shall have determined that such retrenchment is required due to bona fide financial reasons or where there occurs within an institution a bona fide discontinuance, reduction or shift in academic emphasis or professional service needs or for other related bona fide programmatic reasons.
- B. The provisions of this Article shall exclusively govern the discontinued employment of a member(s) of the bargaining unit prior to the expiration of his/her term of appointment through no fault or delinquency on his/her part. It is clearly understood that the nonreappointment of unit members pursuant to the provisions of Article XI is not covered by the terms of this Article.



C. The President of the College shall notify both the President of the Association and the Chapter President in writing of impending retrenchment of unit member(s) and the reasons therefor, including any available documentary evidence pertaining thereto. The President of the Association or his/her designee may respond in writing to the President of the College as to his/her judgments and recommendations based upon the contents of the initial notification letter and/or subsequent to the consultation process described in 19.02

#### D. Notification

- a. In addition to the notification given to the President of the Association under Article 19 01 C, notice of retrenchment to affected unit member(s) shall be made pursuant to the following terms and conditions:
- 1. Whenever the President of the College shall have determined that any unit member shall be retrenched under this Article, the President of the College or his/her designee shall give notice of retrenchment to the affected unit member and the effective date of such retrenchment.
- 2. Such notice shall be sent to the unit member affected as soon as practicable recognizing that, where circumstances permit, it is desirable that the effective date of said notice be sixty (60) days prior to the semester in which the employment of said retrenched unit member(s) shall be discontinued.

#### 19.02 Consultation

- A. In addition to notifying the President of the Association and the President of the Chapter as hereinbefore provided, the President of the College or his/her designee shall meet and confer with the President of the Association or his/her designee regarding the administration's plans for the retrenchment of unit members and the reasons therefore.
- B. Accurate information, statistics or financial data related to any change or plan shall be made available by the President of the College or his/her designee for inspection and/or copying upon request of the President of the Association or his/her designee; provided, however, that this shall not require the College to compile such information, statistics or financial data in the form requested unless already compiled in that form.
- C. In adopting a plan of retrenchment, the judgment of the Board upon recommendation of the College shall be final; provided, however, that said retrenchment shall not be made in an arbitrary, capricious or unreasonable manner.
- D. If and when retrenchment is to occur and it is otherwise financially practicable, the College shall make reasonable efforts to utilize attrition in order to effect the required reductions of unit members.

### 19.03 Reassignment

A. Whenever it shall have been determined to be necessary to retrench any unit member, the College shall reassign an affected unit member to a position within another division, department/ work area or professional service area within the College at which the retrenchment occurs; provided, however, that such reassignment shall only be made to a then existing vacancy in such department/work area or professional service area. No such reassignments shall be made unless such unit member is qualified for such reassignment as determined by the President of the College or his/her designee. A unit member shall be deemed to be qualified by the President of the College or his/her designee if he/she has taught at least eight (8) sections at the college in the work area to which the reassignment is to occur, or, if the provisions of the retraining article, Article 19.08, apply. Such reassignment shall not be made without the assent of the unit member.

#### 19.04 Retrenchment Procedure

- A. The President of the College shall make reasonable efforts to effect the required retrenchment by exhausting attrition and reassignment.
- B. Once a determination has been made to retrench unit members within a particular division, department, work area or professional service area the order of retrenchment shall be:
  - 1. part-time employees in the affected work area:
  - 2. temporary employees in the affected work area:
  - 3. unit members in the affected work area according to reverse seniority.
- C. In selecting among and between full-time unit members, the seniority of each unit member within any department/ work are: /program, whichever is appropriate, at a College shall determine the order in which he/she shall be retrenched from that department/work area/program area, whichever is appropriate, so that the most senior such member shall be last retrenched and the least senior such member shall be first retrenched; provided, however, that such order of retrenchment shall govern only insofar as, pursuant to its initial application, those unit members to be retained are by training, academic credentials, and/or experience as determined by the President of the College or his/her designee qualified to teach the remaining courses offered by such department or within such program area; provided further that if a faculty member is qualified to teach in more than one department/ work area/program according to the criterion of having taught eight (8) sections in that department/ work area/program, then college-wide seniority in the faculty work unit shall prevail in the event of retrenchment.
- D. A unit member who is aggrieved by the order of retrenchment shall upon request be notified of the reasons for such retrenchment as it relates to his/her employment being discontinued.



### 19.05 Seniority

- A. Seniority shall mean a unit member's length of continuous full-time unit service in the professional staff work unit and the faculty work unit at his/her Cullege; provided, however, that seniority for returning administrators shall be calculated pursuant to the provisions of Article 11.04. Continuity of service shall not be broken by a leave of absence, whether paid or unpaid, unless otherwise provided in this Agreement. Seniority shall accrue during a paid leave but shall not accrue during an unpaid leave.
- B. Seniority of a unit member who was on non-state appropriated funds small be counted on a one (1) yes to one (1) year basis in the event that such unit member becomes an "01" employee; provided, however, there is no break in service.
- C. Within sixty (60) days after the execution of this Agreement and by October 15 each year thereafter, the President of the College or his/her designee shall forward to the President of the Association or his/her designee a seniority list, which shall indicate a unit member's college-wide seniority in the professional staff work unit and the faculty work unit and the unit member's seniority in his/her department(s)/program area(s)/work area(s), whichever is appropriate.

#### 19.06 Lay-Off Status

- A. If a unit member is retrenched, no one shall be appointed to perform the released unit member's function within a period of four (4) years from the date of retrenchment; provided, however, that on or before June 10 of each calendar year subsequent to the date of retrenchment the released unit member shall give written notice by certified mail, return receipt requested, to the President of the College or his/her designee of his/her intention to be available on the recall list. Subject to the notice requirement as hereinbefore provided, such released unit member shall remain on the recall list until (1) he/she has been offered reappointment to his/her former function with equivalent tenure and no loss of benefits, or (2) said unit member has failed to meet his/her obligations under the provisions of the Article, (3) but not longer than four (4) years after the effective date of retrenchment.
- B. Notice of vacancies shall be sent to the President of the Association by the President of each College or his/her designee.

#### 19.07 Recall

- A. Whenever during the term of this Agreement it shall be determined to fill in whole or in part any unit position in any work area in which retrenchment has earlier taken place, the President of the College shall recall in inverse order of retrenchment the appropriate unit member who shall have been retrenched from such work area; provided that he/she is qualified by training and/or experience to perform the duties of the position. Whenever an offer to recall has been extended and refused by a unit member, the offer to recall shall be extended to the next unit member on the recall list. For any unit member who accepts a recall appointment to his/her College, all previously accrued seniority and other contract rights shall be retained.
- B. In addition, a unit member who is retrenched and is deemed qualified by his ner training, academic credentials and/or experience as determined by the President of the receiving College or his/her desi nee for any available vacant unit position at the receiving college shall be given priority of consideration for transfer.

#### 19.08 Retraining

- A. Any unit member facing retrenchment shall be eligible for consideration for sabbatical leave regardless of his/her length of service and, if recommended by the President of the College or his/her designee to retrain for a suitable position which would be available concurrent with the date of retrenchment, such sabbatical leave shall be granted subject to the approval of the Board.
- B. No later than sixty (60) days after execution and ratification of this Agreement, the Joint Study Committee shall establish a retrenchment retraining subcommittee which shall study and report on the feasibility of providing retraining opportunities for retrenched unit members.
- C. A unit member who is retrenched or who shall foreseeably be retrenched may request the approval of the President of the College or his/her designee to enter a retraining program without cost to the College and if such approval is granted he/she shall be extended priority of consideration for any position which the President of the College or his/her designee deems the unit member is qualified for subject to the availability of a position. Said retrenched unit member shall, upon acceptance in the regular day program at a public college or university within the Commonwealth, be granted a full tuition remission for the purposes of retraining subject to the approval of the Board of Regents or their designee.

## ARTICLE XX - DEPARTMENT CHAIRPERSONS/CURRICULUM COORDINATORS/PROGRAM COORDINATORS

## 20.01 Department Chairpersons/Curriculum Coordinators/Program Coordinators

The President of the College or his/her designee may appoint on an annual basis a unit member as a department chairperson to assist in the coordination and/or sur provision of instruction. Each such department chairperson shall have a job description specifying duties and responsibilities consistent with this Article and the needs of the College and he/ she shall be responsible to the President of the College or his/her designee for performing such.



#### 20.02 Appointment

The department chairperson/curriculum coordinator/program coordinator shall be appointed by the President of the College in accordance with the procedures described in this Article. The term of the department chairperson shall be for a renewable term of one (1) year unless a vacancy is declared the exist by the President of the College as hereinafter provided or unless the department chairperson is unable to serve.

#### 20.03 Withdrawal from Duties

Notification of at least one (1) semester shall be given by a department chairperson whenever he/ she does not wish to continue in the position of department chairperson. A department chairperson who is the only full-time faculty member in a department or work area cannot resign the department chairperson's responsibilities without also resigning his/her faculty position.

#### 20.04 Jurisdiction

The provisions of this Article shall be applicable to all unit members performing the duties and responsibilities of a department chairperson or curriculum coordinator/work area/program coordinator.

#### 20.05 Department Chairperson/Curriculum Coordinator (Work Area)

- A department chairperson (work area) may be responsible for the following duties:
- A. The submission of the preferred subject matter preparation and preferred class schedule of unit members within his/her department consistent with Article XII.
- B. Implementing the process of evaluation of course materials contained in Article XIII. The department chairperson (work area) shall not be involved in whole or in part in any other evaluation process and shall not participate in any personnel action recommendations.
- C. Advise on the discipline competency of all applicants for vacancies within the department after consultation with members of the department.
- D. Other duties may be assigned to a department chairperson (work area) if specifically contained in a job description and such duties are consistent with the needs of the College and the provisions of this Article.

#### 20.06 Department Chairperson/Curriculum Coordinator (Program)

The department chairperson (program) may be responsible for the following duties:

- A. Assist in the recruitment and orientation of new instructional staff;
- B. Advise on the instructional competency of all applicants for vacant positions within the program after consultation with members of the program unit;
- C. Submit the preferred subject matter preparation and class schedule of unit members within his/her program consistent with Article XII;
- D. Supervise the development of instructional materials and assist in conducting research on the effectiveness of the instructional program consistent with the philosophy and objectives of the College, the requirements of external and/or regulatory agencies and Article VII of this Agreement:
- E. Instruct courses or portions thereof within the program as appropriate;
- F. Assist in the implementation of the following evaluation processes as contained in Article XIII.
  - 1. The process of evaluation of course materials.
  - 2. The process for classroom observation in a clinical or laboratory-like setting wherein the student is developing a variety of occupational and/or pre-professional skills fundamental to the students' performance after completion of designated programs. The Department Chairperson (program) shall not implement the process for evaluation by students nor the process for classroom observation where didactic instruction is involved.
- G. Encourage faculty to develop new methods of instruction;
- H. Develop and prepare for submission all reports and accreditation materials required by governing or accrediting agencies;
- I. Participate in the formulation of the program's budget and administer it within the prescribed limits established by the College;
- J. Evaluate and make recommendations for any employees not in the faculty/professional staff unit assigned to his/her pr. pram area;
- K. Cooper with the President of the College or his/her designee in the development, dissemination and implementation o. Be and/College policy, regulations and procedures;
- L. Meet with the Advisory Committees which support or influence the instructional program in cooperation with his/ her supervisor. Establish liaison and cooperation with external agencies essential to the implementation of the program particularly where use of external facilities or resource personnel is required;
- M. Cooperate and facilitate cooperation with other program areas and/or departments, learning resources, student services and administrative services within the College;
- N. The designated administrator shall implement all evaluation processes for the department chairperson and the



coordinator, if said coordinator does not report to a department chairperson. If the coordinator reports to a department chairperson, the process for the evaluation of course materials shall be performed by the department chairperson;

- O. Other duties may be assigned if specifically contained in a job description and such duties are consistent with the needs of the College and the provisions of this Article;
- P. The department chairperson (program) shall be responsible for the total implementation of a college program and/ or curriculum defined as a series of diverse courses usually leading to a certificate or associate degree and which require multiple faculty in order to provide core courses and where accreditation, licensure or other external governing agencies require a member defined within the unit to perform supervisory functions for approval, maintenance or continuance of the program.

## 20.07 Procedure for the Selection of Department Chairperson®

At least two (2) months prior to the expiration of a term of office of a department chairperson, or upon a declaration of a vacancy, the president of the College shall notify the members of the discipline/work area/program of the need to select a chairperson. The following procedures shall be followed in the selection of a department chairperson:

- 1. The President of the College will post the job description for seven (7) calendar days within the College:
- 2. The members of the department/program have seven (7) calendar days to make application;
- At the expiration of this period of time, if the President of the College or his/her designee determines that there are
  no applicants qualified for the position, the President of the College or his/her designee may post outside the
  College;
- 4. When a department chairperson (discipling/work area) is to be selected, the process shall be:
  At the conclusion of the posted period, members of the department shall meet to review the applicants for the position. Thereafter, the members of the department/program will by secret ballot elect one (1) person from the applicants and thereafter make a recommendation to the President of the College or his/her disignee. If acceptable, the President of the College shall within ten (10) working days of his/her receipt of such nomination recommend appointment of such nominee to the Board.

If unacceptable, the department shall recommend an additional candidate. If the second nominee is unacceptable, the duties of evaluation of course materials shall be assigned to a non-unit member and the balance of responsibilities contained herein shall be performed by the department/work area member originally recommended to the President of the College or his/her designee.

5. When a department chairperson (program) is to be selected, the process shall be:
At the conclusion of the posted period whether by internal or external recruitment processes, members of the department/program shall meet to review all applicants for the position. Thereafter, the members of the program area will by secret ballot determine whether each applicant is qualified according to the qualifications criteria in the job description and certify without ranking them to the President of the College or his/her designee. The President of the College or his/her designee shall recommend the appointment of the department chairperson (program) to the Board.

\* Program Coordinator shall be included in the process.

## 20.08 Evaluation of Department Chairperson

A. Each department chairperson shall be evaluated annually by the President of the College or his/her designee in writing not later than March 30 of each contract year. The evaluation shall be based upon the accomplishment of tasks which have been established and agreed upon between the department chairperson and the President of the College or his/her designee.

The evaluation shall also include the evaluation by each unit member within the work area/program, the results of which shall be recorded on a form; provided, however, that any changes made on the form by the Board shall be preceded by consultation with the Joint Study Committee. The President of the College or his/her designee shall administer the evaluation form in a manner to insure the anonymity of the unit member responding.

B. The President of the College or his/her designee shall notify the department chairs room in writing of his/her recommendation appointment/non-reappointment based upon the above evaluation process. The department chairperson may apond within seven (7) working days to the evaluation.

#### 20.09 Recall

The parties recognize that the recall of a department chairperson by discipline/work area/program members is an unusual occurrence and that such recall should be based upon extraordinary circumstances. The parties agree, therefore, that the following procedures for the recall of a department chairperson by discipline/work area/program members shall not be used until a department chairperson has served at least on (1) academic year from the date of appointment.

#### 1. Informal Procedures

Within five (5) working days of the receipt of a statement setting forth spectric complaints signed by one-third (1/3) of all full-time department members, the department chairperson shall meet to confer with all members of the



department work area. This meeting shall be for the purpose of attempting to find an informal resolution of any complaints set forth in the signed statement.

If the complaints are not resolved to the satisfaction of one-third (%) of the full-time department/work area members, the concerned members should next meet informally with the academic dean to discuss the matter and obtain, if possible, the dean's assistance in reaching an informal settlement of said complaints. Failing this the department/work area members may then proceed to the formal recall procedures hereinafter described.

#### 2. Formal Procedures

A. Upon presentation to the academic dean of a petition signed by one-third (1/2) of the full-time members of the department/work area, excluding the department chairperson, stating specific reasons for recalling the department chairperson, the academic dean shall promptly give fourteen (14) calendar days' written notice to all department/work area members setting forth the time, date and place of a meeting to consider the recall petition and to vote on either a motion that the department/work area chairperson continue in office or a motion to recommend to the President of the College that he/she declare a vacancy to exist in the department/work area.

The department chairperson/work area may be present at this meeting.

- B. The academic dean ar i an impartial person from the faculty at large, who shall be elected by members of the department/work ar a, shall conduct the recall meeting, and if the academic dean and the members of the department/work area shall have so decided, shall conduct successor meetings for the same purpose. The academic dean and such impartial person from the faculty at large whall record any subsequent vote(s) taken within the department/work area on this matter.
- C. A vote by secret ballot of the majority of all full-time department/work area members shall be required to recommend to the President of the College or his/her designee that he/she declares a vacancy to exist in the department chairperson position. If a majority of the department/work area members so vote, the results of the balloting with reasons shall be forwarded to the President of the College or his/her designee. The President of the College shall determine the recall or continuance within ten (10) calendar days and so notify the department/work area with reasons. The President's decision shall be final.

#### 20.10 Workload Reduction

Any unit member who performs the duties and responsibilities of a department chairperson or curriculum coordinator, work area/program coordinator as contained in this Article shall receive a workload reduction of at least one (1) section in a coordance with Article XII, Section 12.05D, whether or not he/she holds such title. A unit member who believes he/she is aggrieved by a determination under this section may appeal to the Workload Grievance Panel in accordance with the provisions of Article XII governing said Panel.

#### 20.11 Compensation

Any unit member who agrees to perform the duties and responsibilities of a Department Chairperson or curriculum coordinator/work area/program. coordinator as contained in this Article shall be compensated at the rate of \$10.00 for each working hour if he/she agrees to perform such duties between Commencement and the first day of Fall classes, during Winter intersession and Spring vacation; provided, however, that such compensation shall not be paid for duties performed on assigned professional days as provided in Article 12.03 D. It is expressly understood that the President of the College or his/her designee shall inform the department chair, or curriculum coordinator/work area/program coordinator in a timely fashion should he/she desire to secure the services of the aforementioned unit member consistent with this Article.

#### ARTICLE XXI - SALARY ADJUSTMENTS

21.01 For the period July 1, 1983 through June 30, 1986, annual salary adjustments shall be implemented as follows:

#### 21.02 General Salary Rate Increases

- A. Effective September 25, 1983, all full-time unit members shall receive a general salary rate adjustment of an amount equal to 4%.
- B. Effective April 1, 1984, all full-time unit members shall receive a general salary rate adjustment of an amount equal to 2%.
- C. Effective July 1, 1984, all full-time unit members shall receive a general salary rate adjustment of an amount equal to 2%.
- D. Effective September 30, 1984, all full-time unit members shall receive a general salary rate adjustment of an amount equal to 6%.
- E. Effective June 30, 1985, all full-time unit members shall receive a general salary rate adjustment of an amount equal to 7%.
- F. Effective September 30, 1985, all full-tim.) unit members shall receive a general salary rate adjustment of an amount equal to 2%.



## 21.03 Performance Based Salary Adjustments

- A Effective September 25, 1983, there shall be established at each College a pool of 1.5% of the total salaries before the application of the 4% general increase which will be used for performance based salary adjustments for unit members who received other than unsatisfactory on their evaluations for the 1982-1983 academic year
- B. Effective September 30, 1984, there shall be established at each College a pool of 2% of the total salaries before the application of the 6% general increase which will be used for performance based salary adjustments for unit members who received other than unsatisfactory on their evaluations for the 1983-1984 academic year.
- C. Effective June 30, 1985, there shall be established at each College a pool of 1.5% of the total salaries before the application of the 7% general increase which will be used for performance based salary adjustments for unit members who received other than unsatisfactory on their evaluations for the 1984-1985 academic year.

## 21.04 Educational Needs/Professional Development

Effective June 26, 1983, again effective July 1, 1984, and effective June 30, 1985, there shall be established at each College a pool of 1% of total salaries for educational needs and professional development purposes. The 1984 and 1985 pools will be calculated before the application of general increases effective July 1, 1984 and June 30, 1985. It is understood these adjustments do not affect base rates. These pools shall be distributed in accordance with the following procedures. Participation in any such program by any member of the bargaining unit shall be undertaken only on a voluntary basis.

#### A. Purpose

- 1. To improve teaching and student advising and to develop new course materials and to relate those to a changing curriculum that is itself responsive to the larger needs of society;
- 2. To develop new teaching skills and an appropriate facility in the use of media, teaching aids and other supportive techniques where those are appropriate;
- 3. To improve student advising techniques and the faculty member's command of relevant bodies of knowledge in this area;
- 4. To increase the faculty member's command of the body of knowledge that constitutes his own discipline;
- 5. Where appropriate, to enable a faculty member to develop a command over a body of knowledge in a related discipline;
- In the case of any professional staff unit member, to improve such professional staff unit member's professional skills and techniques and to relate those to a changing curriculum that is itself responsive to the larger needs of society; and
- 7. In the case of any professional staff unit member, to develop new professional skills and techniques to the same end.
- B. In Fiscal Years 1984 and 1985 no unit member shall be unreasonably denied a minimum of \$300 in educational needs funds provided he/she submits a bona fide request in accordance with A. above. In Fiscal Year 1986 no unit member shall be unreasonably denied a minimum of \$125.00 in educational needs funds provided he/she submits a bona fide request in accordance with A. above.

## C. Individual Programs of Professional Development

Within thirty (30) days of the date of execution of this Agreement, there shall be established at each College a Committee on Professional Development and Retraining, which shall be composed of three (3) unit members appointed by the President of the Association and two (2) persons appointed by the President of the College.

#### 1. The Committee shall:

- Meet and confer with the Dean of Academic Affairs regarding criteria for the awarding of Professional Development Programs and shall utilize the guidelines for such Professional Development Programs;
- b. Receive and review all requests for programs of professional development as may be undertaken pursuant to the provisions of this Article, and the guidelines in effect and thereafter make recommendations to the President of the College regarding approval or disapproval of all programs of professional development submitted pursuant to the provisions of this Article:
- c. Make recommendations regarding the allocation of monies which shall be available for the implementation of programs of professional development as may be undertaken pursuant to the provisions of this Article;
- d. Develop and recommend a policy for the retraining of unit members who may be retrenched, and submit such recommended policy to the President of the College within three (3) months from the date on which the Committee shall have first convened; in discharging this responsibility, the Committee shall seek to compile and disseminate information regarding retraining and job opportunities for members of the bargaining unit.
- e. Advise, when requested, individual unit members who shall have been or may be retrenched pursuant to the provisions of Article XIX of this Agreement for the purpose of assisting such unit member in the development of a program of retraining.



## 2. The President of the College shall:

- a. Upon receipt of the recommendations of the Committee on Professional Development and Retraining, grant such awards as he/she zirall determine in accordance with the provisions of this Article;
- b. Notify the applicants of the approval or disapproval of their program of professional development;
- c. Notify the President of the Association of the grants and programs approved by him/her.

#### D. Definition

For the purpose of this Article, the phrase "individual program of professional development" shall in the case of any individual member of the bargaining unit mean a program designed to effectuate any one or more of the purposes set forth in Section A, which program shall have been developed at the initiation of such unit member, assented to by him/her, and approved as such by the President of the College or his/her designee; it being the common purpose of the parties to this Agreement to provide members of the bargaining unit with the specific means by which to achieve professional growth and to benefit the College.

## 21.05 Promotions and Equity Adjustments

## A. Fiscal Year 1984

- 1. Effective September, 1983, there shall be created at each College a pool of money equivalent to 0.65 percent of the total annual unit salaries of all current unit members whose salaries were paid from the State Maintenance Account as of September 24, 1983, prior to the salary increases contained in Articles 21.02 and 21.03. If a larger pool is necessary to fund the salary adjustments in paragraph 3 below, such funds shall be taken from the fiscal year 1985 pool created in 21.05 B.
- 2. Any monies not distributed in accordance with Article 21.03 shall be added to this pool.
- 3. The money in this pool shall be distributed in the following manner:
  - a. To provide salary adjustments to those unit members still employed at the College who were identified in the May 1983 Equity Study as having more than a four (4) step inequity.
  - b. To provide an additional one-half (1/2) step salary increase on the appropriate salary schedule to faculty who were promoted with an effective date of September 1, 1983, and an additional two (2) step increase to the professional staff unit members who were promoted effective July 1, 1983.
- 4. After the cost of said adjustments in 3 (a) and 3 (b), any unexpended funds in the fiscal year 1984 pool shall be made available for use in the fiscal year 1985 pool.

- 1. Effective September 30, 1984, there shall be created at each College a pool of money equivalent to 0.35 percent of the total annual unit salaries of all current unit members whose salaries were paid from the State Maintenance Account as of September 29, 1984, prior to the salary increases of September 30, 1984 as provided for in Articles 21.02 and 21.03.
- 2. Added to this pool shall be the following:
  - Any monies not distributed from the fiscal year 1984 pool.
  - b. Any monies not distributed in accordance with Article 21.03.
- 3. The money in this pool shall be distributed in the following order:
  - a. To provide an additional one-half (1/2) step salary increase on the appropriate salary schedule to faculty members who were promoted with effective dates of September 1, 1984, and an additional two (2) step increase to the professional staff members who were promoted effective July 1, 1984.
  - b. To provide additional promotions to those eligible unit members still in the employment of the College. Said promotions shall be from the prioritized lists prepared in the 1983-84 academic year. At those Colleges that do not prioritize beyond the number of promotional opportunities available, promotion recommendations shall be made in priority order up to the number of additional promotions from the pool of eligible applicants that was created in October 1983 pursuant to Article 14.04 of the 1980-83 Collective Bargaining Agreement. Unit members who are promoted shall be moved to the next higher rank or classification and shall receive a salary increase equal to 1 1/2 steps for faculty and three (3) steps for professional staff unit members; provided, however, that unit members who are promoted to step one of the next higher rank or classification may receive a salary increase greater than 1 1/2 steps for faculty or 3 steps for professional staff
  - c. The cost of said adjustments as provided for in Article 21.05(B)3 shall not exceed the amount available in the pool.

## C. Fiscal Year 1986

- 1. Effective September 1, 1985, there shall be created at each College a pool of money equivalent to 0.5 percent of the total annual unit salaries of all current unit members whose salaries were paid from the State Maintenance Account as of August 31, 1985.
- 2. Any monies not distributed in accordance with Article 21.03 shall be added to the pool.



- 3. The money in said pool shall be utilized in the following manner:
  - a. To provide an additional one-half (1/2) step salary increase on the appropriate salary schedule to faculty members and an additional two (2) step increase to the professional staff unit members promoted in accordance with the announcement of promotional opportunities made in October, 1984.
  - b. To provide promotions in addition to those previously announced in October. 1984. Unit members who are promoted shall be moved to the next higher rank or classification and shall receive a salary increase equal to 1.1/2 steps for faculty and 3 steps for professional staff unit members; provided, however, that unit members who are promoted to step one of the next higher rank or classification may receive a salary increase greater than 1.1/2 steps for faculty or 3 steps for professional staff unit members.
- 4. The cost of said additional promotions in Article 21.05(C)3 shall not exceed the available amount of money in the pool.
- D.1. No unit member shall be eligible to receive more than one promotion during the life of this Agreement, July 1.

  1983 through June 30, 1986; provided, however, that in the event that the Agreement is extended beyond June 30, 1986, any otherwise eligible unit member would be eligible for promotion after June 30, 1986.
  - 2. It is clearly understood that no waivers of the eligibility requirements found in Article XIV shall be granted to provide for promotions under this Article.

#### 21.06 Promotion Wage Adjustments

Unit members who are promoted shall be moved to the next higher rank or classification and shall receive a salary it. Tease equal to 1 1/2 steps for faculty and 3 steps for professional staff unit members; provided, however, that unit members who are promoted to step one of the next higher rank or classification may receive a salary increase greater than 1 1/2 steps for faculty or 3 steps for professional staff unit members.

## ARTICLE XXII - MANAGEMENT ASSOCIATION COMMITTEE ON EMPLOYEE RELATIONS

#### 22.01 College Level

There shall be established a committee at the College level to be known as the Management Association Committee on Employee Relations. Such Committee shall be comprised of six (6) members: three (3) representing the local College ariministration and three (3) representing the Association. Such representatives shall be respectively appointed by the President of the College and the President of the Chapter. In addition, the President of the College shall designate the chairperson for the College and the President of the Chapter shall designate the chairperson for the Association.

The purpose of said Committee shall be to discuss matters of mutual concern to the employee and the employer. There shall be at least one (1) meeting per month during the academic year with the chairpersonship alternating between the College and the Association; provided, however, that whenever the parties mutually agree there is no need for a meeting during a month there shall not be a meeting.

Both parties may submit items for the agenda to the chairperson at least two (2) weeks in advance of any scheduled Committee meetings. The agenda shall be distributed one (1) week in advance of any scheduled committee meetings.

It is understood that said Committee shall have no power to negotiate, alter, or amend the terms of this Agreement.

## ARTICLE XXIII - NO STRIKE OR LOCK OUT PLEDGE

23.01 The Board agrees that it shall not lock out any or all of its employees for any cause during the term of this Agreement and the Association and its agents agree that they shall not engage in, induce or encourage any strike, work stoppage, slow down or withholding of services by said members.

Nothing contained in this Article shall be deemed to waive, impair or restrict the right of the Board or the Association to seek or pursue any remedy at law or equity provided by the laws of the Commonwealth in the event of a violation of this Article.

## ARTICLE XXIV - SAYINGS CLAUSE

24.01 If any of the provisions of this Agreement shall in any manner conflict with or contravene any federal or state law, statute or the rules and regulations promulgated thereunder, such provisions shall be considered null and void and shall not be binding on the parties.

The Board and the Association acknowledge that during the negotiations which resulted in this Agreement each had the ultimate right and opportunity to make demands and proposals with respect to any subject or matter not removed by law from the applicable areas of collective bargaining, and that the understanding and agreements arrived at by the



5 1/7 parties after the exercise of that right and opportunity are set forth in this Agreement and shall constitute the sole Agreement between the parties.

In recognition of this fact the Board and the Association for the life of this Agreement each voluntarily and unqualifiedly waives the right and agrees that the other shall not be obligated to bargain collectively with respect to any subject or matter not in this Agreement even though such subject or matter may not have been within the knowledge or contemplation of either or both parties at the time they signed this Agreement; provided, however, that nothing in this Article shall prohibit the parties from conducting negotiations during the term of this Agreement regarding the impact on terms and conditions of the Board or its successor to close any College or to merge any College with another educational institution to consolidate, discontinue, or transfer existing functions, educational activities and programs.

The provisions of this Article notwithstanding, the parties may by mutual agreement upon the request of one (1) or both parties reopen negotiations on the provisions of this Agreement prior to the expiration date provided in Article XXVI.

### ARTICLE XXV - HOLDOVER

25.01 In the event that the Board and the Association shall fail to secure a successor Agreement as hereinafter provided in Article XXVI prior to the termination of this Agreement, then this Agreement shall remain in full force and effect until a successor agreement is executed or an impasse in negotiations is reached.

## ARTICLE XXVI - DURATION AND SUCCESSORSHIP

26.01 This agreement shall be in full force and effect from and after July 1, 1983 through June 30, 1986; provided, however, that nothing herein contained shall be deemed to impose on the Board any obligation the discharge of which may be required to be sought pursuant to General Laws, Chapter 150E, Section 7, until such time as such appropriation shall have been duly made by the General Court pursuant to said provision of the General Laws; provided further that notwithstanding the foregoing, whenever the General Court shall not have acted pursuant to said provision and the Board shall have monies allocable to the discharge of any obligation herein contained and any such monies shall at the sole discretion of the Board have been so allocated, such obligation shall be discharged in such measure as such monies so allocated shall permit.

If, in respect to this Agreement, the Governor shall have failed to act pursuant to General Laws, Chapter 150E, Section 7, the Association shall have the right upon thirty (30) days' written notice to the Board to require that the parties to this Agreement shall resume collective bargaining pursuant to the provisions of General Laws, Chapter 150E. Except as provided by express agreement of the parties, no claim of grievance of any collective bargaining agreement between the Board and the MTA/MCCC shall have validity that may be alleged to have occurred prior to the beginning of Academic Year 1983-1984.

In accordance with applicable provisions of Section 10 of Chapter 15A of the General Laws, each local Board of Trustees for each College covered under this Agreement shall appoint, transfer, dismiss, promote, and award tenure to all personnel of said College, subject to policies promulgated or agreements entered into by the Board of Regents.

To the extent permitted by law, the Board of Regents of Higher Education may delegate its authority or any portion thereof to the local Board of Trustees for each College whenever in its judgment such delegation may be necessary or desirable.



## SIGNATURE PAGE

For the Massachusetts Board of Regents of Higher Education and the Massachusetts Regional Community Colleges	For the Massachusetts Teachers Association/Massachusetts Community College Council		
By: Carleton H. LaPone Vice Chancellor for Employee Relations	By:		
Date:	Date:		
By: John B. Duff Chancellor	By:  Jack Carpenter  MTA Consultant		
Date:	Date:		



## MEMORANDUM OF AGREEMENT

The Board of Regents of Higher Education and the Massachusetts Teachers Association/Massachusetts Community College Council (Association) hereby enter into this Agraement, which shall be incorporated into the 1983-86 Collective Bargaining Agreement.

Because the parties have agreed to delete multiple-year contracts from the 1983-86 Collective Bargaining Agreement, this Agreement has been deemed necessary to clarify the parties' understandings regarding the status of multiple-year contracts. Therefore, it is agreed as follows:

- a multiple-year contract shall remain in full force and effect until its expiration or until the unit member becomes eligible for tenure;
- a unit member who is otherwise eligible may apply for tenure during the term of his/her multiple-year. contract;
- 3. multiple-year contracts shall be transferrable from one College to another only at the option of the President of the receiving College or his/her designee and if approved by the Board.

For The Board  Carleton W. VaPorte  Vice Chancellor For  Employee Relations	R. Michael McSweeney Chairman, MTA/MCCC Negotiating Team
Date: 3/83	11/29/84 Date

# **AGREEMENT**

## **BETWEEN**

The Massachusetts Board of Regents
of Higher Education
For The
Massachusetts Regional Community Colleges

## AND

The Massachusetts Community College Council/
Massachusetts Teachers Association
An Affiliate Of
The National Education Association

## For Academic Years

1986-1987 • 1987-1988 • 1988-1989 July 1, 1986, through June 30, 1989



## **NEGOTIATING TEAM PAGE**

### For the Board

#### CAROLYN R. YOUNG

Chair

Associate Community College Counsel

#### JOHN DUNN

Vice Chair

Vice President and Dean of Academic Affairs Springfield Technical Community College

#### GERARD F. BURKE

President

Massasoit Community College

#### PATRICIA CHISHOLM

Dean of Student Services
Bunker Hill Community College

#### FRANK CROWLEY

Contract Administrator/Hearing Officer

#### KEVIN KIERNAN

Dean of Administrative Services Quinsigamond Community College

#### ROBERT MC DONALD

Dean of Academic Affairs
Northern Essex Community College

#### GARY McPHEE

Director of Personnel
Middlesex Community College

#### DONALD ZEKAN

Dean of Administrative Services Massasoit Community College

## For the Association

### R MICHAEL MC SWEENEY

Chairperson

Bunker Hill Community College

#### **DENNIS FITZGERALD**

Vice Chair

Massasoit Community College

#### **GEORGE ASHLEY**

Team Secretary

Holyoke Community College

#### LEO ARNFELD

Bristol Community College

#### JAMES RICE

MCCC President

Quinsigamond Community College

#### THOMAS PARSONS

MCCC Vice President

Massachusetts Bay Community College

#### JACK CARPENTER

MTA Negotiator/Consultant

#### JOHN OSBORNE

Alternate

Northern Essex Community College

## Resource Consultants

#### RAYMOND LEMIEUX

MCCC Treasurer

Springfield Technical Community College

#### CATHERINE A. BOUDREAU

MCCC Communications Coordinator
Massasoit Community College





## TABLE OF CONTENTS

Negoti	ating Teams' Page	1	ARTICLE VIII	
-	ole	4		11
Prezim	JE	•		
ARTIC	T.E. I		ARTICLE IX	
	nition and Definitions	4	Supplemental Benefits	11
1.01		4	9.01 Authorized Leaves with Pay 1	l
1.02	Definitions	4	9.02 Unpaid Leaves of Absence 1	17
1.03	New Positions Added	5	9.03 Insurance and Other Benefits 1	19
1.04	Jurisdiction	5	9.04 Early Retirement Incentive 2	20
1.04	Appendix A	6	9.05 Unit Member Resources,	
	Appendix A	•	Assistance and Protection 2	2:
ARTIC	CLE II			
	nship Between		ARTICLE X	
	ation and Board	6		21
2.01	Fair Practices	6		24
2.02	Complaints	6		
2.03	Safety	6		
2.04	Individual Contracts	6	ARTICLE XI	
2.05	Association Representatives	6	Appointment and	
2.05	Relevant Information	6	Reappointment - Tenure	25
2.07	Association Leave	7		
2.07	Association Leave	•	ARTICLE XII	
			Worklor ', Work Assignment	
	CLE II -A	7		26
Specia	I Joint Study Committee	7	· ·	
				26
	CLE III			26
Use of	Board Facilities	8		27
3.01	Facility Use	8		29
3.02	Bulletin Boards	8		31
3.03		8	Appendix A 3	31
3.04	Unit Facilities and Services	8		
3.05	Access to College Facilities	8	ANTICLE XIII	
				33
ARTTO	CLE IV :			33
	ghts and Responsibilities			, s 33
	Board	8		د،
or the	DOMES	·	13.03 Evaluation of Professional	
A D.TT.	3 T 15/ A			35
	CLE IV -A	0	13.04 Evaluation of Tenured	
College	Governance	9		36
			•	36
ARTIC		_	13.06 Written Reasons	36
Mainte	enance of Records	9		
			ARTICLE XIV	
	CLE VI		Criteria and Process for Promotion	
Deduc	tion of Dues and Agency Fees	9	of Faculty and Professional	
6.01	Dues Deductions	9	•	36
6.02	Agency Service Fee	9		36
				96
ARTIC	CLE VII			U
	nic Freedom & Responsibility	10	14.03 Announcement of	37
7.01	Academic Freedom			
7.01 7.02		10		37
1.02	Academic Responsibilities	10	14.05 Non-State Appropriated Funding 3	8



ARTI	CLE XIV-A		20.02 Appointment	44
Criter	ia and Process for Change of		20.03 Withdrawal from Duties	44
	of Faculty and Professional		20.04 Jurisdiction	44
Staff	Unit Members	38	20.05 Department Chair/Curriculum	77
14 A.				44
	Classification System	38	20.06 Department Chair/Curriculum	77
14 A.		38		45
	O3 Procedures	40	20.07 Procedure for the Selection of	73
14 4 (	Additional Qualifications	40		45
A 7 6 801	A Managina Anamysandin	70		46
ADTT	CLE XV			46
				<del>4</del> 7
	nation, Dismissal,	40		47
-	pline, Resignation	40	20.11 Compensation	41
15.01	Termination	40	ARTICLE XXI	
15.02	Dismissal	40		
15.03	Discipline	41		47
15.04	Resignation	41		47
				47
ARTI	CLE XVI		21.03 Performance-Based	
	of Vacancies in the			47
	ning Unit	41		47
		41		48
	0. m 19.6.		21.06 Education Needs/	
	CLE XVII			48
Iransi	er	41	Salary Schedule	49
ARTI	CLE XVIII		ARTICLE XXII	
Notice	<b>\$</b>	42	Management Association Committee	
				50
ARTIC	CLE XIX		• •	. •
	chment	42	ARTICLE XXIII	
19.01	Basis for Retrenchment	-		50
19.01		42	The Beam of Book Out I stugo	,0
19.02	Consultation	42	ARTICLE XXIV	
	Reassignment	43		
19.04	Retrenchment Procedure	43	Savings Clause	50
19.05	Seniority	43		
19.06	Lay-Off Status	43	ARTICLE XXV	
9.07	Recall	44	Holdover	50
9.08	Retraining	44		
			ARTICLE XXVI	
ARTIC	CLE XX		Duration and Successorship	51
Depart	ment Chairs	44	<del>-</del>	
20.01	Department Chair/Curriculum Coordinators/Program		Signature Page	52
	Coordinators	44	Form Supplement i _ vvv	- -



#### **PREAMBLE**

This Agreement is entered into by and between the Board of Regents of Higher Education (hereinafter the Board), and the Massachusetts Teachers Association/Massachusetts Community College Council (hereinafter the Association) as the exclusive bargaining representative for all employees in the bargaining unit described in Article I. Both parties to this Agreement recognize the unique contributions of the community colleges to education in this Commonwealth. Further, the parties recognize the need to strengthen and secure corruntly college education. This Agreement has as its purpose the promotion of harmonious relations between the Board and the Association. The parties declare their commitment to maintaining and improving the quality of educational services offered by the Community Colleges. To this end the parties recognize their statutory obligations, pursuant to the provisions of General Laws, Chapter 150 E, and the rules and regulations promulgated thereunder, to negotiate in good fait, with respect to wages, hours, standards of productivity and performance, and other terms and conditions of employment.

Both parties recognize and declare that providing quality higher education and services to the citizenry of the Commonwealth with the broadest accessibility is their mutual goal. The education of our students is our primary purpose.

In recognition of these obligations it is hereby agreed as follows:

#### ARTICLE I - RECOGNITION AND DEFINITIONS

#### 1.01 Recognition

The Board hereby recognizes the Association as the exclusive bargaining representative with respect to conditions of employment for all regular full-time employees occupying the positions delineated in Appendix A.

The Board agrees to apply the applicable provisions of this Agreement to those employees whose source remuneration is derived from non-state appropriated funds and who perform the functions of any of those positions delineated in Appendix A, to the extent that the terms of their respective grants or non-state appropriated funding source and the level of funding thereunder allow, as determined by the President of the College or his/her designee. It is understood that the following Articles of this Agreement shall not apply to those individuals occupying said positions, except as otherwise provided in this Agreement:

Article XI Appointment and Reappointment

Article XIX Retrenchment

#### 1.02 Definitions

Academic Year - a period of time encompassing two (2) semesters, beginning no earlier than September 1 and ending no later than May 31, exclusive of Commencement.

Administration – all professional employees of the Board who are not members of the bargaining unit as described in Appendix A.

Administrator - a non-unit professional employee.

Association - the Massachusetts Community College Council/Massachusetts Teachers Association, an affiliate of the National Education Association.

Association Representative – a member of the Association who has been designated as a representative by the Association President in writing to the President of the College; also, a representative of the Massachusetts Teachers Association or the National Education Association as designated by the Association President to the President of the College.

Board – the Board of Regents of Higher Education or any College Board of Trustees as defined in General Laws, Chapter 15 A, whichever the case may be as provided in Article XXVI.

Chancellor – the Chancellor of the Board or an individual acting in that capacity as duly appointed by the Board, in accordance with General Laws, Chapter 15 A.

Cause - shall mean just cause.

College President – the President of a Community College or an individual acting in that capacity as duly appointed by the Board in accordance with General Laws, Chapter 15 A.

Colleges – all facilities and properties of a Community College now or hereinafter established by the Board pursuant to General Laws, Chapter 15 A.

ERIC

Full Taxt Provided by ERIC

Complaint – a written statement setting forth a grievance as hereinafter defined, which includes a statement of all the known facts pertaining to the alleged breach on which the grievance is based, including but not limited to the date(s) when the breach allegedly occurred, the specific contractual provisions alleged to have been breached, and the remedy requested.

Curriculum Coordinator – a unit member who functions in a similar manner as a department chairperson but is responsible for a smaller work area.

Department Chair (Work Area) - a unit member performing the duties d lineated in Article 20.05.

Department Chair (Program) – a unit member performing the duties delineated in Article 20.06 and who may be assigned responsibility to assist in the implementation of a College program and/or curriculum, defined as a series of courses customarily leading to a certificate or associate degree and which accreditation, licensure or other external governing agencies require a member defined within the unit to perform supervisory functions for approval, maintenance and continuance of the program.

Dismissal - the discharge of any unit member for just cause prior to the expiration of his/her appointment.

Faculty – unit members occupying full time positions as instructor, assistant professor, associate professor or professor.

Field - Based Type Work - off-campus educational experience where the faculty member is not responsible for delivering instruction but is responsible for ensuring that instruction is delivered.

Grievance — an allegation by a unit member(s) or by the Association that a specific provision of the Agreement has been breached in its application to him/her or it.

Grievant - a unit member (s) or the Association who seeks a resolution of a grievance pursuant to Article X.

Lay Off Status - the status of a unit member who has been retrenched pursuant to Article XIX.

Professional Judgment — every decision to renew or fail to renew a professional appointment, to terminate any such appointment, to grant or refuse to grant academic tenure, promotion, professional leave, sabbatical leave, educational needs/professional development awards, performance-based salary adjustments, and evaluation results shall be deemed to have been made pursuant to an exercise of professional judgment; and every grievance that, explicitly or by implication, questions the merits of any such decision, but of no other decision, shall be deemed to be a grievance that questions an exercise of professional judgment. In matters of professional judgment, said decision is limited by the specific terms of this Agreement.

Professional Staff Member - a unit member whose primary duties are other than teaching.

Promotion — one step increase on the salary schedule in addition to any step increase awarded in accordance with Article 21.03.

Recall — the removal for cause of a person from his/her position as department chairperson but not the separation of said person from the service of the College.

Retrenchment - the discontinued employment of a unit member prior to the expiration of his/her term of appointment through no fault or delinquency on his/her behalf, pursuant to Article XIX.

Seniority – continuous full-time service by a unit member at his/her College, which is deemed to commence with the unit member's effective date of employment as a unit member.

Temporary Employee – one employed as a unit member to substitute for a unit member holding a regular, multiple year, or tenured appointment for a period of one (1) year or less for the purpose of replacing a unit member on leave or whose employment ended prior to the completion of the fiscal or academic year.

Tenure – an appointment which vests a right to continued employment at a College without limit of time subject to dismissal for just cause in accordance with Article XV and retrenchment in accordance with Article XIX.

Termination - the non-renewal of a contract of a non-tenured unit member.

Transfer – the appointment by the Board of a unit member to a faculty or professional staff position at the same rank or classification in the same or another College in the Community College System.

Unit Member - an employee of the Board occur jing a position as defined in Article I and as delineated in Appendix A.

Work Area - those specific work areas designated by the administration as work areas.

Vacancy - a position for which funding is avail - and which the administration intends to fill.

## 1.03 New Positions Added

The Board recognizes its obligation under the provisions of General Laws, Chapter 150E, to bargain in good faith over the addition of any professional positions within the bargaining unit in excess of the total number of existing bargaining unit positions at a College.

#### 1.04 Jurisdiction

During the term of this Agreement, the Board agrees not to negotiate with any unit member (s) or with any employee organization other than the Association.



#### ARTICLE I - APPENDIX A

The bargaining unit contains the following positions as described in Article 1.01:

Position	Payroll Title Code	Position	Payroll Title Code
Instructor	90-156	Coordinator of Student Acti	vities 90-754
Assistant Professor	90-155	Placement Officer	90-375, 90-717
Associate Professor	90-154	Health Care Counselor	90-771
Professor	90-153	Counselor	90-712
Librarian (in those cases we Librarian is under the su the Director of the Learn	pervision of	AV/TV Coordinator Director of Instructional Me those cases where the Dir	
Center or its equivalent)		under the supervision of t	he Director
Assistant Librarian	90-703	of the Learning Resource	
Cataloguer	90-724	its equivalent)	90-772
	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	Director of Cooperative Edu	ucation 90-766

#### ARTICLE II - RELATIONSHIP BETWEEN ASSOCIATION & BOARD

#### 2.01 Fair Practices

The Board and the Association recognize and affirm their commitment to the policy of non-discrimination with regard to race, color, religious creed, national origin, sex, sexual preference, age, marital status, family relationship and handicap status.

#### 2.02 Complaints

The administration shall within fourteen (14) calendar days send or communicate to a unit member any written complaint or material which the administration believes would adversely affect his/her employment status.

#### 2.03 Safety

In accordance with applicable state or federal law, unit members shall not be required to work under unsafe conditions whenever such conditions have been brought to the attention of the President of the College or his/her designee by the unit member (s) and the College has failed to exercise reasonable efforts to redress the complaint.

#### 2.04 Individual Contracts

All rights, benefits, duties and obligations. unit members as set forth in the Agreement shall during its term be expressly incorporated by reference into and made part of any contract of employment that has been or shall be entered into between the Board and a unit member and no such contract shall be contrary, in whole or in part, to the terms and conditions as set forth herein.

#### 2.05 Association Representatives

- A. The President of the Association shall furnish the Chancellor or his/her designee with a written list by September 15 of each year of officers and representatives of the Association and their terms of office. It is clearly understood that in the implementation of this section, there shall not be more than one (1) Association representative from each College; provided, however, that if a designated representative is unavailable, the President of the Association shall notify the Chancellor or his/her designee as to who shall be that representative's alternate.
- B. The President of the Association shall notify the Chancellor or his/her designee and the Presidents of the affected Colleges or their designees of the names of unit members who have been certified to have been approved by the President of the Association for a workload reduction. The Board shall provide fifteen (15) course sections per semester without cost to the Association. The Association shall pay for an additional fifteen (15) course sections per semester at the usual rate received by part-time faculty at each affected College.
  - Unit members who receive a workload reduction shall teach at least one (1) course per semester. No College shall be required to provide more than three (3) sections of unpaid workload reduction in any semester.
  - For purposes of this Article, seven (7) clock hours release time per week shall be considered one (1) section of release for professional staff unit members.
- C. The Association shall also notify the Presidents of Colleges by May 1 of each year of the names of each local chapter president or his/her designee who has been certified for a workload reduction. Such local chapter president shall not be assigned student advisees and shall have three (3) office hours: provided that the unit members shall provide coverage for the reduction of student advisees who shall be assigned by the President of the College or his/her designee. Such workload reduction shall be subject to the approval of the President of the College or his/her designee. Each College shall make every reasonable effort to schedule such unit members to the maximum opportunity to participate in the workload reduction provided herein.

#### 2.06 Relevant Information

The Board shall provide to the Association in accordance with Chapter 150E, such information as is necessary for the proper discharge of its duties as the exclusive bargaining agent. Such information shall include, but not be limited



to, a copy of the payroll at each College, the number of vacant and filled positions at each College, and the enrollment figures by program at each College; provided, however, that such information shall be provided to the local chapter president by October 15 of each year; provided further that this shall not require the College to compile such information in the form requested unless already compiled in that form.

The President of the College or his/her designee shall provide to the local chapter president copies of the tentative class schedule for all faculty members for the subsequent semester no later than the dates by which the faculty members must be notified of these tentative schedules as provided in Article XII. The President of the College or his/her designee shall also provide to the local chapter president no later than the end of the first week of classes each semester the class schedules of all faculty members whose class schedules differ from their tentative class schedules.

#### 2.07 Association Leave

- A. All meetings referred to in this section shall be scheduled so as to provide the least disruption of classes.
- B. When it is necessary, pursuant to the Grievance Procedure in Article X of this Agreement, for a representative designated by the Association to attend a hearing during a work day, he/she shall upon notice to his/her immediate supervisor be released without loss of pay as necessary in order to participate in the foregoing activities. Any person whose appearance in such hearings as a witness is necessary shall, when possible, obtain coverage for his/her classes satisfactory to the President of the College.
- C. When the Board of Directors' meetings of the Association are scheduled during normal working hours of a work day, unit member representatives to such meetings shall be relieved from all regular duties without loss of pay; provided, however, that such release from duties shall be limited to no more than ten (10) meetings. The President of the Association shall notify the Chancellor or his/her designee and the Presidents of the affected Colleges of the names of the Board of Directors and the dates of the ten (10) scheduled Board of Directors' meetings prior to September 1. Whenever it becomes necessary to schedule additional meetings of the Association's Board of Directors, such members may be released; provided, however, that they first obtain coverage for their classes satisfactory to the President of the College.
- D. Leave of absence with pay may be granted for officers, delegates or alternates of employee organizations for the purpose of attending conventions of their organizations.
- 2.08 Whenever possible grievance meetings shall be scheduled so as not to interfere with professional responsibilities of individuals involved. If it is necessary to meet with the employer during working hours, the grievant, one (1) Association representative who is a member of the bargaining unit, and necessary witnesses may attend without loss of time or compensation for such meetings.

## ARTICLE II-A - SPECIAL JOINT STUDY COMMITTEE

There shall be established under this Agreement a Special Joint Study Committee according to the following provisions:

- 2A.01 There shall be appointed eight (8) members: the Board of Regents of Higher Education shall appoint three (3) Community College Presidents and the Director of Employee Relations, or in his absence, the Associate Director of Employee Relations; the President of the Massachusetts Community College Council shall appoint three (3) officers of the Council and a representative or officer of the Massachusetts Teachers Association.
- 2A.02 The Committee shall meet no less than six (6) times per year, unless mutually agreed otherwise.
- 2A.03 The Committee may ask a neutral to join them on occasion as a useful catalyst in creating a problem-solving atmosphere; either management's representatives or the Association's representatives may invite resource persons to attend committee meetings.
- 2A.04 The Committee shall make rules of order within which to operate.
- 2A.05 All fees and expenses of the Committee (including those necessary to retain a neutral) shall be divided equally between the parties to this Agreement.

The Committee during the life of this Agreement shall:

- 1. Resolve matters of mutual concern between the parties to this Agreement.
- 2. Address issues which require in-depth study and discussion.
- 3. Resolve contract issues referred to it in the recent fact-finder's report and recommendations.
- 4. Explore and identify root causes of current problems between the parties with a view of resolving all such problems.
- 5. Make recommendations to the parties to this Collective Bargaining Agreement during the life of the new contract as recommended by the Fact-Finder.

All reports, findings, and recommendations of the Special Joint Study Committee shall be expeditiously given to the Chancellor of the Board of Regents of Higher Education and the President of the Massachusetts Community College Council.



## ARTICLE III - USE OF BOARD FACILITIES

3.01 Facility Use

Upon request in writing made to the President of a College or to his/her designee, the Association or any College-based chapter thereof shall have the right to meet at such College, if appropriate facilities are available. All requests must be received at least twenty-four (24) hours prior to the time requested for the meeting.

3.02 Bulletin Boards

The President of each Colarge or his/her designee shall assign at least two (2) bulletin boards for the exclusive use and responsibility the Association for the purpose of posting Association notices concerning the administration of the provisions of a Agreement.

3.03 Intra-College Mail

The Association shall be permitted to use the intra-college mail system for the distribution of . ssociation communications. All notices so placed shall bear the signature of the President of the Association or his/her designee(s).

3.04 Unit Facilities and Services

The Board shall make reasonable efforts to maintain at each College:

- 1. Office space currently being used or office space that may become available for the Association for on-campus contract administration;
- 2. Existing furnished employee lounges, restroom and eating facilities;
- 3. Existing assigned space and necessary equipment to carry out their assigned duties;
- 4. Existing parking facilities;
- 5. Existing telephones where currently operating;
- 6. Existing custodial, secretarial and technical assistance.

The President of the College or his/her designee shall constit with the Chapter President of the Association or his/her designee prior to altering the level of existing facilities or services listed herein and currently in use by unit members.

3.05 Access to College Facilities

The Board agrees to make available to unit members access to their work areas and offices on the same basis and to the same extent as such facilities are made available to other College employees, subject, however, to applicable Board policy, rules and regulations governing access to College facilities.

## ARTICLE IV - THE RIGHTS & RESPONSIBILITIES OF THE BOARD

- 4.01 All management's rights and functions, except those which are clearly and explicitly abridged by the specific terms of this Agreement, shall remain vested within the Board. These exclusive rights include, but are not limited to the following:
  - 1. To the executive management and administrative control of its Colleges and their properties and facilities;
  - 2. To hire all employees, to determine their qualifications and the conditions for their continued employment or their dismissal or demotion, and to promote and transfer all such employees;
  - 3. To establish courses of instruction, including special programs and to provide for athletic, recreational and social events for students, all as deemed necessary or advisable by the Board.
  - 4. To decide upon the means and methods of instruction; the duties, responsibilities, and assignment of teachers and other employees with respect to administrative and non-teaching activities; and the terms and conditions of employment:
  - 5. To establish the standards of productivity of its employees; and
  - 6. To establish policies, rules and regulations and practices in carrying out its responsibilities.
- 4.02 The exercise of the foregoing powers, rights, authority, duties and responsibilities by the Board, the adoption of policies, rules, regulations and practices in furtherance thereof, and the use of judgment and discretion in connection therewith shall be limited only to the specific and express terms of this Agreement, and then only to the extent such specific and express terms hereof are in conformance with the Constitution and laws of the Commonwealth of Massachusetts and the Constitution and laws of the United States. Nothing contained in this Agreement shall be deemed or construed to impair or limit the powers and duties of the Board under the laws of the Commonwealth.
- 4.03 It is understood that the matters contained in this Article are not subject to the grievance and arbitration procedures in this Agreement, except as to the limitation stated in this Agreement or unless it can be shown that in the exercise of these rights the Board acted unreasonably and to the detriment of employee rights.



## ARTICLE IV-A - COLLEGE GOVERNANCE

- 4 A.01 The Board recognizes the importance of the advisory role of unit members in matters of College governance, including the improvement and development of academic programs and resources. The Board and the Association recognize that advisory organizations currently exist within the Community College System and that the structures and procedures established by such bodies vary among the Colleges in response to differing conditions, interests and needs of each College. Such advisory organizations or similar organizations shall be maintained or created at each College, after consultation with the Association, to insure advisory comment from unit members and other constituencies of the College. A governance structure shall provide for an open forum for discussion and information sharing for the purpose of providing the President of the College with advisory input prior to the promulgation of College policy.
- 4 A.02 The Board recognizes the importance of the role of unit members in the selection process of unit members; provided, however, that unit members shall also participate in the selection process of administrators where they are currently allowed to do so pursuant to an established and continuous college-wide past practice or written College rule; provided further that the President of the College shall select unit members to serve on selection committees for these purposes after first consulting with the members of the appropriate division/department/work area.

## ARTICLE V - MAINTENANCE OF RECORDS

- 5.01 Each Community College shall maintain an official personnel file for each unit member, which shall be the personnel file consulted when making all personnel decisions and recommendations. Any-and all material contained in this personnel file concerning the unit member shall be open to the unit member with a right to copy at his/her expense, upon written request and by appointment during regular business hours.
  - The unit member shall be responsible for supplying the administration with all of the necessary documents requested by the administration in order to complete the unit member's files. The cost of supplying any materials after the unit member has complied with the aforementioned request shall be paid by the Board.
  - If any additional material is included within a unit member's files after the effective date of this Agreement, the unit member shall be sent a copy of such material within seven (7) calendar days thereafter.
  - The unit member shall have the right to file a statement in response to any written documents placed in his/her files.
- 5.02 Each College shall maintain a grievance file separate from the official personnel file.
- 5.03 The Colleges shall maintain the confidentiality of these files in accordance with state and federal law.
- 5.04 Whenever any individual or individuals inspect the official personnel file of a unit member, the date and name of the individual or individuals conducting such investigation shall be noted in the file.

## . ARTICLE VI - DEDUCTION OF DUES AND AGENCY FEES

#### 6.01 Dues Deductions

- A. Payroll deductions for membership dues of the Association shall be authorized pursuant to applicable laws of the Commonwealth.
- B. The Association shall, at least thirty (30) days prior to the beginning of the academic year or within thirty (30) days of the effective date of this Agreement, give written notice to the Board of the amount of dues which are to be deducted on behalf of the Association and the names of the treasurers of the local affiliates of the Association.

#### 6.02 Agency Service Fee

As a condition of employment during the term of this Agreement, every member of the bargaining unit who is not also a member of the Association shall pay or by payroll deduction shall have paid to the said Association an agency service fee that shall be one hundred per centum (100%) of the then current rate of dues payable by a member of the said Association; provided, however, that no such payment or deduction shall include any amount that represents a cost not related by the provisions of Section 3 of Article IX of the Rules and Regulations Relating to the Administration of Chapter 150 E of the General Laws as such rules and regulations have been promulgated, and as they may be amended from time to time by the Labor Relations Commission. Such ree so required to be paid shall be payable on or after the thirtieth (30th) day next following the beginning of employment of such member of the bargaining unit or on or after the thirtieth (30th) day next following the effective date of this Agreement, whichever shall be later.

Such fee may be paid by payroll deduction if so authorized pursuant to an Agency Service Fee Deduction Authorization; provided, however, that such authorization shall be deemed to have effect only with respect to such sum as is herein provided.



Any other provisions of this Agreement to the contrary notwithstanding every unit member who shall have failed to fulfill the condition of employment as is herein prescribed shall be subject to immediate dismissal and shall be so dismissed by the Board with effect no later than the end of the semester during which the Board shall have acted to dismiss him/her in accordance with the provisions of this Article; provided, however, that such dismissal shall be effected by notice promptly issued by the Board or its designee to such unit member after the Association shall have notified the President that such unit member has not fulfilled the condition herein prescribed. The said notice shall be sent by registered mail, return receipt requested, and shall give such unit member fourteen (14) days from the date of its receipt to fulfill the said condition. Within the said fourteen (14) days, the Board or its designees shall grant such unit member such opportunity to respond to the said notice as the Board may from time to time prescribe for the purposes of this provision.

Whenever such unit member shall not have fulfilled the condition herein prescribed within the said fourteen (14) days, the Boszá shall act to dismiss him/her at its meeting next following the expiration of the said fourteen (14) days; provided, however, that the Board need not so act if such unit member fulfills the said condition prior to the date of such meeting.

The Association shall reimburse the Board for any expenses incurred as a result of being ordered to reinstate an employee terminated at the request of the Association for not paying the agency fee. The Association shall intervene in and defend any administrative or court litigation with respect to any unit memoer's refusal or failure to pay the agency service fee, including, but not limited to, actions or claims arising from an employee's termination or dismissal. In such litigation, the Board shall have no obligation to defend the Association and the Association shall indemnify and hold the Board harmless from any loss occasioned by such litigation.

Any unit member who pays an agency service fee may on request obtain a rebate of part of his/her agency service fee in accordance with General Laws, Chapter 150E, Section 12, and Association procedures:

## ARTICLE VII - ACADEMIC FREEDOM & RESPONSIBILITY

#### 7.01 Academic Freedom

The Board and the Association endorse the principles and standards of academic freedom and academic responcibility as generally and traditionally accepted in institutions of higher education. The parties agree to promote public understanding and support of academic freedom and agreement upon procedures to assure academic freedom in Colleges and universities. Institutions of higher education are conducted for the common good and not to further the interest of either the individual teacher or the institution as a whole. The common good depends upon the free search for truth and its free exposition. Academic freedom is essential to these purposes and applies to both teaching and research.

Academic freedom is the right of scholars in institutions of higher education freely to study, discuss, investigate, teach, exhibit, perform and publish. Freedom in research is fundamental to the advancement of truth. Academic freedom in its teaching aspect is fundamental for the protection of the rights of the teacher in teaching and of the student to freedom in learning.

The teacher is entitled to full freedom in research and in the exhibition, performance and publication of the results of his/her research, to full freedom in the classroom in discussing his/her subject, and most specifically in the selection of his/her classroom materials, including the selection of texts. The teacher is entitled to discuss controversial issues. As both a teacher and scholar, he/she recognizes his/her professional obligation to present various scholarly opinions and to avoid presenting totally unrelated materials, that being fundamental to the advancement of truth. A faculty member has the right to determine the amount and character of the work and other activities he/she pursues outside the College; provided such work and other activities do not interfere with the discharge of his/her responsibilities under the terms of this Agreement. Unit members are entitled to freedom of expression of political belief or affiliation.

#### 7.02 Academic Responsibilities

Academic freedom carries with it correlative responsibilities.

The faculty member has the responsibility to his/her colleagues and the College community to preserve intellectual honesty in his/her teaching and his/her research. He/she respects the free inquiry of his/her associates and avoids interference in their work.

The parties recognize that adherence to the complementary concepts of academic freedom and academic responsibility will most nearly ensure that the greatest contributions to the several Colleges will be made by their most valuable resource, the faculty. The College or university teacher is a citizen and a member of a learned profession affiliated with an educational institution. When he/she speaks or writes as a citizen, he/she should be free from institutional censor-hip or discipline, but his/her special position in the community imposes special obligations. As a person of learning, affiliated with an educational institution, he/she should remember that the public may judge his/her profession and his/her utterances. Hence, he/she should at all times be accurate, should exercise appropriate restraint, should show respect for the opinion of others, and should make every effort to indicate when he/she is not an institutional spokesperson.

Institutions of higher education are committed to the search for truth and knowledge and to contributing to the solution of problems and controversies by the method of rational discussion.



### ARTICLE VIII - AFFIRMATIVE ACTION

- 8.01. The Board and the Association recognize and affirm their commitment to the policy of non-discrimination, equal opportunity and affirmative action in all aspects of employment, including, but not limited to, recruitment, selection, placement, tenure, wages, training, retrenchment, promotion and termination. The Board and the Association recognize that the realization of equal opportunity shall be based on their mutually cooperative good faith efforts to achieve full and promps utilization of women and minorities through the Board's Affirmative Action Policy.
- 8.02 The Board agrees that it will not alter or change any provision of this collective bargaining agreement by the implementation of any Affirmative Action Policy.
- 8.03 The Board and the Association agree to cooperate in the administration of the Board's 1986-89 Affirmative Action Policy and in the implementation of any Affirmative Action Policy as may be adopted by the Board based on or mandated by federal or state law; the Board further agrees that any Affirmative Action Policy which may be developed by it shall be subject to the provisions of Chapter 150 E. The Board agrees further to bargain with the Association concerning any impact an Affirmative Action Policy required by federal or state law may have on matters covered by Chapter 150 E as it relates to the Association as the exclusive collective bargaining representative for all categories of employees described in Appendix A of this Agreement.

## ARTICLE IX - SUPPLEMENTAL BENEFITS

#### 9.01 Authorized Leaves with Pay

#### A. Sick Leave

#### 1. Entitlement

A'l faculty members shall be entitled to ten (10) days of sick leave for each academic year of service. All professional staff members shall be entitled to fifteen (15) days of sick leave for each year of service. Sick leave credit shall begin with the first (1st) full month of employment and accumulate monthly as follows:

- a. Faculty members shall accumulate at the rate of one and one-ninth (1 1/9) days of sick leave for each fall month of employment.
- b. Professional staff members shall accumulate at the rate of one and one-quarter (1%) days of sick leave for each full month of employment.

Credits for periods of less than one (1) full month's employment shall not be allowed. Sick leave not used in any year may be accumulated. No person shall be entitled to a leave of absence with pay on account of sickness in excess of the accumulated sick leave then due, except as provided under Section 9.01 A4 – Sick Leave Bank.

#### 2. Reinstatement

Unit members who are reinstated shall be credited with sick leave credits as have accrued at the termination of their previous service. No credit for previous service may be allowed where reinstatement occurs after an absence of three (3) years or more from the date of termination of their previous service unless approval of the Board is secured for any of the following reasons:

- a. Illness of such person and not because of illness of his/her immediate family;
- b. Dismissal through no fault or delinquency attributable to such person; or
- c. Injury while in the service of the Commonwealth in the line of his/her duties for which such person would be entitled to receive Workmen's Compensation benefits.

#### 3. Use of Sick Leave

Sick leave shall be granted at the sole discretion of the President of the College under the following conditions:

- a. When a unit member cannot perform his/her duties because he/she is incapacitated by personal illness or injury;
- b. When, through exposure to contagious disease, the presence of the person at his/her post of duty would jeopardize the health of others;
- c. In case of serious illness of husband, wife, child, parent of either spouse or of any other person subject to these rules, or of any person living in the immediate household of a person subject to these rules, he/she may be granted sick leave with pay not to exceed seven (7) working days within the fiscal year.
- d. A unit member shall be entitled to sick leave with pay, not to exceed ten (10) consecutive working days, in the event his wife gives birth.

Notification of absences shall be given as early as possible on the first (1 st) day of absence. If such notification is not made, such absence may, at the discretion of the College President or his/her designee, be applied to absence without pay. For any period of absence on account of sickness, the College President or his/her designee may require a physician's certificate proving the necessity of such absence. If such certificate is not filed within seven (7) calendar days after a request therefor, such absence may be applied at the discretion of the College President or his/her designee to absence without pay.



#### 4. Sick Leave Bank

- a. Upon the date of execution of this Agreement, there shall be established, or continued in the case of those Colleges having previously so established, a Sick Leave Bank.
- b. During the term of this Agreement, a unit member who is not a member of the Sick Leave Bank may become so during the month of October by giving written not te to the President of the College or his/her designee that he/she is assigning to the Board one (1) day of his/her personal sick leave accumulation.
- c. The President of the College or his/her designee shall maintain a register of the membership and the number of sick leave days accumulated in the Bank.
- d. Five (5) working days after the exhaustion of his/her personal sick leave accumulation, every member of the Sick Leave Bank shall be entitled to draw upon the Sick Leave Bank, effective thereafter upon notice to the President of the College. The granting of such sick leave shall be subject to the same criteria as regular sick leave days and shall be in all other respects consistent with Board policy; provided, however, that such sick leave shall be available only for the illness of the employee and not for the illness of the family.
- e. Whenever the accumulation of sick leave days in the Sick Leave Bank shall have fallen to fifty (50) days, the President of the College or his/her designee shall notify all members. Thereafter, one (1) ersonal sick day from each member's accumulated sick leave shall be assigned to the Bank unless a member notifies the President of the College or his/her designee in writing within five (5) days after receipt of said notice that he/ she does not wish to remain a member; provided, however, that any member of the Sick Leave Bank wishing to remain a member thereof and who shall have exhausted his/her personal sick leave accumulation on the date of the giving of such notice, shall assign such additional days within fifteen (15) days after the date on which such member is entitled to personal sick leave and shall retain all his/her rights in the Bank until such period of assigning an additional day shall have expired.

No unit member may draw upon the Sick Leave Bank in excess of the number of days to which he/she is entitled as determined by the administration.

A unit member who receives disability compensation provided by statute and who is entitled to any personal sick leave allowance may take so much of his/her personal sick leave allowance payment which, when added to the amount of disability compensation provided by statute, shall result in the payment to him/her of his/ her full salary. The Sick Leave Bank shall not be used for this purpose.

### 5. Certification and Notification of Sici Leave Status

The President of the College or his/her designee shall notify every unit member on or before September 15th of each year of the number of sick days accumulated to his/her credit as of September 1.

Unit members shall be paid twenty (20) percent of the value of their unused accrued sick leave at the time of their retirement; provided, however, that such payment shall not enlarge or diminish those pension benefits for which a unit member would otherwise be entitled. Upon the death of a unit member an amount equal to twenty (20) percent of the value of his/her unused accrued sick leave shall be paid to his/her estate.

Whenever because of prolonged illness, the accumulated sick leave of a unit member is expended, the unit member may request from the Board an extension of sick leave benefits for the purposes provided in this Article.

#### B. Bereavement Leave

Upon notification to the President of the College or his/her designee of the death of the husband, wife, child, parent of either spouse, brother or sister of a person subject to this Agreement, or of a person living in the immediate household of a person subject to this Agreement, leave of absence with pay may be granted for a period not exceeding four (4) consecutive work days.

#### C. Military Leave

- 1. Every unit member shall be entitled, during the term of his/her service in the armed forces of the Commonwealth, under Sections 38, 40, 41 or 60 of Chapter 33 of the General Laws, or during his/her annual tour of duty, not exceeding seventeen (17) days as a member of a reserve component of the armed forces of the United States, to receive pay therefor, without loss of his/her ordinary remuneration and shall in addition be entitled to all leaves of absence provided under this Agreement.
- 2. Any unit member who is a member of a reserve component of the armed forces of the United States and who is called for duty other than the annual tour of duty not exceeding seventeen (17) days shall be subject to the provisions of Chapter 708 of the Acts of 1941, as amended, or of Chapter 805 of the Acts of 1950, as
- 3. Any unit member, who on or after January 1, 1980 shall have tendered his/her resignation or otherwise terminated his/her employment for the purpose of service in the military or naval forces of the United States, and who does or did serve or was or shall be rejected for such service, shall, except as otherwise provided by Chapter 708 of the Acts of 1941, as amended, be deemed to be or to have been on military leave, and no such unit member shall be deemed to have resigned or to have terminated his/her employment until the expiration of two (2) years from the termination of said military or naval service by him/her.

#### D. Court Leave

1. Unit members who are called for jury duty or are summoned to appear as witnesses on behalf of any town, city,



county, state or federal government shall be granted court leave. Notice of service shall be filed with the President of the College or his/her designee upon receipt of summons.

- 2. If jury or witness fees received by a unit member amount to more than his/her rate of compensation, he/she may retain the excess of such fees and shall submit the regular rate of compensation together with a court certificate of service to the Board or its designee, and shall be deemed to be on leave of absence with pay. If the jury or witness fees amount to less than the unit member's regular rate of compensation, he/she shall be deemed to be on leave of absence with pay and he/she shall remit said fees to the Board or its designee with a court certificate of service.
- 3. Expenses submitted by the court for travel, meals and room hire shall be retained by the unit member and shall not be considered part of the jury or witness fees.
- 4. Whenever a unit member is called for jury duty or summoned to appear as a witness and such jury duty of appearance occurs during his/her vacation, there shall be no necessity to account to the Board or the College for any fees received during such period.
- 5. When a unit member has been granted court leave for jury or witness service, and is excused by proper court authority, he/she shall report back to his/her official place of duty whenever the interruption in jury or witness service will permit four (4) or more consecutive hours of employment.
- 6. Court leave shall not affect employment rights of unit members.
- 7. Court leave shall not be granted when a unit member is the defendant or is engaged in personal litigation.

#### E. Other Leaves

Such other leaves as are herein authorized shall also be available to unit members, provided that whenever the granting of any such leave is discretionary, such discretion shall be exercised by the President of the College or his/her designee. Leave of absence with pay may be granted for the following reasons:

- 1. To permit unit members who are veterans to pay tribute at the funeral in Massachusetts of dead veterans. The President of the College or his/her designee shall grant leave of absence with pay to veterans who are members of firing squads, color details, pall bearers, buglers or escorts participating in such service.
- 2. Unit members shall be entitled to leave of absence with pay to allow for loss of time due to prophylactic inoculation required as a result of their employment. If such absence with pay exceeds one (1) week, the Board of its designee shall, if legally appropriate and as soon as practicable, initiate a Workmen's Compensation claim and further payments because of such prophylactic inoculation shall cease.
- 3. Unit members shall be entitled to leave of absence with pay to allow for the period of absence due to quarantine because of exposure to contagious disease in the regular performance of duty.
- 4. Red Cross blood donations.
- 5. Oral, written and physical examination for state service conducted by the Division of Civil Service or the Division of Registration; and physical examination for state retirement.
- 6. To attend hearings in Industrial Accident cases as the injured person or as a witness therein. Any witness fees received by such injured person or witness shall be paid to the Board.
- 7. To allow aftendance by unit members who are delegates or alternates to state or national conventions of the following veterans' organizations: American Legion, AMVETS or World War II, Disabled American Veterans, Legion of Valor, Marine Corps League, Order of the Purple Heart, United Spanish War Veterans, Veterans of Foreign Wars, Reserve Officers Association of the United States, Veterans of World War I of the U.S.A.; Inc.
- 8. Leave of absence with pay not to exceed two (2) hours shall be granted to any unit member, if he/she makes application therefore to permit him/her to vote in the voting precinct, ward or town in which he/she is entitled to vote; provided that the hour of opening and the hour of closing of the polls at such place would otherwise preclude his/her traveling to or from the polls.

## F. Vacation Leave for Professional Staff Members

#### 1. Entitlement

- a. All professional staff members who work a twelve (12) month year shall be entitled to an annual vacation leave of twenty-two (22) days.
- b. The vacation year shall be for the period of July 1 to June 30 inclusive.
- c. Professional staff members having an aggregate of more than one (1) day of leave without pay and/or absence without pay in any calendar month shall not receive a vacation credit for that month.
- d. All professional staff members shall be credited as of the first day of each month with vacation leave with pay accrued prior to that date not to exceed F.2 below.

#### 2. Accrual of Vacation Days

Professional staff members shall accrue vacation days subject to the following provisions:

2. For less than one (1) year of service beginning July 1 and ending on June 30, vacation leave of two (2) days for each full month of service not to exceed twenty-two (22) days.



- b. For one (1) full year's service through six (6) full years of service beginning on the first (1st) working day of July and ending on June 30, vacation leave of 1.83 days for each full calendar month.
- c. For seven (7) full year's service through sixteen (16) full years of service beginning on the first (1st) working day of July and ending on June 30, vacation leave of 1.92 days for each full calendar month shall be granted.
- d. For seventeen (17) full year's service through twenty-four (24) full years of service beginning on the first (1st) working day of July and ending on June 30, vacation leave of 2.08 days for each full calendar month shall be granted.
- e. For a professional staff unit member who has completed twenty-five (25) years or more of service on June 30 of any vacation year shall be granted 2.50 days for each full calendar month for a total of thirty (30) days of vacation leave.

#### 3. Scheduling of Vacation Days

Vacation leave shall normally be scheduled and taken during the vacation year in which it becomes available; provided, however, that a professional staff member may, at his/her discretion, carry over up to sixty-four (64) vacation days from year to year; provided further that in no event shall vacation credits in excess of sixty-four (64) days be carried over for more than one (1) year unless otherwise approved by the Board. Unused vacation days in excess of sixty-four (64) days shall be converted to sick leave at the end of each fiscal year.

Persons who are eligible for vacation under these rules, whose services are terminated by dismissal through no fault or delinquency of their own, by retirement, or by entrance into the armed forces, shall be paid an amount equal to the vacation allowance as earned but not granted in the vacation year prior to such dismissal, retirement, or entrance into the armed forces, and in addition that portion of the vacation allowance earned in the vacation year during which such dismissal, retirement, or entrance into the armed forces occurred, up to the time of separation; provided that no monetary or other allowance had already been made therefor.

Vacation days may be used in fractions of one-half (1/2) day.

#### 4. Vacation Status Report

The President of the College or his/her designee shall notify each professional staff member on or before July 15 of the number of vacation days accumulated to his/her credit as of July 1 preceding.

#### G. Holiday Pay

All unit members during the term of this Agreement shall be entitled to the following holidays:

New Year's Day
Martin Luther King Day
Washington's Birthday
Evacuation Day
Patriots' Day
Memorial Day
Bunker Hill Day

Independence Day
Labor Day
Columbus Day
Veterans' Day
Thanksgiving Day
Christmas

#### H. Personal Days

#### 1. Entitlement

Each faculty member shall be entitled to two (2) personal days per academic year. A faculty member with an initial employment date after December 31 shall be allotted one (1) personal day for the remainder of the academic year.

Each professional staff member shall be entitled to three (3) personal days per fiscal year. Those professional staff members with initial employment dates between July 1 and September 30 shall be allotted three (3) days; those with initial employment dates between October 1 and December 31 shall be allotted two (2) days, and those with initial employment dates after January 1 shall be allotted one (1) personal day for the remainder of the fiscal year.

#### 2. Usage

A unit member who wishes to use a personal day shall, whenever practicable, submit a request no later than three (3) days in advance. Such request shall not be unreasonably denied. Each faculty member using a personal leave day shall obtain coverage for his/her classes or, where appropriate, assign a self-directed learning experience.

Personal leave days may not be accumulated beyond the termination of the academic year for which they are allotted.

#### I. Sabbatical Leave

#### 1. Purpose

The purpose of a sabbatical leave chall be for professional growth, research or study which may involve travel and which benefits the unit member and the College.

#### 2. Criteria for Sabbatical Leave

The following criteria shall be considered in determining who should be granted subbatical leave.



- a. That the objectives of the sabbatical leave, if attained, would substantially contribute to the professional growth of the unit member.
- b. That the objectives of the sabbatical leave, if attained, would assist the unit member in substantially contributing to institutional needs and attainment of institutional purposes.
- c. That the unit member has the ability to achieve the goals of the project or plan based on his/her past experience and formal educational background.
- d. That the attainment of the objectives of sabbatical leave as proposed are realistic in terms of time, costs and other related variables.
- e There exists independent financial support from other funding sources concerned with the proposed plan or project where College funding sources are otherwise unavailable.

#### 3. Eligibility

- a. Unit members must have completed six (6) years of continuous full-time service to the College to be eligible for consideration for sabbatical leave. Time granted for other types of professional leaves shall not apply to the six (6) year period required for sabbatical leave eligibility. In addition, a unit member must have received a summary evaluation of other than unsatisfactory on his/her most recent evaluation to be considered eligible.
- b. A unit member who may have previously received a sabbatical leave shall be eligible for consideration of a subsequent leave after six (6) years of continuous service upon returning from the original sabbatical leave.

#### 4. Limitation

Sabbatical leave is not an inherent right of the unit member and there is no obligation on the part of the Board to approve any and all sabbatical leave requests made by unit members. No more than six (6) percent of the full-time unit members at a given College shall be granted sabbatical leave in any academic year unless otherwise approved by the Board.

#### 5. Length of Sabbatical

A unit member granted sabbatical leave may be granted one-half (½) year of leave at full or half salary, one (1) full year at half salary, or one-half (½) year or full year at half workload at full or half salary. A year is defined by the duration of the unit member's contract in the year prior to sabbatical.

## 6. Limitations on Sabbatical Stipend/Outside Employment

The Board encourages unit members to seek additional funds which would enhance directly the purpose of the sabbatical leave.

- a. Non-salary funds (travel, materials, and such other types of expenses) obtained from outside sources directly related to the purpose of the sabbatical leave shall not reduce the sabbatical leave stipend.
- b. Salary funds obtained from outside sources which directly relate to the attainment of the specific objectives of the sabbatical leave shall not reduce the sabbatical leave stipend unless the total amount of outside salary funds and sabbatical stipend exceed the current annual salary of the unit member or its prorated equivalent in the carrof a half-year sabbatical leave. In no instance shall any activity required to earn or receive this outside salary interfere directly or indirectly with the purpose of time available for which the sabbatical leave was granted.
- c. Salary funds derived from outside source(s) which in addition to the sabbatical stipend exceed the current salary or its proration and which interfere with the purpose of the sabbatical leave shall result in the sabbatical stipend being reduced an equal amount so as to result in an amount equal to the current salary or its prorated equivalent.
- d. Failure to report additional outside salary shall result in the immediate reduction in a like amount of the sabbatical stipend if the leave is still in process. Should funds remaining in the sabbatical stipend be inadequate to liquidate the additional unreported remuneration or should the sabbatical leave have been completed, the unit membershall return a like amount to the College within six (6) months of the discovery of the unreported forms of remuneration above.

#### 7. Post-Sabbatical Service

Prior to the granting of such leave, said unit member shall enter into a written agreement with the Board that upon the termination of such leave he/she shall return to the service of the Board and serve as a unit member within the System for a period of one (1) year, and that in default of completing such service, he/she shall refund to the Commonwealth, unless excused therefrom by the Board for reasons satisfactory to it, an amount equal to such proportion of the salary as the amount of service not actually rendered as agreed bears to the whole amount of services agreed to be rendered.

#### 8. Process

a. The applicants for sabbatical leave shall submit their proposed plan on a standard form concurrently to their immediate supervisor and the standing Sabbatical Leave Committee which shall be appointed annually by the President of the College. Applications must be submitted no later than seven (7) months prior to the starting date of the leave.



- b. The President of the College shall annually invite recommendations for appointment to the Sabbatical Leave Committee. In determining the membership of the Sabbatical Leave Committee, the President of the College shall appoint, insofar as possible, unit members who have successfully completed sabbatical leaves and who represent the unit membership. Unit membership on said Committee shall be proportionally representative of the faculty and professional staff members eligible to apply for sabbatical leave. The Committee size should be of an uneven number not to exceed seven (7) nor be less than five (5). A unit member shall not serve on the Committee during a year in which he/she applies for sabbatical.
- c. The Committee shall rank each eligible candidate who has applied for sabbatical leave in terms of the criteria stated for sabbatical leave. The recommendations of the Sabbatical Leave Committee in rank order shall be concurrently submitted to the Dean(s) and the applicants no later than six (6) months prior to the starting date of the leave. The immediate supervisor shall forward his/her recommendation to the appropriate Dean with a plan for coverage of the sabbaticant's classes, services, or other professional responsibilities by other unit members and, where appropriate, an estimate of the cost of replacement relative to existing budgeted salaries. Said recommendations shall be forwarded no later than six (6) months prior to the starting date of the leave.
- d. The Dean(s), acting as a committee, shall consider the recommendations of the Sabbatical Leave Committee and the immediate supervisor and recommend to the President of the College candidates for sabbatical leave with a plan for coverage of the sabbaticant's classes, services, or other professional responsibilities by unit members and, where appropriate, a final estimate of the replacement costs. Copies shall be transmitted to the applicant, the applicant's immediate supervisor and the Chairperson of the Sabbatical Leave Committee.
- e. The President of the College shall evaluate the recommendations of the Sabbatical Leave Committee and the Dean and recommend sabbatical leave for those unit members who he/she deems rank highest on the criteria; provided, however, that if the President's recommendation differs from that of the Sabbatical Leave Committee, the President of the College shall in addition attach his/her recommendation and a written statement of reasons therefore which shall be based on applicable criteria established for the granting of sabbatical leaves. Said recommendations shall be forwarded to the Board no later than five (5) months prior to the start of the leave.
- f. The Board shall notify an applicant, where practicable, no later than three (3) months prior to the start of the leave whether his/her application has been approved.

#### 9. Funding

A unit member shall, if possible, provide coverage for .is/her classes, services or other professional responsibilities for the period during which he/she is on sabbatical leave; provided, however, that if in the determination of the President of the College or his/her designee it is necessary to hire replacement(s) for the sabbaticant's classes, services or other professional responsibilities, such replacement shall to the extent practicable be provided for within the College's existing salary funds by reducing services, classes and/or by hiring part-time replacements.

#### 10. Report of Activity and Accomplishment

A unit member who is granted sabbatical leave shall submit a report of the results of the sabbatical leave within ninety (90) days of the expiration of said leave on such form as may be promulgated by the Board. The report shall include:

- a. An account of activities during the leave, including travel, itineraries, institutions visited and persons consulted.
- b. A statement of progress made on the sabbatical leave as proposed in the application and an explanation of any significant changes made in the program.
- c. An appraisal of the relationship between the results obtained and those anticipated in the sabbatical leave program statement.
- d. A final account of all sources and kinds of salary support, other than the sabbatical stipend, with a description of the sabbaticant's activity, if any, necessary to receive each such type of fund.

#### 11. Impact of Sabbatical on Conditions of Employment

- a. Unit members shall fully participate in the following fringe benefits during the sabbatical year and continue to contribute to those requiring employee contributions:
  - 1. sick leave accumulation
  - 2. leave for death or illness in immediate family
  - 3. term insurance
  - 4. medical insurance
  - 5. retirement based on actual salary paid for year
  - 6. membership in the Health and Welfare Fund.
- b. Unit members shall participate in vacation day accumulation on a prorated basis, determined by their non-sabbatical period of employment during the contractual year.



- e. Unit members may use their sabbatical year as a year of service toward eligibility requirements and may be considered for evaluation purposes in promotion and tenure decisions; provided, however, that a unit member shall not be promoted during the year he/she is on a sabbatical leave.
- d. This section shall not be applicable to non-state appropriated funded unit members; provided, however, that service rendered by such unit members shall be counted on a one (1) year for one (1) year basis in determining eligibility for sabbatical leave after such unit member becomes an "01" employee.

#### 9.02 Unpaid Leaves of Absence

Any unit member granted an unpaid leave of absence shall retain those benefits accrued prior to the period of his/her leave, including seniority; but shall not continue to accrue any benefits while on leave. Upon returning from leave such unit member shall be placed upon the salary schedule at the step and rank he/she held prior to the leave, as adjusted by the provisions of any collective bargaining agreements between the Board and the Association. There shall be no promotion nor shall there be any entitlement to any performance based awards that might have become available during the leave of absence. This clause shall apply to all unpaid leaves of absence subject to the exception cited in Section (d.) below, unless as otherwise provided herein. Unpaid k aves shall not be unreasonably denied.

#### A. Professional Leave

#### 1. Purpose

Upon the application of a unit member and a recommendation of the President of the College, the Board or its designee may grant to such unit member leave without pay for up to three (3) years for professional reasons as provided herein. The purpose for which a unit member may submit his/her application for such unpaid leave shall include, but shall not be limited to:

- a. advanced study;
- b. participation in a program of exchange teaching in accordance with the conditions set forth under the provisions of 9.02 D;
- c. participation in a program related to his/her professional responsibilities;
- d. service as an officer or staff member of any recognized professional organization;
- e. service in a public office to which he/she has been elected or appointed and for such other purposes as may be allowed under the laws of the Commonwealth.

#### 2. Eligibility

Unit members shall be eligible for such leave after six (6) years of full-time service. In addition, the unit member must have received a summary evaluation of other than unsatisfactory on his/her most recent evaluation to be considered eligible.

#### 3. Conditions

The granting of a professional leave shall be subject to the following conditions:

- a. the applicant's professional duties permit his/her absence for the period of time requested; and/or
- b. the leave is of value to the individual and to the College as determined by the President of the College or his/her designee.

#### 4. Procedure

Applicants shall prepare a proposal which describes the prospective activity and indicates the contribution it will make to the individual concerned and the College. The proposal shall be submitted to the appropriate Dean six (6) months in advance of the requested leave; provide 1, however, that the President of the College or his/her designee may waive the six (6) month application period. Prior to making his/her recommendation to the President of the College, the Dean upon request shall first inform the applicant of his/her intended recommendation. The Dean shall forward his/her recommendations to the President of the College, with a copy to the unit member within thirty (30) days of receipt.

The President of the College shall notify the unit member and the appropriate Dean of his/her decision and shall forward his/her recommendation to the Board within sixty (60) days of receipt.

#### 5. Continuation

Any unit member requesting a continuation of said leave for any succeeding period beyond that initially granted shall do so at least one (1) semester prior to the expiration of said leave.

#### B. Maternity Leave

#### 1. Entitlement

A female unit member who is employed by the Board and who has given notice, when possible, at least three (3) months prior to her anticipated date of departure is entitled to be absent from such employment for a period certified by her physician due to disabilities caused or contributed to by pregnancy and recovery therefrom. Sick leave may be utilized for any period of disability; provided that the President of the College or his/her designee may require that her physician certify that any period in excess of five (5) days to which sick leave is applied is medically necessary as a result of or to recover from said disability.



#### 2. Conditions

Disabilities caused or contributed to by pregnancy, abortion, miscarriage, childoirth, and recovery therefrom shall be treated like any other temporary disability. After all employee-designated sick and vacation leave have been exhausted, a maternity leave of up to twelve (12) months without salary shall be granted, except that adjustments in the duration of the leave may be made by the President of the College or his/her designee to insure that such leave is least disruptive of the instructional progress of students.

#### 3. Reinstatement

The unit member shall be restored, subject to retrenchment and any other provisions resulting in faculty or professional staff member reductions, to the same or substantially similar position with the same salary and fringe benefits which she had attained at the time maternity leave was granted, plus any adjustments made as a result of collective bargaining; provided that the unit member returns to full-time service within twelve (12) months from the beginning of the leave or within twelve (12) months from the beginning of any additional leave granted by the President of the College or his/her designee. The President of the College or his/her designee may allow a unit member up to twelve (12) months additional leave; provided, however, that such decision shall not be grievable. For up to twelve (12) months from the beginning of maternity leave, part-time return to service may be arranged by approval of the President of the College or his/her designee when such can be reasonably accommodated; provided, however, that in the case of an employee holding a terminal appointment, a leave shall not extend beyond and provisions for re-employment shall not apply beyond the termination date of the appointment unless the employer agrees in writing to such extension; provided further that such decision to allow part-time return shall not be grievable.

#### C. Family Leave

#### 1. Entitlement

Upon written application to the President of the College, including a statement of reasons, any full-time unit member who has been employed at least one (1) year and who has given notice at least one (1) semester prior to his/her anticipated date of departure, unless otherwise approved by the President of the College or his/her designee, 'hall be granted parental or adoptive leave without pay from such employment for a period not exceeding one (1) academic year. The President of the College or his/her designee may grant a unit member a half-time leave with full benefits; provided, however, that such decision shall not be grievable.

#### 2. Purposes

The purpose for which a unit member may submit his/her application for such unpaid leave shall be the need to care for or to make arrangement for the care of an unemancipated minor child of the unit member, spouse of the unit member or parent of the unit member.

#### 3. Reinstatement

A unit member shall be restored, subject to retrenchment and any other provisions resulting in faculty or professional staff member reductions, to the same or a substantially similar position with the same salary and fringe benefits which he/she attained at the time family leave was granted, plus any adjustments made as a result of collective bargaining; provided that the unit member returns within one (1) academic year from the beginning of the leave or within one (1) year of the beginning of any additional leave granted by the President of the College or his/her designee. The President of the College or his/her designee may allow a unit member up to one (1) academic year additional leave; provided, however, that such decision shall not be grievable.

#### D. Exchange Teaching

- 1. Notwithstanding any other provisions of this Article to the contrary, a leave of absence of up to two (2) years may be granted to any faculty member upon application for the purpose of participation in exchange teaching programs in other states, territories, countries, or an educational or cultural program related to his/her professional responsibilities. On return from such leave, a faculty member shall be placed at the salary held by him/her at the time the leave commenced, except as provided in (2) below.
- 2. In the event there is in the judgment of the President of the College or his/her designee a valid performance evaluation conducted by the host College, the substance of that evaluation sh. be considered as evidence of performance valifying the faculty member for possible award of a performance based award.
- 3. Each faculty member shall submit a certified statement from the appropriate academic administrator of the host college before the first (1 st) semester of his/her return describing in detail the exchange teaching responsibility and/or educational/cultural experience.

#### E. Professional Staff Leave

Professional staff members may, upon mutual agreement between the professional staff member and the President of the College or his/her designee, he granted professional leave without pay for a continuous period of not less than two (2) weeks nor more than eight (8) weeks during the months of June, July and August; provided, however, that unit members granted such leave shall not be denied eligibility for promotion, sabbatical or professional leave, tenure or performance based awards.



## 9.03 Insurance and Other Benefits

## A. Health and Accident Insurance

Unit members shall continue to be covered under the State's Group Health and Accident Insurance plan currently in effect pursuant to the provisions of Chapter 32 A of the General Laws as amended or as such plan may be made available under applicable law of the Commonwealth.

Unit members shall continue to be covered under the State's Retirement Plan pursuant to the provisions of General Laws, Chapter 32, or such plan as may be made available under applicable laws of the

## C. Workmen's Compensation

Unit members shall continue to be covered under the provisions of the State Workmen's Compensat on Act, pursuant to General Laws, Chapter 152.

#### D. Tax · Sheltered Annuities

The Board shall continue to provide for the purchase of tax-sheltered annuities by unit members pursuant to the provisions of General Laws, Chapter 15, Section 18A, or Chapter 15A.

The spouse and/or child or children, including any adopted or stepchild or children of any unit member, who after the date of execution of this Agreement, shall have been admitted as a student in the regular day program at any College in the Community College System shall be entitled to matriculate as a student in such program without payment of any tuition save as is provided in Massachusetts General Laws; provided, however, said tuition free enrollment within the Community College System shall be limited to the day division only.

In addition, effective for the semester following the date of execution of this Agreement, each unit member and the spouse and dependent child or children shall be eligible for system-wide tuition remission as follows:

- 1. For enrollment in any state-supported course or program at the undergraduate or graduate level at any Community College, State College, or University excluding the M.D. Program at the University of Massachusetts Medical School, full tuition remission shall apply;
- 2. For enrollment in any non-state supported course or program offered through continuing education, including any community service course or program at any Community College, State College or University, fifty percent (50%) tuition remission shall apply;
- 3. Tuition remission shall apply to non-credit as well as credit-bearing courses;

in accordance with all the provisions and conditions of the System-wide Tuition Remission Policy for Higher Education Employees issued by the Chancellor on May 21, 1984. This policy shall remain in effect for the duration of this agreement.

## F. Travel and Conference Expenses

- 1. Travel expenses, including reasonable charges for hotel rooms and registration fees to all approved conferences, meetings, workshops, student activities and any other business of the College which is required and authorized by the President of the College or his/her designee in his/her absence, shall be paid by the Board.
- 2. Whenever use of an employee's private car is necessary and has been authorized by the appropriate administrator, the approved mileage rate of twenty-two (22) cents per mile shall be allowed unless a higher rate is otherwise authorized by law and approved by the Board. This approved rate covers all charges, including by way of example garage, parking and toll charges. Whenever a unit member has been assigned, in writing, a second work location other than his/her principal work location by the President of the College or his/her designee, travel shall be paid either for the distance from his/her home to the second work location or from his/ her principal work location to the second work location, whichever is nearer.
- 3. Whenever use of any other mode of transportation is necessary and has been so authorized, the cost of all fares less federal taxes shall be allowed; provided that such receipted bills shall be first submitted for such charges.
- 4. Reimbursement shall not be made for expenses incurred for the sole benefit of the traveler, such as by way of example valet service, entertainment and laundry service.
- 5. When an employee on full travel status is engaged in travel, reimbursement shall be allowed for actual meal expenses incurred; provided that said expenses shall not exceed the following meals allowance schedule:

Meal Breakfast Lunch Supper	Maximum Allowance \$2.50 4.00 7.00	Applicable Period 3:01 a.m. to 9:00 a.m. 9:01 a.m. to 3:00 p.m.
oupper	7.00	3:01 p.m. to 9:00 p.m.

## G. Health and Welfare Fund

The Board and the Association agree to establish a Health and Welfare Fund or Funds which shall be jointly administered by an equal number of trustee members. The trustees' task, among other things, will be the



selection of any additional benefits or the increasing of existing benefits dependent upon the size of the fund.

2. Effective July 1, 1986, the Board agrees to contribute to the Health and Welfare Fund four dollars (\$4.00) each calendar week for each full-time equivalent bargaining unit member.

Effective July 1, 1987, the Board agrees to contribute to the Health and Welfare Fund five dollars (\$5.00) each calendar week for each full-time equivalent bargaining unit member.

Effective July 1, 1988, the Board agrees to contribute to the Health and Welfare Fund six dollars (\$6.00) each calendar week for each full-time equivalent bargaining unit member.

The contributions made by the Board to the Health and Welfare Fund shall be used for health and welfare benefits  $\omega$  be determined by the trustees of the Trust Agreement and to pay the operating and administering expenses of the Fund. The contributions shall be made by the Board in an aggregate sum within forty-five (45) days following the end of the calendar month during which contributions were collected.

- 3. No dispute over a claim for any benefits extended by this Health and Welfare Fund shall be subject to Article X Grievance Procedure.
- 4. It is expressly agreed and understood that the Board does not accept, nor is the Board to be charged with hereby, any responsibility in any manner connected with the determination of liability to any bargaining unit member claiming under any of the benefits extended by the Health and Welfare Fund; such liability shall be limited to the contributions indicated under subsection 9.04(G) above.

### 9.04 Early Retirement Incentive

#### A. Eligibility

Any unit member who has served at least ten (10) years in the community college system, who is eligible to retire under the retirement system of the Commonwealth of Massachusetts, and who is at least fifty-five (55) years of age as of the anticinated date of retirement shall be eligible to receive an early retirement incentive subject to notification provisions below.

#### B. Notification

A Retiree must apply by notifying the President of the College in writing of his/her intent to retire not less than one (1) year in advance of his/her retirement date; provided, however, that this notice requirement may be waived for those who intend to retire within one (1) year after the execution of this Agreement who are otherwise eligible and have applied in writing; provided further that this requirement shall be waived for unit members who have been notified that they will be retrenched or unit members who will retire because of medical reasons consistent with Article 9.01 A.3 who are otherwise eligible and have applied in writing.

#### C. Compensation

An eligible unit member who retires in accordance with the foregoing conditions shall receive an early retirement incentive equal to the applicable percentage of his/her salary as of the date of retirement in accordance with the following schedule:

Early Retirement Incentive as a Percentage of S	alary
---	-------

Age on Date of Retirement	Retirement Date Last Fiscal Day of				
55-60 61 62 63 64	May- August 30 % 25 % 20 % 15 % 10 %	September 25.0 % 20.8 % 16.7 % 12.5 % 8.3 %	October 20.0% 16.7% 13.3% 10.0% 6.7%	November 15.0 % 12.5 % 10.0 % 7.5 % 5.0 %	Dec April 10.0% 8.3% 6.7% 5.0% 3.3%

Payment shall be made after the date of retirement and may be spread over a period not to exceed twelve (12) months as . .ermined by the President of the College or his/her designee.

#### D Maximum Payment

The early retirement incentive and the amount payable to the Retiree in accordance with Article 9.01 A.5 together shall in no case exceed seventy percent (70%) of the Retiree's salary as of the date of his/her retirement.

## E. Special One-Time Bonus for Super-Annuated Unit Members

Any unit member who has attained the age of sixty-five (65) as of the date of execution of this Agreement, and who retires prior to one year subsequent to date of execution of this Agreement, shall be accorded the same rights under this Article as an employee who is sixty-four (64) years old, but less than sixty-five (65) years old as of the late of execution of this Agreement.



#### F. Retrenchment

Any unit member who is retrenched at the age of sixty-five (65) or older shall be accorded the same rights under this Article as an employee who is sixty-four (64) years old.

#### 9.05 Unit Member Resources, Assistance and Protection

The parties are aware that substance abuse is detrimental to the integrity of the College community. In the event unit members are in need of assistance for substance abuse, the parties agine each College campus shall have one unit member designated by the Association to assist and counsel any member seeking assistance. The College administration shall assist the Association where practicable to expedite implementation of any applicable contract provisions and to bring to the attention of the unit member other assistance offered to public employees.

No unit member seeking assistance from the Employee Assistance Program of the Commonwealth or otherwise shall be deprived of employment or any contractual benefits solely as a result of seeking such assistance.

No materials relative to treatment for a substance abuse problem may become part of a unit member's personnel file unless the parties mutually agree otherwise.

#### FORM SUPPLEMENT

IX-1

Application for Sabbatical Leave

Page No.

i

## ARTICLE X - GRIEVANCE PROCEDURE

- 10.01 It is the intent of the parties to this Agreement to use their best efforts to encourage the informal and prompt settlement of grievances which may arise between the Association or a member or members of the bargaining unit and the Board. In recognition of this intent, the parties agree that they shall use the procedure set forth in this Article for the resolution, strictly pursuant to the terms of this Agreement, of all disputes involving the application of this Agreement. For issues involving affirmative action and/or discrimination a unit member is encouraged to proceed under the Board of Regents Affirmative Action Policy, which contains a separate grievance procedure in a forum devoted exclusively to those issues. Filing a grievance under the Affirmative Action Grievance Procedure shall not abrogate the right of a unit member to file a complaint with the appropriate government agency which handles affirmative. crimination matters, i.e. Massachusetts Commission Against Discrimination, Equal Employment Opportunity Commission, etc. The Association further agrees that it shall not initiate proceedings in any other forum in respect of any matter that is or may become the subject of a grievance as hereinbefore defined until it shall have first exhausted the procedures provided herein.
- 10.02 A. Any member (s) of the bargaining unit may initiate and pursue a grievance through the first (1 st) two (2) steps of the grievance procedure without intervention of the Association, provided that a representative of the Association shall be afforded the opportunity to be present at any conferences held; and provided further that any disposition made of any grievance under this Article shall not be inconsistent with the terms of this Agreement. Any member of the bargaining unit may be represented by the Association at any step of the grievance procedure.
  - B. Failure of a grievant (s) to comply with any of the provisions of this Article shall be deemed to be a waiver of the right to seek resolution of the grievance under the terms of this Article. In determining whether there has been any such failure to comply with any of the provisions of this Article, time shall be of he essence and any failure of the grievant to comply with any of the time limits prescribed herein shall be deeme \_ .o be a waiver of the right to seek resolution of the grievance under this Article; provided, however, that the time limits prescribed herein may be extended in any specific instance by mutual agreement of the parties or by oral agreement confirmed in writing.
  - C. The grievant(s) may file a complaint with the Chancellor or his/her designee in the event that the administration fails to comply with any of the provisions of this Article, including time limits. The Chancellor or his/her designee shall determine within fourteen (14) calend ar days whether or not the administration has failed to comply with the provisions of this Article; provided that the administrator who is the subject of a complaint under this paragraph shall not be designated to determine that complaint. No decision shall be rendered prior to providing the grievant an opportunity to express his/her position and, if possible, resolve the issue. If the Chancellor or his/her designee determines that the administration has failed to so comply, then he/she shall state in writing the remedy he/she deems appropriate. Copies of the Chancellor's or his/her designee's decision shall be sent to the Association and the appropriate administrators. The decision of the Chancellor or his/her designee shall be final.
  - D. The resolution of a grievance at any step shall not be deemed to be an admission by the Board or the Association of any violation or breach of the terms of this Agreement or that such grievance is judicially cognizable or legally sufficient pursuant to any applicable provisions of the If the Commonwealth nor shall it establish any precedent nor grant to the Association or any unit mem ling to initiate proceedings or pursue a remedy in any other forum.
  - E. If any member or members of the bargaining unit shall initiate in any other forum or under the College's or Board's Affirmative Action Plan any proceeding that relates to any matter that is the subject of a pending or disposed of grievance or arbitration in respect of which such member or members is or are the grievant(s), such grievance pro-



ceeding shall terminate as of the date of the initiation of such other proceeding and the grievance and arbitration procedures delineated in this Article shall in inapplicable to such grievance; provided that such unit member(s) shall have no further legal recourse under the provisions of this Article.

- F. A grievance may be withdrawn without prejudice at any level.
- G. The Board agrees not to interfere, restrain, or coerce any unit member because of his/her filing a grievance and/or his/her participation in any of the grievance proceedings.
- H. Whenever possible grievance meetings shall be scheduled so as not to interfere with professional responsibilities of individuals involved. If it is necessary to meet with the employer during working hours, the grievant, one (1) Association representative who is a member of the bargaining unit, and necessary witnesses may attend without loss of time or compensation for such meetings.

### 10.03 Disposition of Grievances

# A. Informal Adjustments - Immediate Supervisor

Whenever possible, unit member(s) and the Association shall first attempt in good faith to adjust their grievances with the immediate supervisor or within the College's administrative structure up to the level of the President of the College or his/her designee.

### **B. Formal Adjustments**

All complaints shall be filed on a standard form. If the grievances involves an action by the Board or a matter of general system-wide applicability, the complaint shall be filed at Step Two within thirty (30) calendar days after the grievant knows, or should have known, of the alleged acts or condition on which the grievance is based; provided, however, that the President of the Association or his/her designee must first certify in writing that the grievance is of general system-wide applicability. Within ten (10) calendar days of receipt of said certification, the Chancellor or his/her designee shall determine whether or not the grievance is of system-wide applicability or applies to the Board. If the Chancellor or his/her designee determines that the grievance is of system-wide applicability or applies to the Board, he/she shall meet and confer with the grievant for the purpose of resolving the grievance and shall render a written decision within fifty (50) calendar days after receipt of certification.

If determined not to be of system-wide applicability or applicable to the Board, the unit member (s) or the Association shall have fourteen (14) calendar days from receipt of the hearing officer's determination in which to file the grievance at Step One. The Chancellor or his/her designee may upon mutual consent consolidate at Step Two multiple grievances which volve the same issue. All other grievances shall be processed in accordance with the following procedures:

# 10.04 Step One - President of the College or His/Her Designee

Within thirty (30) calendar days after the grievant knows or should have known of the alleged act or omission on which the grievance is based, the grievant shall present the complaint and all evidence upon which he/she relies or intends to rely as supporting his/her claim for relief to the President of the College or his/her designee. The Complaint shall state all issues and contract violations upon which the grievance is based. The President of the College or his/her designee shall meet with the grievant for the purpose of resolving the grievance and shall, within thirty (30) calendar days of receipt of the written complaint and evidence, render his/her decision and reasons therefor in writing to the grievant.

### 10.05 Step Two - Chancellor

If the grievance is not resolved at Step One or the written decision of the President of the College or his/her designee is not rendered within the time specified, the grievant may then appeal the decision in writing to the Chancellor or his/her designee (on a standard form) with a copy to the President of the College or his/her designee. The appeal shall be filed within ten (10) calendar days of receipt of the President's or his/her designee's decision or the end of the time specified in Step One for said decision, whichever is sooner. Upon receipt of the appeal, the President of the College or his/her designee shall submit to the Chancellor or his/her designee the grievance form filed at Step One, all evidence introduced to date, and copies of all decisions. No further issues or contract violations may be added subsequent to the Step Two hearing; provided, however, that the grievant shall have an opportunity to provide rebuttal evidence.

Failure to so file with the Chancellor or his/her designee within the time specified shall be deemed to be acceptance of the decision rendered at Step One. The Chancellor or his/her designee shall meet with the grievant for the purpose of conducting a hearing and shall render a written decision within forty (40) calendar days of receipt of the appeal.

### 10.06 Step Three - Arbitration

- A. Within forty (40) calendar days after receipt of the decision of the Chancellor or his/her designee or where no decision has been issued within forty (40) calendar days after it was due, arbitration of a grievance may be initiated in accordance with the following provisions:
  - 1. The Association shall have the exclusive right to initiate arbitration of a grievance. The decision or award of the arbitrator inall be final and binding for the Association, the employee and the Board in accordance with applicable provisions of state law.
  - 2. The Association may initiate arbitration of a grievance only if the resolution of the grievance has been duly authorized by the Association and so certified by the President of the Association or his/her designee.



- 3. The Association shall initiate arbitration by filing a demand for arbitration with the American Arbitration Association and with the Chancellor or his/her designee within forty (40) calendar days of receipt of the decision of the President of the College or where no decision has been issued within forty (40) calendar days after it was due.
- 4. Such arbitration shall be conducted in accordance with the rules and regulations of the American Arbitration Association in effect on the date of said submission, unless otherwise provided herein; provided, however, that the jurisdiction of the arbitrator to inquire into any issue or to render any award is governed solely by the provisions of this Agreement.

### B. Limit of the Arbitrator's Jurisdiction

Subject to the provisions of this Agreement, the arbitrator shall have no authority or jurisdiction to:

- 1. Arbitrate such portion of any grievance which is removed from the jurisdiction of the Arbitrator by the express terms of this Agreement.
- 2. Add to, alter, or amend any terms or conditions of this Agreement.
- 3. To inquire into or arbitrate any issue not presented by the original complaint.

## C. Authority of the Arbitrator

Unless otherwise provided in this Agreement, the arbitrator shall have the authority to make a final and binding award on any dispute concerning the interpretation or application of this Agreement. The arbitrator's authority in matters which are arbitrable is limited to a determination as to whether the provisions set forth in this Agreement were violated; provided that in matters of professional judgment, the arbitrator shall determine whether the application of such to the grievant has been arbitrary, capricious or unreasonable. Beyond such determination, the arbitrator shall be without power, right or authority to make a decision or to substitute his/her judgment for that of the Board or its representatives, except as otherwise provided in this Agreement.

The arbitrator shall have no authority to arbitrate:

- 1. Any incident which occurred or failed to occur prior to the ratification date of this Agreement.
- 2. The failure or refusal by the Board to renew the contract of or to reappoint a unit member in the first (1st) three (3) years of regular appointments.
- 3. Affirmative action/discrimination.
- 4. Basis for retrenchment

### D. Award of Arbitrator

If the arbitrator determines no express provision of this Agreement has been breached in its application to the grievant as claimed, he/she shall dismiss the grievance. If the arbitrator determines that his Agreement has been so breached, he/she may, subject to the provisions of this Article and except as hereinafter provided, provide an appropriate remedy for the breach; provided, however, that in making any monetary award, the arbitrator shall only provide compensation for actual damages directly attributable to such breach, and shall in no event make any award by way of penal damages.

# E. Dismissal and Retrenchment

If a unit member's employment is discontinued as a result of dismissal or retrenchment, and the arbitrator determines based on clear and convincing evidence that the decision of the President of the College or his/her designee was arbitrary, capricious or unreasonable, he/ she shall remand the matter for reassessment by the President of the College or his/her designee of the original decision but shall not have the result or effect of granting any binding award. After remand, the President of the College or his/her designee on remand is arbitrary, capricious or unreasonable, he/she may file an appeal to the original arbitrator (if the original arbitrator is unable to hear the case, the parties shall choose a new arbitrator according to the terms of this Agreement) in accordance with the provisions of this Article. On appeal, if the arbitrator determines based on clear and convincing evidence, that the decision of the President of the College or his/her designee is arbitrary, capricious or unreasonable, he/she shall have the power to make the grievants whole; provided, however, that any monetary award shall be reduced by way of mitigation by an amount equal to the total income received by the grievant during the period for which monetary damages are so awarded. If the arbitrator determines that the Step One or Two decision was not issued within the applicable time limits contained in this Article, he/she may in his/her first decision provide a binding award in accordance with the foregoing procedures.

### F. Promotion

- 1. It is expressly understood that decisions regarding promotion and initial placement are made pursuant to the exercise of professional judgment. Whenever a unit member files a grievance under this Article, the act or decision grieved shall be reviewable by an arbitrator as an exercise of professional judgment and whether the application to the grievant was arbitrary, capricious or unreasonable.
- 2. A unit member shall submit a complaint at Step One to the President of the College or his/her designee within thirty (30) calendar days after he/she knew or should have known of the act or conditions upon which the grievance is based. The President of the College or his/her designee shall meet with the grievant and shall render his/her decision and the reasons therefor in writing to the grievant within thirty (30) calendar days of receipt of the complaint.



If the grievance is not resolved at Step One or if the written decision of the President of the College is not rendered within the time specified, the grievant may appeal to Step Two. The appeal shall be filed within ten (10) calendar days from the date it was due. The Chancellor or his/her designee shall meet with the grievant and shall render a written decision within forty (40) calendar days of receipt of the appeal.

- 3. If the grievance is not resolved at Step Two or if the written decision of the Board is not rendered within the time specified, arbitration may be initiated in accordance with Article X. If arbitration is so initiated and the arbitrator determines based on clear and convincing evidence that the decision of the President of the College or his/ her designee was arbitrary, capricious or unreasonable, he/she shall remand the matter for reassessment by the President of the College or his/her designee of the original decision but shall not have the result or effect of granting any binding award. After remand, the President of the College shall have thirty (30) days to ender a decision. If the grievant believes the decision of the President of the College or his/her designee on ren, and is arbitrary, capricious or unreasonable, he/she may file an appeal to the original arbitrator (if the original arbitrator is unable to hear the case, the parties shall choose a new arbitrator according to the terms of this Agreement) in accordance with the provisions of this Article. On appeal, if the arbitrator determines based on clear and convincing evidence that the decision of the President of the College or his/her designee is arbitrary, capricious or unreasonable, he/she may provide an affirmative and binding award; provided, however, that if so awarded, the grievant shall be promoted to the next appropriate step and provided further that any monetary award shall begin to accrue one (1) year after the date of issuance of the first (1st) arbitration award. If the arbitrator determines that the Step One or Step Two decision was not issued within the applicable time limits contained in this Article, he/she may in his/her first (1st) decision provide a binding award in accordance with the foregoing conditions.
- G. The granting or failu e to grant tenure shall be arbitrable but any award is not binding.
- H. Notwithstanding any rule of the American Arbitration Association to the contrary, in making his/her decision, the arbitrator shall apply the express provisions of this Agreement and shall not alter, amend, extend or revise any term or condition hereof.
- I. All fees and expenses of the arbitrator shall be divided equally between the parties. Each party shall bear the cost of preparing and presenting its own case.

### ARTICLE X - APPENDIX A

### Release of Confidential Materials for Grievance Administration

In accordance with the responsibility of the Board of Regents of Higher Education (Board) to provide to the Massachusetts Community College Council/Massachusetts Teachers Association (MCCC/MTA) such information as may be necessary to the discharge of its duties as the exclusive bargaining agent under Chapter 150E and to maintain the confidentiality of personal data in accordance with state and federal law, the Board and the Association agree:

- 1. Whenever the MCCC/MTA Grievance Coordinator or his/her designee needs personnel file information from a College in order to intelligently evaluate and prepare a grievance involving evaluations, promotions or performance-based awards, the MCCC/MTA Grievance Coordinator or his/her designee shall make a written request for such information to the President of the College from where the grievance originated.
- 2. Within ten (10) calendar days of receipt of such information request, the President or his/her designee shall give written notice to all bargaining unit members involved that materials contained in their personnel files and/or evaluative materials are being released to the Union for purposes of grievance administration.
- 3. The President of the College or his/her designee shall mail the requested information within sixty (60) calendar days of receipt of the request for information to the MCCC Grievance Coordinator.
- 4. The time limits prescribed herein may be extended in any specific instance by mutual agreement of the parties or by oral agreement confirmed in writing.
- 5. The College may charge the MCCC/MTA a reasonable sum to cover the cost of reproduction of the materials requested.
- 6. The Grievance Coordinator or his/her designee shall maintain the confidentiality of file materials in accordance with state and federal law.

	FORM SUPPLEMENT		
X - G1	Step One Complaint	Page No.	ii
X – G2	Association Evidence	Page No.	iii
X – G3	Management Evidence	Page No.	iv
X - G4	Step One Decision	Page No.	v
X – G5	Step One Appeal	Page No.	vi
X - G6	Step Two - Notice to Meet with the Chancellor or His/Her Designee	Page No.	vii
X – G7	Step Two Decision	Page No.	viii
X - G8	Arbitration Approval Request	Page No.	ix

24



# ARTICLE XI - APPOINTMENT AND REAPPOINTMENT - TENURE

### 11.01 Appointments

### A. General

Except as expressly provided for by Board policy, the authority to offer and make effective appointments and reappointments rests with the Board and no employee or agent of the Board or the Colleges may expressly or by implication make effective any appointment or reappointment.

- 1. Each unit member's reappointment to a bargaining unit position shall be in accordance with the provisions of this Article.
- All appointments and reappointments to unit positions at a College shall be made by the Board upon recommendation of the President of the College. Notification of such appointment shall be in writing as provided in Article XII.
- 3. Unit members who are temporarily shifted to other than state appropriated funds shall, during and upon their return to state appropriated funds, retain all rights and benefits they would have had if they had not been transferred.

### **B.** Types of Appointment

- 1. A regular appointment shall be one which creates an interest in employment for a specified term of one (1) year or less. All employees holding regular appointments shall be subject to non-reappointment without cause.
- 2. A tenured appointment may be offered only to those faculty members in the ranks of Assistant Professor, Associate Professor and Professor and to professional staff members in the classification of Professional Staff II. III and IV.

### 11.02 Reappointment and Non-Reappointment of Regular Appointments

- A. The non-reappointment of a faculty or professional staff member shall be in accordance with the following procedures:
  - 1. Termination at the end of the first (1st) through third (3rd) years of service; notice provided by February 15 of that year of service:
  - 2. Termination at the end of the fourth (4th) year of service or later; notice provided by October 15 of the year of termination.
- B. Non-reappointment of a unit member on a regular appointment in the first (1st) three (3) years shall be without cause, except for written notice requirements required in Section 11.02. Reasons may be provided at the discretion of the President of the College; neither the reasons nor the decision shall be subject to the grievance procedure. For the fourth (4th) year of regular appointment, the notice of non-reappointment of non-tenured unit members shall be subject to the exercise of professional judgment. Such notice shall contain a statement of reasons in writing. Termination in the fourth (4th) year or later of unit members whose source remuneration is paid from non-state appropriated funds shall be accompanied by a statement of reasons; provided, however, that no statement shall be required if non-reappointment is due to insufficient funds or to the terms and conditions of the non-state appropriated funding source and provided further that said non-reappointment shall be grievable to step two and shall be subject but not arbitrable to the exercise of professional judgment and whether application to the grievant was arbitrary, capricious or unreasonable.
- C. Termination of a unit member in his/her fifth (5th) regular appointment or later shall be for just cause.
- D. Notice shall be in writing given by the President of the College or his/her designee. Any unit member holding a regular appointment who does not receive such notice shall be entitled to inquire of the President of the College as to the failure to give notice. The President of the College or his/her designee shall respond to the inquiry within fourteen (14) calendar days.
- E. Failure to give notice of reappointment shall not prohibit reappointment of the unit member. Failure to give proper notice shall constitute reappointment of the unit member for one (1) academic or calendar year, whichever is applicable, but shall not thereby entitle the unit member concerned to academic tenure, to any further appointment with or without academic tenure, or to further notice of non-reappointment.
- F. Each unit member holding a regular appointment shall notify the President of the College in writing of his/her intent to accept or reject a reappointment within thirty (30) days after receipt of notice of reappointment. Failure by the unit member to notify the President of the College of acceptance shall constitute a rejection of reappointment.

### 11.03 Reappointment of Unit Member

- A. Tenure may be granted by the Board on recommendat on of the President of the College and shall relate only to the specific College and not the entire Community College System.
- B. Eligibility Tenure Contracts
  - 1. The unit member must have served at least six (6) full years as a unit member, at least three (3) of which have been in his/her current job function.
  - 2. The unit member must have received other than unsatisfactory on his/her most recent summary evaluation.



### C. Procedures

- 1. The shall be established by November 21 of each year a Unit Personnel Practices Committee elected by members of the unit. The size and composition of this Committee shall be determined by the President of the College or his/her designee and shall reflect, whenever possible, the ratio of teaching faculty to professional staff in the unit; provided, however, that a unit member shall not serve on the Committee during the year he/she is a candidate for tenure. The Committee shall elect a chairperson.
- 2. The appropriate Dean(s) s'all notify all eligible unit members by October 1 of their sixth (6th) or later regular appointment that they may be considered for tenure during the academic year if otherwise eligible; provided that copies shall be forwarded to the immediate supervisor and to the Unit Personnel Practices Committee.
- 3. The immediate supervisor and the Unit Personnel Practices Committee shall review all relevant material within the individual's official personnel file and shall forward recommendations for either tenure or a only (1) year terminal appointment to the appropriate Dean (s) by March 15th.
- 4. The appropriate Dean(s) shall review the recommendations of the immediate supervisor and the Unit Personnel Practices Committee; and whenever practicable consult with the immediate supervisor and the Chair of said Committee; and thereafter shall forward his/her recommendations to the President of the College by April 15.
- 5. The President of the College shall review the recommendations of the appropriate Dean(s) and shall forward his/her recommendation to the Board by May 1. The unit member shall be notified of the President's recommendation within twenty (20) calendar days thereafter.
- 6. A recommendation for a one (1) year terminal contract shall be accompanied by a statement of reasons.

### 11.04 Protection of Bargaining Unit Upon Appointment of Returning Administrators

Any unit member who assumes administrative duties outside the bargaining unit and subsequently returns to faculty or professional staff unit member status shall, at the option of the President of the College or his/her designee either:

(1) retain all previously accrued unit seniority and be placed upon the salary schedule at the step and rank he/she held prior to assuring administrative appointment, as adjusted by the provisions of any collective bargaining agreement between the Board and the Association; or (2) shall relinquish all prior accrued unit seniority and be placed upon the salary schedule at a step and rank to be determined by the President of the College or his/her designee; provided, however, that a unit member who accepts an acting administrative appointment shall upon return to the unit retain all prior accrued unit seniority and be placed upon the salary schedule at the rank and step he/she held prior to assuming administrative appointment as adjusted by the provisions of any collective bargaining agreement between the Board and the Association. If an acting administrative appointment is converted to a regular administrative appointment, no rights of return provided under (2) above shall be accorded to the acting appointee unless the regular position has been posted and awarded to the acting appointee. It is agreed that the return of an administrator to the unit under either of the options stated above will have no adverse effect on present unit members.

11.05 Subsequent to July 31, 1981, the faculty lines of Professor, Associate Professor, and Professional Staff Classifications III and IV shall be occupied only by full-time faculty and/or professional staff members; provided, however, that the foregoing shall not require displacement from an upper faculty position of any incumbent administrator who occupies such position. or to August 1, 1981. Nothing contained in this Article shall be construed to impair the rights of the parties with regard to those challenges to returning administrators pending on the date of ratification and approval of this Agreement.

# 11.06 Multiple-Year Contracts

Because the parties agreed to delete multiple-year contracts from the 1983-86 Collective Bargaining Agreement, the parties further agree as follows:

- A. A multiple-year contract shall remain in full force and effect until its expiration or until the unit member becomes eligible for tenure.
- B. A unit member who is otherwise eligible may apply for tenure during the term of his/her multiple-year contract.
- C. Multiple-year contracts shall be transferable from one College to another only at the option of the President of the receiving college or his/her designee and if approved by the Roard.

# ARTICLE XII - WORKLOAD, WORK ASSIGNMENT AND WORKING CONDITIONS

### 12 01 General Provision

Unit members shall be given a contract stating the subjects or functions to which the unit member is to be assigned. There shall be included a job description which shall define the duties to be performed.

### 12.02 Work Assignment

A. The President of the College or his/her designee shall consider as advisory written notice from each faculty member and, or department chair/curriculum coordinator is to the preferred courses to be taught. Such written notice must be received by the President of the College or his/her designee at least one (1) month prior to the date the College must notify each faculty member as contained herein. The President of the College or his/her designee



- shall notify in writing each faculty member of his/her tentative courses and number of sections to be taught by the end of the April payroll period for the Fall semester and by the end of the November payroll period for the Spring semester.
- B. The President of the College or his/her designee shall consider as advisory, written notices from each faculty member and/cr department chair/curriculum coordinator as to the preferred class schedule. Such written notice must be received at least one (1) month prior to the date the College must notify each faculty member as contained herein. The President of the College or his/her designee shall notify each faculty member in writing of his/her tentative class schedule by the end of the May and December payroll period for the succeeding Fall and Spring terms, respectively. The class schedule indicates the days and times each week that the faculty member has contact responsibility with students and does not refer to the academic calendar defined elsewhere.
- C. Whenever changes in the tentative class schedule or subject matter are necessary, the President of the College or his/her designee shall notify the affected faculty member(s) in writing at his/her official residence or a temporary address supplied by the faculty member on leave. The change in the tentative class schedule or subject matter preparation shall be consistent with Section 12.01.

# 12.03 Workload of Faculty

- A. Faculty workload shall consist of:
  - 1. Instructional workload, which includes:
    - a. teaching in both traditional and non-traditional learning modes;
    - b. instructional preparation;
    - c. assessment of student performance.
  - 2. Non-instructional workload shall consist of:
    - a. student assistance/advisement:
    - b. college service, which includes:
      - 1. serving as advisor to student activities;
      - 2. serving on governance, ad hoc, college standing committees, system-wide task forces or committees, or labor-management committees:
      - 3. preparing grant proposals;
      - 4. participating in college, division, department or other related college meetings and/or activities;
      - 5. participation in the improvement and development of academic programs and resources, including recruitment.
    - c. college recognized community service, provided that such service is not compensated by an outside funding source;
    - d. professional development activities, which include:
      - 1. related graduate study;
      - 2. related in-service training:
      - 3. research and other College recognized contributions to a faculty member's area of competence;
      - 4. participation in College recognized professional associations, including teachers' organizations.

### B. Instructional Workload

- 1. The standard faculty instructional workload shall be twenty-four (24) units of instruction per year with two (2) preparations per semester. This standard is not intended to be a fixed, absolute maximum of 24 units of instruction per year, but is intended to achieve greater uniformity in the system with 24 units being the target norm. Any significant variation above the norm could subvert the purpose of the standard; provided further that faculty with more than 24 instructional hours per semester will continue to receive adjustments in their non-instructional workload as per the appended provision of Article XII of the 1977-80 Collective Bargaining Agreement. A unit of instruction shall mean per week:
  - a. one (1) fifty (50) minute lecture or seminar,
  - b. one and one half (1½) hours of laboratory instruction;
  - c. one and one-half (1½) hours of clinical instruction; provided that the effect of the implementation of this ratio shall not be considered as a violation of the standard of 24 units;
  - d. two (2) hours of individualized instruction or other non-traditional modes of instruction;

27

e. where team teaching involves multiple faculty, each faculty member shall be allotted instructional workload credit in accordance with R.1. a, b, c and d above, except that it shall be based on one-half (½) the course units as described above, regardless of the number of faculty involved in teaching, during the initial implementation of a team structure for a course. 'a sis shall also apply to an individual teacher who joins an existing tesm for the first time. Thereafter, credit shall be based in direct proportion to the percentage of total course time that he/she teaches.



The President of the College or his/her designee shall adjust the ratios in b, c, d and e if in his/her judgment, after consultation with the unit member the difficulty in instruction, the difficulty and duration of preparation, or the provision of assistance in instruction or such other related instructional, clinical or programmatic factors merit such adjustment. The President of the College or his/her designee may upon mutual agreement assign a faculty member non-instructional activities such as course, program or curriculum development, professional development activities or administrative duties; provided that he/she is qualified to perform such activities in lieu of a proportional number of instructional units.

2. Faculty members shall not be expected to generate more than 420 student credit hours per semester within the applicable workload requirement or more than 300 student credit hours per semester for instruction of English Composition, English as a Second Language, introductory foreign language courses, and remedial and/or developmental courses; to be determined by the number of students enrolled at the end of add/drop period; provided that if the total number of student credit hours generated exceeds these limits, there shall be a proportional adjustment in the succeeding semesters, consistent with the provisions of 12.03 B.

It is expressly agreed that during the term of this Agreement a faculty member's workload shall not be increased or decreased beyond his/her 1986 workload; provided, however, it is agreed by the parties that if a faculty member agrees to an increase in the number of course preparations beyond two (2), the President of the College or his/her designee and the faculty member shall mutually agree as to the corresponding decrease in advisees, and/or office hours and/or college service. New faculty hired shall be assigned an instructional workload consistent with the existing workload assignments of other faculty members within their respective academic department, work area or division. It is agreed by the parties that whenever a dispute in the instructional and non-instructional workload exists, the standard to be used shall be the appended provisions of the workload article of the 1977-80 Collective Bargaining Agreement.

### C. Non-Instructional Workload

### 1. Student Assistance/Advancement

A faculty member shall be responsible for advising 20-25 students assigned to him/her by the President of the College or his/her designee.

It is the agreement of the parties that it is not the intent of this Article to allow a unit member to schedule in advance student advisement time during his/her office hour schedule. It is agreed, however, that if a student advisee appears of his/her own volition during a unit member's office hours, a unit member may utilize that time for student advisement responsibilities.

It is further understood that the administration has the obligation to assist the unit member in contacting student advisees and that the administration cannot determine the scheduling of this advisement time.

Each faculty member shall hold a group meeting with his/her advisees at the beginning of the academic year and shall be available to meet with each advisee at least once each semester and at such other reasonable times as requested by the advisee.

If in the judgment of the President of the College or his/her designee after consultation with the faculty member the number of students should be more than or fewer than 20-25, the faculty member's non-instructional workload may be adjusted.

### 2. Office Hours

During the academic year, faculty members shall maintain at least five (5) posted office hours per week on at least four (4) days. Office hours shall be scheduled in accordance with the following procedure:

- a. A faculty member shall submit a preferred office hours' schedule to the President of the College or his/her designee prior to the commencement of the academic year.
- b. The President of the College or his/her designee shall either approve or disapprove the schedule.
- c. In the event the President of the College or his/her designee shall disapprove the schedule, the faculty member shall be given an opportunity to review and discuss their respective positions with the President of the College or his/her designee.
- d. In the event the faculty member believes he/she is aggrieved by the decision of the President of the College or his/her designee, he/she may appeal to the Workload Grievance Panel in accordance with the procedures outlined in Article 12.06.

Such schedule shall be posted by faculty members by the end of the first (1st) week of classes.

### 3. College Service

In addition to posted office hours, a faculty member shall be on campus and available on any day during the scademic year at such times as required by the President of the College or his/her designee to participate in scheduled activities con stent with 12.03 A2 b.

### 4. Community Service and Professional Development

A faculty member shall be responsible for such community service and professional development activities as mutually agreed by the President of the College or his/her designee and the faculty member.

### D. Professional Days

The President of the College or his/her designee may assign to any faculty member up to sever. (7) days during the



academic year for scheduled orientation and registration programs, commencement and convocation activities and such other College-sponsored activities he/she deems appropriate. Professional days shall be assigned no earlier than three (3) days prior to the first (1st) day of classes in the Fall semester and no earlier than four (4) days prior to the first (1st) day of classes in the Spring semester. Such duties and responsibilities may be assigned to individual faculty members in blocks of one-half (½) day or more; provided that a faculty member assigned more than one-half (½) day shall be credited with one (1) full day; and provided further that a faculty member shall not be required to discharge these responsibilities over more than fourteen (14) days during the academic year.

### E. Academic Year

- 1. The academic calendar shall be a period of time encompassing two (2) semesters, beginning no earlier than September 1 and ending no later than May 31, exclusive of Commencement. Classes shall begin no earlier than the day after Labor Day and end no later than December 24 for the Fall semester; provided, however, that the President of the College and the President of the local Association Chapter may by mutual agreement set other dates for the academic calendar, including beginning classes prior to the day after Labor Day. When first (1st) semester classes begin after September 5, the President of the College may, after consultation with the local MACER, increase the class time from fifty (50) minutes to fifty-five (55) minutes for that semester consistent with the established past practice. There shall be at least a four (4) week recess between semesters.
- 2. In those instructional programs that are required by outside accrediting agencies to run beyond the academic year, it may be necessary for faculty to perform duties beyond the academic year. When such work is assigned by the President of the College or his/her designee, the faculty member shall receive a salary adjustment in direct proportion to the duties assigned, in accordance with the provisions of Article XXI. All such faculty will be placed on nine month contracts with all rights and benefits under this Agreement.
- 3. No faculty member who is not at the time of the execution of this Agreement required to work beyond the academic year may be required to do so without his/her consent. Any faculty member may accept such additional responsibilities upon the specific request of the President of the College or his/her designee and will receive compensation in accordance with the provisions of Article 21.05.
- 4. Nothing in this Agreement shall preclude the President of the College from requiring such additional responsibilities of new hires in such programs as are referenced in paragraph 2 above.
- 5. Notwithstanding any provision in this Agreement to the contrary, no College shall be required to establish or maintain any instructional programs offered outside the academic year such that they are subject to Article 12.03 E.2 above or Article XXI nor shall any College be precluded from establishing or maintaining any such programs which similarly are not subject to the terms of this Agreement.
- 6. Section 12.03 E.2 programs that currently extend beyond the academic year include:

ESL Clinical Assistant Program
ESL Electronics Program
Licensed Practical Nurse
LPN Update
Medical Laboratory Technician
Medical Radiography
Nuclear Medicine Technology
Radiologic Technology
Radiologic Technology
Respiratory Therapy

7. Prior to the implementation of programs not listed in paragraph 6 above which would require faculty member (s) to work beyond the academic year, the President of the College shall notify the President of the Association and the President of the Chapter and shall meet and confer with the President of the Association or his/her designee regarding said programs. Accurate information, statistics or financial data related to such programs shall be made available by the President of the College or his/her designee for inspection and/or copying upon request of the President of the Association or his/her designee; provided, however, that this shall not require the College to compile such information, statistics or financial data in the form requested unless already compiled in that form. Nothing in this provision shall be regarded as an abridgement of the rights and responsibilities of the Board provided in Article IV.

### F. Outside Employment

Full-time employment by the Board shall be considered the primary employment of each faculty member. Outside employment is work for which compensation is received and which is not the normal duties and responsibilities assigned to a faculty member as an employee of the Board. Outside employment which requires a faculty member's absence from the College during his/her normally scheduled working hours is presumed to interfere with the performance of the duties and responsibilities of that faculty member and is hereby prohibited.

### 12.04 Workload of Professional Staff

- A. Professional staff workload shall unsist of:
  - 1. professional duties as contained in a letter of appointment and position description;
  - 2. student advisement, if assigned;



- 3. college service, which includes:
  - a. service as advisor for college-approved student activities;
  - b. service on governance, ad hoc, college standing committees, system-wide task forces or committees; or labor-management committees;
  - c. preparing grant proposals:
  - d. participating in college, division, department or other related college meetings and/or activities as the President of the College or his/her designee may deem appropriate;
  - e. participation in the improvement and development of academic programs and resources, including recruitment.
- 4. College recognized community service, provided that such service is not compensated by an outside funding source, and professional development; provided that such community service and professional development activities are mutually agreed to by the President of the College or his/her designee and the professional staff member.
- 5. Assigned instructional responsibilities and related preparation; provided that customarily professional staff members shall not be assigned traditional academic discipline responsibilities; and provided further that no professional staff member shall be required to teach subject matter which is beyond the scope of his/her academic and/or professional competency and experience as determined by the President of the College or his/her designee.

### B. Work Year

All professional staff employees shall work a twelve (12) month year.

### C. Work Schedule

The President of the College or his/her designee shall consider as advisory written notice from the professional staff member as to his/her preferred work assignment if received on or before June 1st prior to the fiscal year(s) covered by this Agreement. The President of the College or his/her designee shall notify the professional staff member in writing of his/her work assignment no later than July 1. Such work assignment shall be consistent with the needs of the Coalege.

- 1. The workload of each professional staff member shall consist of such duties and responsibilities in accordance with his/her respective work schedules as may be assigned by the President of the College or his/her designee.
- 2. During the term of this Agreement, the normal work week for professional staff members shall be 27% hours per week.
- 3. During the term of this Agreement, the normal work day shall not exceed 7½ hours, excluding lunch.
- 4. Whenever a professional staff member is required to work more than 37½ hours in a particular week, he/she shall receive compensatory time on an hour-to-hour basis within the next sixty (60) working days, subject to mutual agreement between the professional staff member and the President of the College or his/her designee.
- 5. The customary work week for professional staff members shall be Monday through Friday, within the hours of 8:00 a.m. to 5:00 p.m. but in no case shall a professional staff member be required to work more than five (5) days in any seven (7) consecutive day period; provided that a professional staff member who is assigned to teach may request a revised work schedule.
- 6. Upon request of a professional staff member, the President of the College or his/her designee may assign up to three (3) days during the fiscal year in half-day segments or greater for participation in off-campus activities outside those assigned under the provisions of 12.04; one of these days shall be granted for the day following the Thanksgiving holiday. Such requests shall not be unreasonably denied.
- 7. All time spent in student advisement (if assigned), in attendance at conferences, meetings and student activities when such attendance is required during the work day shall be counted in the professional staff member's weekly and daily hours.
- 8. Upon request of a professional staff member the President of the College or his/her designee may, where practicable, grant a flexible schedule; provided, however, that the President's decision shall be final and non-grievable.
- 9. Upon request of a professional staffmember, the President of the College or his/her designee may, where practicable, grant a ten (10) month year option at 10/12 of pay and benefits.
  - a. The unit member shall apply to his/her immediate supervisor.
  - b. The decision of the immediate supervisor may be appealed to the President of the College who shall cause to be convened a Committee composed of two appointees of the President of the College and one appointee of the President of the Association.
  - c. The recommendation of the Committee may be appealed by the applicant to the President of the College whose decision shall be non-grievable.



# 12.05 Worklead Reduction

A department chair/curriculum coordinator (program and work area/discipline) shall be granted a reduction in instructional workload; provided that the reduction shall not exceed eighty (80) percent nor be less than twenty (20) percent. The amount of workload reduction shall be dependent upon the duties performed by the department chair/curriculum coordinator and shall be solely determined by the President of the College or his/her designee. Such reduction shall not be made in an arbitrary, capricious or unreasonable manner.

# ARTICLE XII - APPENDIX A

# 12.05 Workload for Faculty Members

The workload for faculty members shall include office hours for students, instructional workload and non-instructional workload as defined below. The faculty customary workweek shall be Monday through Friday, but in no case shall a faculty member be required to work more than five days in any seven consecutive day period.

In exceptional cases, if it is necessary to schedule faculty for days other than Monday through Friday, the President or his/her designee shall meet with the affected faculty member (s) to discuss such change in schedule.

In no case shall any change be implemented without at least one semester prior notification to the affected faculty member (s). The affected faculty .nember (s) shall have access to the grievance procedure as provided for in

# A. Instructional Workload

The instructional workload shall contain all faculty activity directly related to the preparation and/or conduction of instruction.

- 1. <u>Preparation</u> includes content and pedagogical research, the creation of instruction materials, development of student evaluation instruments and procedures, the evaluation of student performance and any activity related to the instructional process. It is understood that faculty members at their option need to utilize off-campus resources in fulfilling their preparation time.
- 2. Contact is the actual time the instructor spends with students in an instructional method.
- 3. The instructional workload shall be a minimum of 24 hours per week and a maximum of 33 hours per week.
- 4. In determining the instructional workload the actual contact hour shall be added to the preparation time allotted on the following basis:

# **B.** Allocation of Preparation Time

Preparation time shall be allocated on the following basis:

# 1. Non-Laboratory-like Courses

For the first section or part thereof of such a course offered each week, the faculty member shall be allotted one and one-third (1%) hours of preparation time for each credit hour or its proration which that course contains.

# 2. Laboratory-like Courses

For the first section or part thereof of a course offered each week:

The faculty member shall be allotted one and one-third (1½) hours of preparation for each credit hour of didactic instruction. The faculty member shall be allotted two (2) hours of preparation for each credit hour of equivalent laboratory instruction requiring two or more contact hours.

# 3. Seminar-like Courses

For the first section or part thereof of seminar-like instruction whether within a laboratory or non-laboratory type course where the instructor is in direct interaction with students, such as small group discussions, clinical conferences, the faculty member shall be allotted one and one-third (1½) hours of preparation time for each credit hour or its proration which that cours? contains.

# 4. Addita nal Sections of Original Preparation

For the second or subsequent section or part thereof of a course offered each week, the faculty member shall be allotted one-half (½) the preparation time credited in either one or two or three above.

- 5. Where team teaching involves multiple faculty, each faculty member shall be allotted preparation time in accordance with B.1, 2, 3 and 4 above except that it shall be based on one-half (½) the credit hour or proration which that course contains regardless of the number of faculty involved in teaching.
- 7. One hour of preparation will be allotted per each three (3) credit hour course or its prorated equivalent where the faculty member is using material of a mediated or programmed nature wherein the faculty member has neither primary responsibility for adapting or preparing materials nor is primarily responsible for the evaluation of student progress.
- 8. The President or his/her designee shall [compute] at the end of the "add" period of each semester the actual instructional workload for each full-time faculty member according to the above formula. The President or his/her designee will, on the basis of the faculty member's instructional workload:



31

WAY THE ARMEN STATES

- a. Reduce the office hour and/or non-instructional load in writing proportionately for any faculty member whose instructional workload exceeds the minimum of twenty-four (24) hours;
- c. Determine in writing after discussing alternatives with the affected faculty member whose load is below the twenty-four (24) hour minimum:
  - 1. whether additional course section (s) shall be taught by the faculty member; and/or
  - 2. whether the faculty member shall provide activities related to course, program or curriculum development;

and/or

- 3. whe' er the faculty member shall provide activities such as independent study, contract learning, learning resources development, assessment or prior learning or similar instructional activities;
- 4. whether the faculty member shall provide activities such as additional office hours, advise more stucents, engage in additional college/community service, or professional development activities; and/or
- 5. whether the faculty member shall provide activities of an administrative nature for which he/she is deemed qualified;
- 6. whether the faculty member will provide additional tutorial laboratory instruction, where a faculty member is teaching two or more sections of courses designed to develop basic skills, whether remedial or supplemental, credit or non-credit or their equivalent in time and/or content as determined by the Presi dent or his/her designee.
- f. The President or his/her designee may project the instructional workload requirement over the full academic year rather than on a semester basis based upon the unique college or program needs after discussion with the affected faculty member. The instructional workload shall not exceed thirty-three (33) hours in any one semester, without the agreement of the faculty member. Provided, however, the instructional workload shall not exceed sixty-six (66) hours for way academic year.

### B. Office Hours

Each faculty member shall provide such hours on campus (off campus at an instructional site, if approved) to assist students in hir/her courses.

### C. Non-Listructional Workload

Faculty shall provide eight (8 urs per week or equivalent on a semester basis or its annual equivalent in student advisement and/or college and/or community service as determined by the President or his/her designee consistent with Article XIII (Evaluation).

c. Student advisement shall equate to the following hours per week of the non-instructional requirement:

Assigned Students	Equated Hours Per Wee
Less than 8	1
8-13	2
14-19	3
20 - 25	4
26 - 31	5
32 - 37	6
38-43	7

#### D. Load Reduction

Where faculty members are given load reduction within the instructional workload for any activity other than teaching, the number of hours required for the activity will be equal to twice the credit hour reduction with the proportional reduction in office hours of one hour per three credit hour equivalent load reduction.

	FORM SUPPLEMENT		
XII - I	Workload Computation Form	Page No.	x
XII-2	Workload Materials Acknowledgement	Page No.	xi



# ARTICLE XIII - EVALUATION

# 13.01 Evaluation Objectives

The evaluation of unit members is directed to the following objectives:

- 1. Assessment of the professional performance of the unit member.
- 2. Improvement of performance and quality of instruction.
- 3. To provide a basis upon which decisions shall be made concerning the reappointment, promotion, performancebased salary adjustments, tenure, sabbatical and professional leaves, and termination, dismissal and discipline of

# 13.02 Evaluation of Faculty Members

# A. Faculty Evaluation Criteria

Evaluation of faculty members shall be uniformly applied and based upon total performance with primary emphasis on teaching effectiveness, including consideration of the following criteria:

- 1. Development and improvement of instructional methodology;
- 2. Establishment of course objectives, course content and instructional activities;
- 3. Establishment of appropriate and fair procedures and instruments for student evaluation;
- 4. Student advising competency and accessibility; including the observance of established office hours;
- 5. College service, including:
  - a. serving as advisor to student activities: and/or
  - b. serving on governance, ad boc, college standing committees, system-wide task forces or committees, and labor/management committees; and/or
  - c. preparing grant proposals; and/or
  - d. participating in college, division, department or other related College meetings and/or activities;
  - e. participation in the improvement and development of academic programs and resources, inc., ling

### B. Procedure

The procedure for evaluating faculty members shall consist of five (5) processes: (1) student evaluation; (2) course materials evaluation; (3) classroom observation evaluation; (4) student advisement and college service evaluation; and (5) summary evaluation. The foregoing procedure shall be implemented on an annual basis except

### 1. Student Evaluation

- a. The President of the College or his/her designee shall be responsible for administering the student evaluation
- b. Student evaluation packets for each class containing instruments and instructions shall be distributed to each faculty member by November 15 except for tenured faculty for whom student evaluations shall be performed every other year in accordance with the provisions of 13.04.
- c. It is expressly agreed that the faculty member being evaluated shall not be present in the classroom when the student evaluation is being administered and that all instruction to students with regard to such student evaluation shall be included in writing on the instrument, provided further that the designated unit or nonunit professional shall return the student evaluation directly to the President of the College or his/her designee. The administering of the student realization shall be the responsibility of the President of the Coalege or his/her designee who shall determine who among unit or non-unit professionals all administer such student evaluation. Student evaluations shall be valid only if signed by the student; provided, however, that faculty members shall not be entitled to the identity of the student responding unless such student evaluation is used as a basis for dismissal or other disciplinary action and such will be communicated to
- d. The data from the student evaluation shall be tabulated and copies sent to the President of the College or his/ her designer. The raw data shall be retained by the College for a period of one (1) year during which time the faculty member shall have access thereto upon written request.
- e. The President of the College or his/her designee shall review the tabulated data and shall prepare a student evaluation summary and forward a copy to the faculty member by December 15.
- f. The faculty member shall have seven (7) working days in which to respond to such summary evaluation.

# 2. Course Materials

- a. The faculty member shall forward to the President of the College or his/her designee no later than the end of the second (2nd) week of classes in each semester the following materials for each course:
  - (1) Course outline



- (2) Instructional objectives and requirements students must meet to complete the course; and
- (3) Description of procedures to be used in evaluating students.

The confidentiality of these materials shall be maintained.

- b. Upon request of the affected unit member the President of the College or his/her designee shall consider advisory input from the appropriate department chair/curriculum coordinator relating to the unit member's course materials.
- c. The President of the College or his/her designee shall review the course materials and, except for tenured faculty during a non-evaluation year, evaluate them. In reviewing said materials the President of the College or his/her designee may seek additional information from other sources he/she may deem appropriate, including but not limited to department chair.
- d. The President of the College or his/her designee shall return all course materials to the faculty member and, except for tenured faculty during a non-evaluation year, a copy of the evaluation of the course materials by the end of the fifth (5th) week of classes in each semester and, if requested, shall meet and confer with the faculty member to discuss the reasons therefor.
- e. The faculty member shall have seven (7) working days in which to respond to the evaluation.

### 3. Classroom Observation

- a. The President of the College or his/her designee shall conduct at least one ) classroom observation per academic year, except for tenured faculty during a non-evaluation year.
- b. Each faculty member shall be advised of a two (2) week period during which the President of the College or his/her designee shall conduct classroom observation; provided that each faculty member shall be given at least twenty-four (24) hours' notice of the date of classroom observation; provided, however, that a one (1) time postponement may be requested by a faculty member, if requested twenty-four (24) hours in advance. Each faculty member may submit to the President of the College or his/her designee supplementary course materials regarding the planned classroom activities during said two (2) week period.
- c. Pre-and Post-Observation Conferences: For each classroom observation conducted by the President of the College or his/her designee there shall upon request of the affected unit meml occur a pre-observation and post-observation conference between the unit member and the President of the College or his/her designee. The post-observation conference shall occur not later that wo (2) weeks after the classroom observation unless otherwise mutually agreed.
- d. The President of the College or his/her designee shall within fourteen (14) days of observation prepare an evaluation of the classroom observation in accordance with the criteria set forth in 13.02A and shall forward a copy to the faculty member.
- e. The faculty member shall have seven (7) working days in which to respond to the evaluation.

### 4. Student Advisement and College Service

- a. The President of the College or his/har designee shall evaluate, except for tenured faculty during a non-evaluation year, the performance of the unit member's assigned duties and responsibilities consistent with the criteria established in this Article. The procedure for evaluating performance shall be as hereinafter described.
- b. The faculty member shall submit documentation relating to performance of his/her duties and responsibilities, including but not limited to a log of student advisement and college service activities to the President of the College or his/her designee by the last day of classes in each semester. A tenured unit member shall be required to submit only a log of student advisement and a list of college service activities performed during a non-evaluation year.
- c. The President of the College or his/her designee shall review the document (s) and may seek additional information from other sources he/she deems appropriate. The President of the College or his her designee shall prepare an evaluation of student advisement and college service performed since the last such evaluation and shall forward a copy of the evaluation and, if requested, shall meet and confer with the faculty member to discuss the reasons there or.
- d. The faculty  $m_{cl}$  is shall have seven (7) working days in which to respond to the evaluation.

### 5. Summary Evaluation

The President of the College or his/her designee shall after receipt of all the foregoing evaluation components and prior to February 1 develop a summary evaluation of each faculty member, except for tenured faculty during a non-evaluation year, and shall consider only the foregoing evaluation components. The results of the components shall be applied in a uniform manner and shall be assigned the following weights:

Student Evaluation	30%
Course Materials Evaluation	25%
Classroom Observation Evaluation	25%
Student Advisement Evaluation	10%
College Service Evaluation	10%



A copy of the summary evaluation shall be forwarded to the faculty member by February 1, upon receipt of which the faculty member shall he seven (7) working days to respond to the summary evaluation.

# 13.03 Evaluation of Professional Straff Members

# A. Professional Staff Evaluation Criteria

Evaluation of professional staff shall be uniformly applied and based on total job performance, including the following criteria:

# 1. Professional performance including:

- a. conformance with assigned workload as established by the appropriate administrator in accordance with Article XII; and
- b. the rendering of effective assistance to students, faculty and staff of the College, individually and/or collectively in accordance with Article XII.
- 2. Student advising competency and accessibility, if appropriate.
- 3. College service, including:
  - a. serving as advisor to student activities; and/or
  - b. serving on governance, ad hoc, college standing committees, system-wide task forces or committees, or labor/management committees; and/or
  - c. preparing grant proposals; and/or
  - d. participating in college, division, department or other related college meetings and/or activities as the President of the College or his/her designee may deem appropriate; and/or
  - e. participation in the improvement and development of academic programs and resources, including recruitment.
- 4. Assigned instructional responsibilities and related preparation in accordance with Article XII

### B. Procedure

The procedure for evaluating professional staff members shall consist of three (3) processes: (1) work performance evaluation, (2) student advisement and college service, and (3) summary evaluation. The foregoing procedure shall be implemented on an annual basis except as provided in 13.04.

# 1. Evaluation of Work Performance

# a. P Evaluation Conference

By ally 1 or the beginning of a professional staff member's appointment, the President of the College or his/her designee shall request a pre-evaluation conference with the affected professional staff member for the purpose of developing his/her position description, which shall contain a list of duties and responsibilities and may include mutually agreeable work objectives, if appropriate. The professional staff member shall not be required to participate in this conference. The President of the College or his/her designee shall upon request meet with a professional staff member during the evaluation year to discuss his/her progress.

b. The President of the College or his/her designee shall complete the work performance evaluation by January 15 of the first appointment and by June 1 of the first and subsequent appointments, except for tenured professional staff members during a non-evaluation year. The professional staff member shall receive a copy of the evaluation and shall have seven (7) working days in which to respond to the evaluation.

# c. Post-Evaluation Conference

The President of the College or his/her designee shall upon request meet and confer with the professional staff member to discuss the remons for his/her evaluation within fourteen (14) days following the completion of said evaluation.

# 2. Evaluation of Student Advisement and College Service

- a. The professional staff member shall submit to the President of the College or his/her designee documentation relating to performance of his/her duties and responsibilities, including but not limited to a log of student advisement, if appropriate, and college service activities by December 30 and May 30. A tenured professional staff member shall be required to submit only a log of student advisement and a list of college service activities performed during a non-evaluation year.
- b. The President of the College or his/her designee shall review the document(s) and prepare an evaluation of student advisement and college service, except for tenured professional staff members in non-evaluation year. He/she may seek additional information from other sources. The President of the College or his/her designee shall forward a copy of the evaluation to the professional staff member by January 15 of the first appointment and by June 10 of the first and subsequent appointments and shall if requested meet and confer with the professional staff member to discuss the reasons therefor.
- c. The professional staff member shall have seven (7) working days in which to respond to the evaluation.

# 3. Summary Evaluation

The President of the College or his/her designe: shall after receipt of all the foregoing evaluation components and prior to February 1 of the first appointment and by June 15 of the first and subsequent appointments



develop a summary evaluation of each professional staff member, except for tenured professional staff members in a non-evaluation year, and shall consider only the foregoing evaluation components. The results of the components shall be applied in a uniform manner and shall be assigned the following weights:

Work Performance	90%
College Service	10%

# 13.04 Evaluation of Tenured Unit Members

- A. Upon receiving tenure, a unit member shall receive a summary evaluation every other year; provided, however, that a tenured unit member who receives an unsatisfactory summary evaluation shall again be evaluated in the subsequent year.
- B. For all faculty members tenured during the 1983-86 Collective Bargaining Agreement or earlier, one-half shall be evaluated in all components as well as a summary evaluation during the academic year 1987-1988; and one-half during the academic year 1988-1989 in order of inverse seniority within a division.
- C. For all professional staff members tenured during the 1983-86 Collective Bargaining Agreement or earlier, one-half shall be evaluated in all components as well as a summary evaluation by June 15, 1988; and one-half by June 15, 1989, in order of inverse seniority.
- D. All faculty members receiving tenure during the 1986-89 Collective Baryaining Agreement or later shall not be evaluated during the first year of tenure. All professional staff members receiving tenure during the 1986-89 Collective Bargaining Agreement or later shall not be evaluated during the first year of tenure after having moved to a June evaluation cycle.
- E. If a tenured faculty member is on leave status of any kind during the fall semester in a year in which he/she is scheduled to be evaluated, or if a tenured professional staff member is on leave status of any kind for more than six (6) months during a fiscal year in which he/she is scheduled to be evaluated, he/she shall be evaluated during the next year in which he/she returns to the College and every other year thereafter.
- F. At the end of each semester in a tenured unit member's non-evaluation year, the President of the College or his/her designee shall forward to the unit member a written acknowledgement listing those components required to be submitted and so received by the President or his/her designee and returned to the unit member (See Workload Materials Acknowledgement, XII 2, in the Form Supplement).

# 13.05 Reallocation of Weights Assigned to Evaluation Components

The weights assigned to e duation components shall be reallocated proportionally as determined by the President of the College or his/her designee, if the workload assignments of a unit member have been adjusted by assigning non-instructional activities to a faculty member pursuant to Article 12.03.

### 13.06 Written Reasons

Upon request, a unit member shall be provided with written reasons for his/her evaluation(s).

#### FORM SUPPLEMENT Student Evaluation of Instructor Page No. xii XIII-E1 Student Evaluation Summary Page No. xiv XIII-E1(S) Course/Instructional Materials Evaluation Form Page No. XIII-E2 XV Page No. XIII-E2 Attachment xvii Process for Classroom/Instructional Observation Page No. xviii XIII-E3 Page No. Student Advisement Log XX XIII-E4 Page No. xxii XIII-E5 College Service Evaluation **Faculty Summary Evaluation** Page No. xxviii XIII-E6 Page No. xxi Professional Staff Position Description XIII-E7 Page No. xxiv XIII-E8 Professional Staff Work Performance Evaluation **Professional Staff Summary Evaluation** Page No. xxi v XIII-E9

# ARTICLE XIV - CRITERIA AND PROCESS FOR PROMOTION OF FACULTY AND PROFESSIONAL STAFF UNIT MEMBERS

14.01 A prometion shall be defined as one step increase on the salary schedule in addition to the step increase as provided in Article 21.03.

### 14.02 Eligibility for Promotion

- A. Unit members who meet the following minimum qualifications shall be considered for promotion to the next higher step on the Salary Schedule, pursuant to the provisions of Article XXI.
- B. The unit member must have been deemed other than unsatisfactory on his/her most recent summary evaluation.



C. A unit member shall be eligible to apply for promotion when he/she has served two years in the bargaining unit from his/her initial appointment or since his/her last promotion, whichever is later.

D. As of the date of execution of this Agreement, unit members holding the following ranks or classifications shall be eligible for no more than the following number of promotions during the course of employment in the Community Colleges:

Instructor/Professional Staff I	Four
Assistant Professor/Professional Staff II	Three
Associate Professor/Professional Staff III	Two
Professor/Professional Staff IV	One

# 14.03 Announcement of Promotional Opportunities

- A. In accordance with Article 21.04 there shall be established at each College in the second and third years of the contract a pool equivalent to .25% of the total annual unit salaries of all current unit members whose salaries were paid from the State Maintenance Account before the implementation of the new salary schedule or the granting of any step or base rate increases, which will be used for promotions.
- B. The President of the College shall by October 1 each year declare the number of potential promotions as determined by the President of the College that he/she intends to utilize for qualified unit members. Each College shall make reasonable efforts, where financially practicable, to add an amount of money from the State Maintenance Account e quivalent to .5% of the total annual unit salaries of all current unit members whose salaries were paid from the State Maintenance Account before the implementation of the new salary schedule or the granting of any step or base rate increases, to the pool established in 14.03 A above.
- C. The President of the College or his/her designee shall not impose or use any quotas of any kind by rank or classification, college, department, division or work area and no such quota shall govern the eligibility for promotion of any unit member.
- D. Notwithstanding any provision in this Agreement to the contrary, the President of the College shall not be required to award more promotions than the pool can fund; provided, however, that the President will award as many promotions as the promotion pool will fully fund; provided further that the President shall not be precluded from funding additional promotions from other sources in his/her sole discretion.

## 14.04 Procedures

- A. Each unit member who is eligible for promotion pursuant to the criteria specified in this Agreement may apply for promotion to his/her immediate supervisor with a copy to the appropriate Dean prior to October 15 (See Faculty/Professional Staff Promotion Application, XIV-1, in Form Supplement).
- B. The immediate supervisor shall apply the following criteria uniformly and shall make a recommendation for or against promotion of those unit members eligible to be considered for promotion under the provisions of this Article. In making his/her recommendations, the immediate supervisor shall consider only the following in priority order:
  - 1. The unit member's performance, including all relevant material in his/her file since his/her last promotion or appointment to the College, whichever is more recent;
  - 2. The unit member's seniority.

The immediate supervisor shall apply the criteria in B above uniformly within his/her area of supervision.

- C. For those unit members being recommended for promotion, a list up to the number of promotional opportunities shall be established and forwarded to the appropriate Dean and on request to eligible unit members prior to March 1. The immediate supervisor shall, on request, provide a written statement of reasons to a unit member placed lower on the priority list.
- D. The appropriate Dean(s) shall review the recommendations and shall submit a prioritized list up to the number of promotional opportunities to the President of the College and on request to eligible unit members prior to April 1. The Dean shall consider only the criteria listed in 14.04B as they apply to his/her area of supervision. If the Dean disagrees with the immediate supervisor and changes the order of priority, the Dean shall give a statement of reasons in writing to the affected unit member.
- E. The President of the College shall review the recommendations of the immediate supervisor and the appropriate n(s) and shall consider only the criteria listed in 14.04B as they relate to the College in making his/her recommendations to the Board. If the President of the College disagrees with the Dean and changes the order of priority, he/she shall give a statement of reasons in writing to the affected unit member.
- F. Promotions snall become effective on the first (1st) day of the October pay period for faculty members and for proferzional staff.
- G. It is expressly understood that the provisions of this Article shall be applied in a manner which is not arbitrary, capricious, discriminatory or unreasonable.

# 14.05 Non-State Appropriated Funding

Service provided by a unit member while on non-state appropriated funding shall be counted for the purposes of this Article in the event that such unit member becomes an "01" employee; provided, however, that at the time of making application for a grant the President of the College or his/her designee shall when otherwise financially practicable and in conformance with law make reasonable efforts to request sufficient funds to allow for the promotion process for non-state appropriated funded unit members. It is understood by the parties that eligibility and criteria for promotion for non-state appropriated funded unit members shall be the same as for state-funded members.

### FORM SUPPLEMENT

XIV-1

Faculty/Professional Staff Promotion Application

Page No.

xxiv

# ARTICLE XIV-A - CRITERIA AND PROCESS FOR CHANGE OF RANK OF FACULTY AND PROFESSIONAL STAFF UNIT MEMBERS

# 14A01 Rank and Classification System

- A. There shall be four (4) ranks for faculty members:
  - 1. Instructor
  - 2. Assistant Professor
  - 3. Associate Professor
  - 4. Professor
- B. There shall be four (4) classifications for professional staff unit members:
  - 1. Professional Staff 7
  - 2. Professional Staff II
  - 3. Professional Staff III
  - 4. Professional Staff IV

### 14 A.02 Fligibility for Promotion

- A. L'aculty and professional staff unit members who meet the following minimum qualifications shall be considered for change to a higher faculty rank or professional staff classification:
  - 1. The unit member must have served at least two (2) years in his/her present faculty rank at his/her College as of September 15 or professional staff classification as of July 15.
  - 2. The unit member must have been deemed other than unsatisfactory on his/her most recent summary
  - 3. Qualifications for change of rank hereinafter set forth are not to be construed to limit the right of the recommending authorities to specify additional criteria when such criteria are customarily required for specialized or professional areas.
- B. Minimum Qualifications for Consideration for Change of Rank For Faculty

Rank	Degree*	Time in Rank	Total Experience
Assistant Professor	Doctorate	2 years	4 years
2.00000	Master's +		
	15-30 gr. cr. hrs. Master's or	2 years	5 years
	equivalent	2 years	6 years
Associate Professor	Doctorate	2 years	6 years
	Master's + 15-30 gr. cr. hrs. Master's or	2 years	7 years
	equivalent	2 years	8 years
Professor	Doctorate Master's +	2 years	8 years
	15-30 gr. cr. hrs. Master's or	2 years	9 years
	equivalent	2 years	10 years
ODerman most be from	recionally according institutions.		





# C. Computation of Total Experience for Faculty

For purposes of determining total experience, prior experience shall be credited as follows:

- 1. One (1) year of credit for each year of teaching experience within the Community Colleges or at a community college outside the System:
- 2. One (1) year of credit for each year of prior full-time post-secondary teaching at other than a community
- 3. One-half (1/2) year of credit for each year of prior related work experience to a maximum of ten (10) years of credit:
- 4. One-half (½) year of credit for each year of full-time directly related professional staff responsibilities at the college level:
- 5. One-half (1/2) year of cardit for each year of full-time elementary or secondary school directly related teaching experience to a maximum of ten (10) years of experience.
- 6. A Master's Jegree plus fifteen (15) thirty (30) graduate credit hours shall be the equivalent of one (1) year of experience.
- 7. A Doctorate shall be the equivalent of two (2) years of experience.

# D. Minimum Qualifications for Consideration for Change of Rank for Professional Staff

Class.	Degree*	Time in Class.	Total Experience
Prof. Staff	Doctorate	2 years	4 years
П	Master's + 15-30 gr. cr. hrs. Master's or	2 years	: 5 years
	equivalent	2 years	6 years
Prof. Staff	Doctorate	2 years	6 years
Ш	Master's + 15-30 gr. cr. hrs. Master's or	2 years	7 years
DC	equivalent	2 years	8 years
Prof. Staff	Doctorate	2 years	8 years
IV	Master's +		
	15-30 gr. cr. hrs. Master's or	2 years	9 years
*Degrees must be from	equivalent	2 years	10 years

<sup>\*</sup>Degrees must be from regionally accredited institutions.

# E. Computation of Total Experience for Professional Staff

- 1. One (1) year of credit for each year of experience in a directly related professional staff function within the Community College System or at a community college outside the System;
- 2. One (1) year of credit for each year of prior post-secondary experience in a directly related professional staff function at other than a community college;
- 3. One-half (½) year of credit for each year of prior related work experience to a maximum of ten (10) years of credit:
- 4. One-half (½) year of credit for each year of full-time college teaching experience;
- 5. One-half (½) year of credit for each year of full-time elementary or secondary school directly related professional staff function to a n. iximum of ten(10) years of credit;
- 6. A Master's Degree plus fifteen (15) thirty (30) graduate credit hours shall be the equivalent of one (1) year of experience.
- 7. A Doctorate shall be the equivalent of two (2) years of experience.

- 1. Part-time work and/or teaching experience shall be credited proportionally on the apprementioned criteria. No fractional total shall be credited.
- 2. No credit shall be given for summer session or evening division employment.
- 3. The Board may in its discretion waive the degree and/or experience criteria.
- 4. Degree requirements shall not apply to any unit member presently employed who does not have a Master's Degree.



- G. Fulfillment of the minimum formal requirements for change to a higher rank or classification and/or recommendations by the President of the College creates no right to change in rank or classification by implication and such authority remains within the sole discretion of the Board.
- H. The President of the College may waive the time in rank or classification and/or education and experience c. iteria requirement.

### 14 A.03 Procedures

- A. The President of the College shall not impose or use any quotas of any kind by rank or classification, college, department, division or work area and no such quota shall govern the eligibility for the change in rank of any unit member.
- B. Beginning with title changes effective in 1988, all unit members who meet the eligibility criteria in 14A.02 A1, A2 and A3 shall automatically be considered for change in rank or classification. The appropriate Dean shall determine who meets the above-referenced eligiousty criteria and shall then forward to the President of the College by March 15 each year his/her recommendation for a change in rank or classification or no change based upon the additional criteria provided in 14A.04.
- C. The President of the College shall review the recommendations of the Deans and shall announce his/her decision by April 15.

### 14A.0. Additional Qualifications

Ir addition to meeting the above-listed qualifications for the title changes effective in 1988, a unit member who also meets at least one of the following four criteria since the last change in rank or date of hire, whichever is later:

- A. evidence of significant relevant professional development;
- B. significant contribution to the College or community service;
- C. falling in the top 20% of the College faculty in his/her most recent two successive student evaluations;
- D. highly effective instructional performance of a faculty member or highly effective performance of a professional staff member in the professional judgment of the President of the College;

shall be accorded a change in rank; provided the additional criteria will not be used in any monetary award; provided further that it is the express intent of the parties that these additional criteria shall neither restrict nor expand the bases used for monetary awards in the 1983-86 Collective Bargaining Agreement which continued in this Agreement.

14A.05 It is expressly agreed that the implementation of this Article shall not result in the reduction of rank or classification of any unit member.

# ARTICLE XV - TERMINATION, DISMISSAL, DISCIPLINE & RESIGNATION

#### 15.01 Termination

Due notice of intention of the Board to terminate a unit member must be provided in accordance with Article XI.

### 15.02 Dismissal

Dismissal is defined as the discharging of a unit member for just cause prior to the expiration of his/her appointment and shall not be invoked except through due process.

- 1. Charges relating to dismissal must be filed with the President of the College and may be filed only by the appropriate administrator to whom the individual is responsible, except that charges may also be made by the President of the College.
- 2. The President of the College shall notify the unit member that he/she shall be recommended for dismissal and, if appropriate request is made, the dismissal shall be preceded by a discussion with the President of the College. The matter may be concluded by mutual consent of the parties at this point.
- 3. a. If the matter is not concluded, the President of the College shall frame with reasonable particularity a formal statement of charges and shall serve the statement of charges upon the unit member. The unit member may be represented by the Association.
  - b. Within ten (10) days of the service of the statement of charges upon the unit member, he/she shall indicate in writing whether he/she requests a formal hearing. If no such request is received within (10) days, it shall be presu ned that the unit member waives his/her right to a hearing.
- 4. The President of the College shar conduct the hearing, if held.
- 5. During the hearing, the unit member shall be entitled to have counse! of his/her choice which may include representation by the Association.
- 6. A verbatim record of the hearing shall be taken and a copy shall be made available to the unit member at his/her request for inspection and copying; provided that the Board and the Association shall share the cost of providing a copy of the transcript to the unit member.



- 7. a. The unit member shall be afforded an opportunity to obtain necessary witnesses and documentary evidence and the President of the College shall to the extent practicable secure the cooperation of such witnesses and make available necessary documents and other evidence within his/her control.
  - b. The unit member and the College shall have the right to confront and cross-examine all adverse witnesses. When witness cannot or will not appear, but the President of the College determines that the interests of fairness require admission of his/her affidavit, the President of the College shall identify the witness, disclose his/her affidavit, and give such statement appropriate probative weight in view of the parties' inability to cross-examine.
- 8. If the decision is for dismissal, the affected unit member may appeal to the Chancellor or his/her designee within ten (10) days after receipt of the decision. The entire record so established, including the transcript, shall form the grievance record and shall be forwarded to the Chancellor or his/her designee upon receipt of the Step One Appeal. Said appeal shall be construed as a grievance at Step Two of the Grievance Procedure, Article X, but the following expedited procedure shall apply:
  - a. The Chancellor or his/her designee shall meet with the affected unit member for the purpose of conducting a hearing and shall render a written decision within fifteen (15) calendar days of receipt of the appeal.
  - b. Within fifteen (15) calendar days after recaipt of the decision of the Chancellor or his/her designee or within fifteen (15) days after it was due, the Association may initiate arbitration of a grievance involving dismissal of a unit member in accordance with the rules of the American Arbitration Association.

### 15.03 Discipline

Nothing in this Article shall preclude the Board or its representatives from disciplining unit members by means less than discharge, including but not limited to suspension with or without pay, provided that such discipline shall be for just cause; and provided further that a unit member who is suspended without pay shall upon written request be entitled to a hearing within fourteen (14) calendar days after receipt of such request and to back pay in the event the suspension is reversed.

## 15.04 Resignation

A unit member shall give notice of resignation at least thirty (30) calendar days prior to the effective date of resignation, unless otherwise mutually agreed by the unit member and the President of the College or his/her designee.

15.05 Unless otherwise specifically modified herein, the provision of Article X shall be applicable to this Article.

# Al'TICLE XVI - FILLING OF VACANCIES IN THE BARGAINING UNIT

16.01 A notice of vacancy in the bargaining unit shall be dated and shall include a statement of duties, position title, salary range for the position, anticipated effective date, closing date of application and all qualifications for said position as determined by the President of the College or his/her designee and shall indicate the unit status of the position.

All notices of said vacancies shall be posted on bulletin boards customarily used to notify unit members. In addition, the Board shall administer Article XVI in a manner consistent with Article VIII and the Board's Affirmative Action Policy.

Copies of all unat vacancies as described above shall be forwarded to the President of the Association or his/her designee and the local Chapter President within three (3) days after the President of the College or his/her designee determines that a vacancy exists.

- 16.02 Vacancies as defined shall be filled by unit members within the College at which the vacancy occurs when in the professional judgment of the President of the College or his/her designee such unit members are the best qualified applicants. If the President of the College or his/her designee determines that two (2) or more applicants are equally best qualified, priority of consideration shall be given in the following order:
  - 1. to the retrenched unit member within the Community College System;
  - 2. to the unit member v thin the department where the vacancy occurs;
  - 3. to the unit member at the College where the vacancy occurs;
  - 4. to the unit member employed at another college in the Community College System.

Any appointments to a vacancy shall be consistent with the affirmative action goals and in accordance with the Board's Affirmative Action Policy.

### ARTICLE XVII - TRANSFER

### 17.01 Transfer Procedures

In the event the transfer is initiated by the unit member, application for transfer shall be made directly to the President of the College or his/her designee to which or within which transfer is desired; provided, however, that applicants shall inform the President of the College from which transfer is desired of any such application at the time application is made. Candidates for transfer are subject to the same selection procedures as any other application for an available



position. It is expressly understood that transfer is not automatic and is subject to the discretion of the receiving College; provided that if the President of the receiving College or his/her designee determines that two (2) or more applicants are equally best qualified, priority of consideration shall be given in the following order:

- 1. to the unit member within the department where the vacancy occurs;
- 2. to the unit member at the College where the vacancy occurs;
- 3. to the unit member employed at another College in the Community College System.
- 17.02 A unit member who is tenured at the time of his/her transfer to another Community College shall retain his/her tenure. A unit member with a regular appointment which provides him/her a just cause standard for termination shall retain such standard upon his/her transfer to another Community College.

## **ARTICLE XVIII - NOTICES**

# 18.01 Written Notices, Communications, Etc.

All notices, recommendations, reports and official communications required by this Agreement shall be in writing and shall be deemed to be given if delivered by hand or if mailed certified mail, return receipt requested, or first class mail, and addressed to the person concerned at his/her address as shown on the records of the College or if to administrators of the College or the Board to their regular place of official business.

# 18.02 Notice of Class Cancellations

If classes are cancelled all faculty and professional staff shall be notified in accordance with the notification procedure established at each College that they need not report to work; provided, however, that during the semester break, when classes are not in session, professional staff unit members shall be notified in accordance with the same procedure. Such procedure shall provide for either radio or telephone notice or both to be given at least one (1) hour prior to the beginning of the first (1st) scheduled class. In emergency situations, notice shall be given as soon as is practicable thereafter.

# ARTICLE XIX - RETRENCHMENT

### 19.01 Basis for Retrenchment

- A. The Board may from time to time retrench one (1) or more members of the bargaining unit whenever in the exercise of its sole discretion it shall have determined that such retrenchment is required due to bona fide financial reasons or where there occurs within an institution a bona fide discontinuance, reduction or shift in academic emphasis or professional service needs or for other related bona fide programmatic reasons.
- B. The provisions of this Article shall exclusively govern the discontinued employment of a member (s) of the bargaining unit prior to the expiration of his/her term of appointment through no fault or delinquency on his/her part. It is clearly understood that the nonreappointment of unit members pursuant to the provisions of Article XI is not covered by the terms of this Article.
- C. The President of the College shall notify both the President of the Association and the Chapter President in writing of impending retrenchment of unit member (s) and the reasons therefor, including any available documentary evidence pertaining thereto. The President of the Association or his/her designee may respond in writing to the President of the College as to his/her judgments and recommendations based upon the contents of the initial notification letter and/or subsequent to the consultation process described in 19.02.

### D. Notification

In addition to the notification given to the President of the Association under Article 19.01 C, notice of retrenchment to affected unit member (s) shall be made pursuant to the following terms and conditions:

- 1. Whenever the President of the College shall have determined that any unit member shall be retrenched under this Article, the President of the College or his/her designee shall give notice of retrenchment to the affected unit member and the effective date of such retrenchment.
- 2. Such notice shall be sent to the unit member affected as soon as practicable recognizing that, where circumstances permit, it is desirable that the effective date of said notice be sixty (60) days prior to the semester in which the employment of said retrenched unit member (s) shall be discontinued.

### 19.02 Consultation

- A. In addition to notifying the President of the Association and the President of the Chapter as hereinbefore provided, the President of the College or his/her designee shall meet and confer with the President of the Association or his/her designee regarding the administration's plans for the retrenchment of unit members and the reasons therefor.
- B. Accurate information, statistics or financial data related to any change or plan shall be made available by the President of the College or his/her designee for inspection and/or copying upon request of the President of the Association or his/her designee, provided, however, that this shall not require the College to compile such information, statistics or financial data in the form requested unless already compiled in that form.



- C. In adopting a plan of retrenchment, the judgment of the Board upon recommendation of the College shall be final; provided, however, that said retrenchment shall not be made in an arbitrary, capricious or unreasonable manaer.
- D. If and when retrenchment is to occur and it is otherwise financially practicable, the College shall make reasonable efforts to utilize attrition in order to effect the required reductions of unit members.

# 19.03 Reassignment

Whenever it shall have been determined to be necessary to retrench any unit member, the College shall reassign an affected unit member to a position within another division, department/work area or professional service area within the College at which the retrenchment occurs; provided, however, that such reassignment shall only be made to a then existing vacancy in such department/work area or professional service area. No such reassignments shall be made unless such unit member is qualified for such reassignment as determined by the President of the College or his/her designee. A unit member shall be deemed to be qualified by the President of the College or his/her designee if he/she has taught at least eight (8) sections at the college in the work area to which the reassignment is to occur, or, if the provisions of the retraining article, Article 19.08, apply. Such reassignment shall not be made without the assent of the unit member.

# 19.04 Retrenchment Procedure

- A. The President of the College shall make reasonable efforts to effect the required retrenchment by exhausting attrition and reassignment.
- B. Once a determination has been made to retrench unit members within a particular division, department, work area or professional service area the order of retrenchment shall be:
  - 1. part-time employees in the affected work area;
  - 2. temporary employee: in the affected work area;
  - 3. unit members in the affected work area according to reverse seniority.
- C. In selecting among and between full-time unit members, the seniority of each unit member within any department/work area/program, whichever is appropriate, at a College shall determine the order in which he/she shall be retrenched from that department/work area/program area, whichever is appropriate, so that the most senior such member shall be last retrenched and the least senior such member shall be first retrenched; provided, however, that such order of retrenchment shall govern only insofar as, pursuant to its initial application, those unit members to be retained are by treining, academic credentials, and/or experience as downwised by the President of the College or his/her designee qualified to teach the remaining courses offered by such department or within such program area; provided further that if a unit member is qualified to teach in more than one department/work area/program according to the criterion of having taught eight (8) sections in that work area, then college-wide seniority in the professional staff work unit and faculty work unit shall prevail in the event of retrenchment.
- D. A unit member who is aggrieved by the order of retrenchment shall upon request be notified of the reasons for such retrenchment as it relates to his/her employment being discontinued.

# 19.05 Seniority

- A. Seniority shall mean a unit member's length of continuous full-time unit service in the professional staff work unit and the faculty work unit at his/her College; provided, however, that seniority for returning administrators shall be calculated pursuant to the provisions of Article 11.04. Continuity of service shall not be broken by a leave of paid leave but shall not accrue during an unpaid leave.
- B. Seniority of a unit member who was on non-state appropriated funds shall be counted on a one (1) year to one (1) year basis in the event that such unit member becomes an "01" employee; provided, however, there is no break in service.
- C. Within sixty (60) days after the execution of this Agreement and by October 15 each year thereafter, the President of the College or his/her designee shall forward to the President of the Association or his/her designee a seniority list, which shall indicate a unit member's college-wide seniority in the professional staff work unit and the faculty work unit and the unit member's seniority in his/her department(s)/program area(s)/work area(s), whichever is appropriate.

### 19.06 Lay-Off Status

A. If a unit member is retrenched, no one shall be appointed to perform the released unit member's function within a period of four (4) years from the date of retrenchment; provided, however, that on or before June 10 of each calendar year subsequent to the date of retrenchment the released unit member shall give written notice by certified mail, return receipt requested, to the President of the College or his/her designee of his/her intention to be available on the recall list. Subject to the notice requirement as hereinbefore provided, such released unit member shall remain on the recall list until (1) he/she has been offered reappointment to his/her former function with equivalent tenure and no loss of benefits, or (2) said unit member has failed to meet his/her obligations under the provisions of this Article, (3) but not longer than four (4) years after the effective date of retrenchment.



B. Notice of vacancies shall be sent to the President of the Association by the President of each College or his/her designee.

### 19.07 Recall

- A. Whenever during the term of this Agreement it shall be determined to fill in whole or in part any unit position in any work area in which retrenchment has earlier taken place, the President of the College shall recall in inverse order of retrenchment the appropriate unit member who shall have been retrenched from such work area; provided that he she is qualified by training and/or experience to perform the duties of the position. Whenever an offer to recall has been extended and refused by a unit member, the offer to recall shall be extended to the next unit member on the recall list. For any unit member who accepts a recall appointment to his/her College, all previously accrued seniority and other contract rights shall be retained.
- B. In addition, a unit member who is retrenched and is deemed qualified by his/her training, academic credentials and/or experience as determined by the President of the receiving College or his/her designee for any available vacant unit position at the receiving college shall be given priority of consideration. A unit member who is tenured at the time of his/her hire at another Community College shall retain his/her tenure. A unit member with a regular appointment which provides him/her a just cause standard for termination shall retain such a standard upon his/her hire at another Community College.

### 19.08 Retraining

- A. Any unit member facing retrenchment shall be eligible for consideration for sabbatical leave regardless of his/her length of service and, if recommended by the President of the College or his/her designee to retrain for a suitable position which would be available concurrent with the date of retrenchment, such sabbatical leave shall be granted subject to the approval of the Board.
- B. No later than sixty (60) days after execution and ratification of th. Agreement, the Joint Study Committee shall establish a retrenchment retraining subcommittee which shall study and retrenched unit members.
- C. A unit member who is retrenched or who shall foreseeably be retrenched may request the approval of the President of the College or his/her designee to enter a retraining program without cost to the College and if such approval is granted he/she shall be extended priority of consideration for any position which the President of the College or his/her designee deems the unit member is qualified for subject to the availability of a position. Said retrenched unit member shall, upon acceptance in the regular day program at a public college or university within the Commonwealth, be granted a full tuition remission for the purposes of retraining subject to the approval of the Board of Regents or their designee.

# ARTICLE XX - DEPARTMENT CHAIRS/CURRICULUM COORDINATORS/PROGRAM COORDINATORS

# 20.01 Department Chairs/Curriculum Coordin tors/Program Coordinators

The President of the College or his/her designee may appoint on an annual basis a unit member as a department chair to assist in the coordination and/or supervision of instruction. Each such department chair shall have a job description specifying duties and responsibilities consistent with this Article and the needs of the College and he/she shall be responsible to the President of the College or his/her designee for performing such.

#### 20.02 Appointment

The department chair/curriculum coordinator/program coordinator shall be appointed by the President of the College in accordance with the procedures described in this Article. The term of the department chair shall be for a renewable term of one (1) year unless a vacancy is declared to exist by the President of the College as hereinafter provided or unless the department chair is unable to serve.

#### 20.03 Withdrawal from Duties

Notification of at least one (1) semester shall be given by a department chair whenever he/she does not wish to continue in the position of department chair. A department chair who is the only full-time faculty member in a department or work area cannot resign the department chair's responsibilities without also resigning his/her faculty position.

#### 20.04 Jurisdiction

The provisions of this Article shall be applicable to all unit members performing the duties and responsibilities of a department chair or curriculum coordinator/work area/program coordinator.

### 20.05 Department Chair/Curriculum Coordinator (Work Area)

- A department chair (work area) may be responsible for the following duties:
- A. The submission of the preferred subject matter preparation and preferred class schedule of unit members vithin his/her department consistent with Article XII.
- B. Implementing the process of evaluation of course materials contained in Article XIII. The department chair (work area) shall not be involved in whole or in part in any other evaluation process and shall not participate in any personnel action recommendations.



- C. Advise on the discipline competency of all applicants for vacancies within the department after consultation with members of the department.
- D. Other duties may be assigned to a department chair (work area) if specifically contained in a job description and such duties are consistent with the needs of the College and the provisions of this Article.

# 20.06 Department Chair/Curriculum Coordinator (Program)

The department chair (program) may be responsible for the following duties:

- A. Assist in the recruitment and orientation of new instructional staff;
- B. Advise on the instructional competency of all applicants for vacant positions within the program after consultation with members of the program unit;
- C. Submit the preferred subject matter preparation and class schedule of unit members within his/her program consistent with Article XII;
- D. Supervise the development of instructional materials and assist in conducting research on the effectiveness of the instructional program consistent with the philosophy and objectives of the College, the requirements of external and/or regulatory agencies and Article VII of this Agreement;
- E. Instruct courses or portions thereof within the program as appropriate;
- F. Assist in the implementation of the following evaluation processes as contained in Article XIII.
  - 1. The process of evaluation of course materials.
  - 2. The process for classroom observation in a clinical or laboratory-like setting wherein the student is developing a variety of occupational and/or preprofessional skills fundamental to the students' performance after completion of designated programs. The department chair (program) shall not implement the process for evaluation by students nor the process for classroom observation where didactic instruction is involved.
- G. Encourage faculty to develop new methods of instruction;
- H. Develop and prepare for submission all reports and accreditation materials required by governing or accrediting agencies;
- I. Participate in the formulation of the program's budget and administer it within the prescribed limits established by the College;
- J. Evaluate and make recommendations for any employees not in the faculty/professional staff unit assigned to his/her program area;
- K. Cooperate with the President of the College or his/her designee in the development, dissemination and implementation of Board/College policy, regulations and procedures;
- L. Meet with the Advisory Committees which support or influence the instructional program in cooperation with his/her supervisor. Establish liaison and cooperation with external agencies essential to the implementation of the program particularly where use of external facilities or resource personnel is required;
- M. Cooperate and facilitate cooperation with other program areas and/or departments, learning resources, student services and administrative services within the College;
- N. The designated administrator shall implement all evaluation processes for the department chair and the coordinator, if said coordinator does not report to a department chair. If the coordinator reports to a department chair, the process for the evaluation of course materials shall be performed by the department chair;
- O. Other duties may be assigned if specifically contained in a job description and such duties are consistent with the needs of the College and the provisions of this Article:
- P. The department chair (program) shall be responsible for the total implementation of a college program and/or curriculum defined as a series of diverse courses usually leading to a certificate or associate degree and which require multiple faculty in order to provide core courses and where accreditation, licensure or other external governing agencies require a member defined within the unit to perform supervisory functions for approval, maintenance or continuance of the program.

# 20.07 Procedure for the Selection of Department Chair\*

At least two (2) months prior to the expiration of a term of office of a department chair, or upon a declaration of a vacancy, the President of the College shall notify the members of the discipline/work area/program of the need to select a chair. The following procedures shall be followed in the selection of a department chair:

- 1. The President of the College will post the job description for seven (7) calendar days within the College;
- 2. The members of the department/program have seven (7) calendar days to make application;
- 3. At the expiration of this period of time, if the President of the College or his/her designee determines that there are no applicants qualified for the position, the President of the College or his/her designee may post outside the College;
- 4. When a department chair (discipline/work area) is to be selected, the process shall be:

  At the conclusion of the posted period, members of the department shall meet to review the applicants for the position. Thereafter, the members of the department/program will by secret ballot elect one (1) person from the



applicants and thereafter make a recommendation to the President of the College or his/her disignee. If acceptable, the President of the College shall within ten (10) working days of his/her receipt of such nomination recommend appointment of such nominee to the Board.

If unacceptable, the department shall recommend an additional candidate. If the second nominee is unacceptable, the duties of evaluation of course materials shall be assigned to a non-unit member and the balance of responsibilities contained herein shall be performed by the department/work area member originally recommended to the President of the College or his/her designee.

5. When a department chair (program) is to be selected, the process shall be:

At the conclusion of the posted period whether by internal or external recruitment processes, members of the department/program shall meet to review all applicants for the position. Thereafter, the members of the program area will by secret ballot determine whether each applicant is qualified according to the qualifications criteria in the job description and certify without ranking them to the President of the College or his/her designee. The President of the College or his/her designee shall recommend the appointment of the department chair (program) to the Board.

\* Program Coordinator shall be included in the process.

# 20.08 Evaluation of Department Chair

A. Each department chair shall be evaluated annually by the President of the College or his/her designee in writing not later than March 30 of each contract year. The evaluation shall be based upon the accomplishment of tasks which have been established and agreed upon between the department chair and the President of the College or his/her designee.

The evaluation shall also include the evaluation by each unit member within the work area/program, the results of which shall be recorded on a form (See Department Chair Evaluation Forms XX - 1 or XX - 2 in Form Supplements); provided, however, that any changes made on the form by the Board shall be preceded by consultation with the Joint Study Committee. The President of the College or his/her designee shall administer the evaluation form in a manner to insure the anonymity of the unit member responding; provided, however, that a department chair who is non-reappointed based on these evaluations may, upon request, have direct access to the evaluations and the identity of the authors thereof. The evaluations by unit members shall be used for the sole purpose of the recommendation to appoint/non-reappoint the department chair.

B. The President of the College or his/her designee shall notify the department chair in writing of his/her recommendation for appointment/non-reappointment based upon the above evaluation process. The department chair may respond within seven (7) working days to the evaluation. After the recommendation is implemented, the department chair evaluations completed by unit members shall be removed from the personnel file of the evaluated unit member.

### 20.09 Recall

The parties recognize that the recall of a department chair by discipline/work area/program members is an unusual occurrence and that such recall should be based upon extraordinary circumstances. The parties agree, therefore, that the following procedures for the recall of a department chair by discipline/work area/program members shall not be used until a department chair has served at least one (1) academic year from the date of appointment.

#### A. Informal Procedures

Within five (5) working days of the receipt of a statement setting forth specific complaints signed by one-third (1/3) of all full-time department members, the department chair shall meet to confer with all members of the department work area. This meeting shall be for the purpose of attempting to find an informal resolution of any complaints set forth in the signed statement.

If the complaints are not resolved to the satisfaction of one-third (½) of the full-time department/work area members, the concerned members should next meet informally with the Academic Dean to discuss the matter and obtain, if possible, the Dean's assistance in reaching an informal settlement of said complaints. Failing this the department/work area members may then proceed to the formal recall procedures hereinafter described.

### **B. Formal Procedures**

- 1. Upon presentation to the Academic Dean of a petition signed by one-third (½) of the full-time members of the department/work area, excluding the department chair, stating specific reasons for recalling the department chair, the Academic Dean shall promptly give fourteen (14) calendar days' written notice to all department/work area members setting forth the time, date and place of a meeting to consider the recall petition and to vote on either a motion that the department/work area chair continue in office or a motion to recommend to the President of the College that he/she declare a vacancy to exist in the department/work area.
  - The department chair/work area may be present at this meeting.
- 2. The Academic Dean and an in. partial person from the faculty at large, who shall be elected by members of the department/work area, shall conduct the recall meeting, and if the Academic Dean and the members of the department/work area shall have so decided, shall conduct successor meetings for the same purpose. The Academic Dean and such impartial person from the faculty at large shall record any subsequent vote(s) taken within the department/work area on this matter.

46



3. A vote by secret ballot of the majority of all full-time department/work area members shall be required to recommend to the President of the College or his/her designee that he/she declares a vacancy to exist in the department chair position. If a majority of the department/work area members so vote, the rusults of the balloting with reasons shall be forwarded to the President of the College or his/her designee. The President of the College shall determine the recall or continuance within ten (10) calendar days and so notify the department/work area with reasons. The President's decision shall be final.

# 20.10 Workload Reduction

Any unit member who performs the duties and responsibilities of a department. Lair or curriculum coordinator/work area/program coordinator as contained in this Article shall receive a workload reduction of at least one (1) section in accordance with Article XII, Section 12.05 D, whether or not he/she holds such title.

### 20.11 Compensation

Any unit member who agrees to perform the duties and responsibilities of a department chair or curriculum coordinator/work area/program coordinator as contained in this Article shall be compensated at the rate of \$25.00 for each working hour if he/she agrees to perform such duties between Commencement and the first day of Fall classes, during Winter intersession and during Spring vacation; provided, however, that such compensation shall not be paid for duties performed on assigned professional days as provided in Article 12.03 D. It is expressly understood that the President of the College or his/her designee shall inform the department chair, or curriculum coordinator/work area/program coordinator in a timely fashion should he/she desire to secure the services of the aforementioned unit member consistent with this Article.

### FORM SUPPLEMENT

XX - 1	Department Chair/Curriculm Coordinator Evaluation Form Department Chair/Work Area Evaluation Form	Page No.	xxv
XX - 2		Page No.	xxvi

# ARTICLE XXI - SALARY ADJUSTMENTS

21.01 Effective June 29, 1986, unit members will be moved to the closest step on Salary Schedule Column A that is at least three hundred dollars (\$300.00) more than their current salary.

# 21.02 General Salary Rate Increases

- A. Effective June 29, 1986, all full-time unit members shall receive a general salary rate adjustment of 1.5% by moving to Salary Schedule Column B.
- B. Effective June 28, 1987, all full-time unit members shall receive a general salary rate adjustment of 3.9% by moving to Salary Schedule Column C.
- C. Effective June 26, 1988, all full-time unit members shall receive a general salary rate adjustment of 5.0% by moving to Salary Schedule Column D.

# 21.03 Performance Based Salary Adjustments

- A. Effective September 28, 1986, unit members who received other than unsatisfactory on their most recent performance evaluations shall receive performance-based salary adjustments equal to one step on Salary Schedule Column B.
- B. Effective September 27, 1987, unit members who received other than unsatisfactory on their most recent performance evaluations shall recieve performance-based sala. y adjustments equal to one step on Salary Schedule
- C. Effective September 25, 1988, unit members who received other than unsatisfactory on their most recent performance evaluations shall receive performance-based salary adjustments equal to one step on Salary Schedule Column D.

### 21.04 Promotions

### A. Fiscal Year 1988

Effective September 27, 1987, there shall be created at each College a pool of money equivalent to .25 percent of the total annual unit salaries of all current unit members whose salaries were paid from the State Maintenance Account as of September 26, 1987, prior to the salary increases of September 27, 1987, as provided for in Article 21.03, to fund premotions in accordance with Article 14.03.

#### B. Fiscal Year 1989

Effective September 25, 1988, there shall be created at each College a pool of money equivalent to .25 percent of the total annual unit salaries of all current unit members whose salaries were paid from the State Maintenance Account as of September 24, 1988, prior to the salary increases of September 25, 1988, as provided for in Article 21.03, to fund promotions in accordance with Article 14.03.



47. 98

### 21.05 Additional Compensation

- A. In those Colleges where faculty are currently required to perform duties beyond the academic year as defined in Article 12.03 E1 or may be required to do so in accordance with the provisions of Article 12.03 E2, those faculty members shall receive a salary adjustment in direct proportion to the additional duties assigned, as provided in paragraphs B and C below.
- B. In those programs with an instructional component extending beyond the academic year for which the faculty member is responsible for providing direct instruction, such additional compensation shall be based on 1/160 (0.00625) of his/her annual salary for each day worked and shall be in remuneration for both instructional and non-instructional workload in accordance with Article XII. To the extent that a faculty member's assigned instructional summer workload exceeds his/her average instructional workload during the academic year, he/she shall receive an additional 0.00084 of his/her annual salary for each instructional hour beyond that average.
- C. Where faculty are only required to perform occasional hours of field-based work, the number of hours shall be determined by the President of the College or his/her designee after consultation with the faculty member before such additional duties are scheduled to commence. Additional compensation shal'; computed by multiplying 0.00084 by the faculty member's annual salary by the number of hours assign.
- D. Faculty who are requested to and agree to perform work covered by this Agreement beyond the academic year in the programs listed in Article 12.03 E6 shall be compensated in accordance with paragraphs B and C above.

# 21.06 Educational Needs/Professional Development

Effective June 29, 1986, again effective June 28, 1987, and effective June 26, 1988, there shall be established at each College a pool of 1% of total salaries for educational needs and professional development purposes. Each pool will be calculated before the application of general increases provided in 21.02. It is understood that these adjustments do not affect base rates. These pools shall be distributed in accordance with the following procedures. Participation in any such program by any member of the bargaining unit shall be undertaken only on a voluntary basis.

### A. Purpose

- 1. To improve teaching and student advising and to develop new course materials and to relate those to a changing curriculum that is itself responsive to the larger needs of society:
- 2. To develop new teaching skills and an appropriate facility in the use of media, teaching aids and other supportive techniques where those are appropriate;
- 3. To improve student advising techniques and the faculty member's command of relevant bodies of knowledge in this area;
- 4. To increase the faculty member's command of the body of knowledge that constitutes his/her own discipline;
- 5. Where appropriate, to enable a faculty member to develop a command over a body of knowledge in a related discipline;
- 6. In the case of any professional staff unit merror, to improve such professional staff unit member's professional skills and techniques and to relate those to a changing curriculum that is itself responsive to the larger needs of society; and
- 7. In the case of any professional staff unit member, to develop new professional skills and techniques to the same end.
- B. Unit members shall submit applications in writing for funding for their individual programs of professional development accompanied by receipts (See Application Form for Educational Needs/Professional Development, XXI-1, in Form Supplement); requests for equipment to be used outside the classroom shall be accompanied by a statement of particular purpose expanding on one or more of the purposes in 21.04A above.
- C. In Fiscal Years 1987 and 1988 no unit member shall be unreasonably denied a minimum of \$600 in educational needs funds provided he/she submits a bona fide request in accordance with A. and B. above. In Fiscal Year 1989 no unit member shall be unreasonably denied a minimum of \$330.00 in educational needs funds provided he/she submits a bona fide request in accordance with A. and B. above.

# D. Individual Programs of Professional Development

Within thirty (30) days of the date of execution of this Agreement, there shall be established at each College a Committee on Professional Development and Retraining, which shall be composed of three (3) unit members appointed by the President of the Association and two (2) persons appointed by the President of the College.

### 1. The Committee shall:

- a. Meet and confer with the Dean of Academic Affairs regarding criteria for the awarding of Professional Development Programs and shall utilize the guidelines for such Professional Development Programs;
- b. Receive and review all requests for programs of professional development as may be undertaken pursuant to the provisions of this Article, and the guidelines in effect and thereafter make recommendations to the President of the College regarding approval or disapproval of all programs of professional development submitted pursuant to the provisions of this Article;



48

- c. Make recommendations regarding the allocation of monies which shall be available for the implementation of programs of professional development as may be undertaken pursuant to the provisions of this Article;
- d. Develop and recommend a policy for the retraining of unit members who may be retrerched, and submit such recommended policy to the President of the College within three (3) months from the date on which the Committee shall have first convened; in discharging this responsibility, the Committee shall seek to compile and disseminate information regarding retraining and job opportunities for members of the bargaining unit.
- e. Advise, when requested, individual unit members who shall have been or may be retrenched pursuant to the provisions of Article XIX of this Agreement for the purpose of assisting any such unit member in the development of a program of retraining.

### 2. The President of the College shall:

- a. Upon receipt of the recommendations of the Committee on Professional Development and Retraining, grant such awards as he/she shall determine in accordance with the provisions of this Article;
- b. Notify the applicants of the approval or disapproval of their program of professional development;
- c. Notify the President of the Association of the grants and programs approved by him/her.

### E. Definition

For the purpose of this Article, the phrase "individual program of professional development" shall in the case of any individual member of the bargaining unit mean a program designed to effectuate any one or more of the purposes set forth in Section A, which program shall have been developed at the initiation of such unit member, assented to by him/her, and approved as such by the President of the College or his/her designee; it being the common purpose of the parties to this Agreement to provide members of the bargaining unit with the specific means by which to achieve professional growth and to benefit the College.

		SALARY SCHEI	DULE	
STEP	A	В	c	D
1	19809	20107	20891	21955
2	20709	21020	21840	22952
3	21609	21934	22789	23950
4	22509	22847	23738	24947
5	23409	23761	24687	25945
6	24309	24674	25636	26942
7	25209	25588	26586	27940
8	26109	26501	27535	28937
9	27009	27415	28484	29935
10	27909	28328	29433	30932
11	28809 .	29242	30382	31930
12	29709	30155	31331	32927
13	30609	31069	32280	33924
14	31509	31982	33230	34922
15	32409	32896	34179	35919
16	33309	33809	35128	36917
17	34209	34723	36077	37914
18	35109	35636	37026	38912
19	36009	36550	37975	39909
20	36909	37463	38924	40907
21	37809	38377	39874	41904
22	38709	39290	40823	42902
23	39609	40204	41772	43899
24	40509	41117	42721	44897
25	41409	42031	43670	45894
26	42309	42944	44619	46892
27	43209	43858	45568	47889
28	44109	44772	46518	48887



# ARTICLE XXII - MANAGEMENT ASSOCIATION COMMITTEE ON EMPLOYEE RELATIONS

# 22.01 College Level

XX - 1

There shall be established a committee at the College level to be known as the Management Association Committee on Employee Relations. Such Committee shall be comprised of six (6) members: three (3) representing the local College administration and three (3) representing the Association. Such representatives shall be respectively appointed by the President of the College and the President of the Chapter. In addition, the President of the College shall designate the chairperson for the College and the President of the Chapter shall designate the chairperson for the Association.

The purpose of said Committee shall be to discuss matters of mutual concern to the employee and the employer. There shall be at least one (1) meeting per month during the academic year with the chairpersonship alternating between the College and the Association; provided, however, that whenever the parties mutually agree there is no need for a meeting during a month there shall not be a meeting.

Both parties may submit items for the agenda to the chairperson at least two (2) weeks in advance of any scheduled Committee meetings. The agenda shall be distributed one (1) week in advance of any scheduled committee meetings.

It is understood that said Committee shall have no power to negotiate, alter, or amend the terms of this Agreement.

# ARTICLE XXIII - NO STRIKE OR LOCK OUT PLEDGE

23.01 he Board agrees that it shall not lock out any or all of its employees for any cause during the term of this Agreement and the Association and its agents agree that they shall not engage in, induce or encourage any strike, work stoppage, slow down or withholding of services by said members.

Nothing contained in this Article shall be deemed to waive, impair or restrict the right of the Board or the Association to seek or pursue any remedy at law or equity provided by the laws of the Commonwealth in the event of a violation of this Article.

# ARTICLE XXIV - SAVINGS CLAUSE

24.01 If any of the provisions of this Agreement shall in any manner conflict with or contravene any federal or state law, statute or the rules and regulations promulgated thereunder, such provisions shall be considered null and void and shall not be binding on the parties.

The Board and the Association acknowledge that during the negotiations which resulted in this Agreement each had the ultimate right and opportunity to make demands and proposals with respect to any subject or matter not removed by law from the applicable areas of collective bargaining, and that the understanding and agreements arrived at by the parties after the exercise of that right and opportunity are set forth in this Agreement and shall constitute the sole Agreement between the parties.

In recognition of this fact the Board and the Association for the life of this Agreement each voluntarily and unqualifiedly waives the right and agrees that the other shall not be obligated to bargain collectively with respect to any subject or matter not in this Agreement even though such subject or matter may not have been within the knowledge or contemplation of either or both parties at the time they signed this Agreement; provided, however, that nothing in this Article shall prohibit the parties from conducting negotiations during the term of this Agreement regarding the impact on terms and conditions of the Board or its successor to close any College or to merge any College with another educational institution to consolidate, discontinue, or transfer existing functions, educational activities and programs.

The provisions of this Article notwithstanding, the parties may by mutual agreement upon the request of one (1) or both parties reopen negotiations on the provisions of this Agreement prior to the expiration date provided in Article XXVI.

### ARTICLE XXV - HOLDOVER

25.01 In the event that the Board and the Association shall fail to secure a successor Agreement as hereinafter provided in Article XXVI prior to the termination of this Agreement, then this Agreement shall remain in ft." force and effect until a successor agreement is executed or an impasse in negotiations is reached.



101

# ARTICLE XXVI - DURATION AND SUCCESSORSHIP

26.01 This agreement shall be in full force and effect from and after July 1, 1986 through June 30, 1989; provided, however, that nothing herein contained shall be deemed to impose on the Board any obligation the discharge of which may be required to be sought pursuant to General Laws, Chapter 150 E, Section 7, until such time as such appropriation shall have been duly made by the General Court pursuant to said provision of the General Laws; provided further that notwithstanding the foregoing, whenever the General Court shall not have acted pursuant to said provision and the Board shall have monies allocable to the discharge of any obligation herein contained and any such monies shall at the sole discretion of the Board have been so allocated, such obligation shall be discharged in such measure as such monies so allocated shall permit.

If, in respect to this Agreement, the Governor shall have failed to act pursuant to General Laws, Chapter 150E, Section 7, the Association shall have the right upon thirty (30) days' written notice to the Board to require that the parties to this Agreement shall resume collective bargaining pursuant to the provisions of General Laws, Chapter 150E.

Except as provided by express agreement of the parties, no claim of grievance of any collective bargaining agreement between the Board and the MTA/MCCC shall have validity that may be alleged to have occurred prior to the beginning of Academic Year 1986-1987.

In accordance with applicable provisions of Section 10 of Chapter 15 A of the General Laws, each local Board of Trustees for each College covered under this Agreement shall appoint, transfer, dismiss, promote, and award tenure to all personnel of said College, subject to policies promulgated or agreements entered into by the Board of Regents.

To the extent permitted by law, the Board of Regents of Higher Education may delegate its authority or any portion thereof to the local Board of Trustees for each College whenever in its judgment such delegation may be necessary or desirable.



### SIGNATURE PAGE

For the Massachusetts Board of Regents of Higher Education and the Massachusetts Regional Community Colleges

Carolyn R. Young Associate Community College Counsel & Chairperson

Carleton H. LaPorte Director of Employee Relations,

Date:

For the Massachusetts Teachers Association/Massachusetts Community College Council

Chairperson

MCCC Negotiating Team

Ey:

John nc-Negotiator Massachusetts Teachers Association

Date: <u>TUNE</u> /6, 1987

### APPLICATION FOR SABBATICAL LEAVE

College	Work Area
Number of years of seniority in the collect	tive bargaining unit
Number of years since last previous sabbat	cical
Check the type of sabbatical for which you	ı are applying:
<ul> <li>( ) Half year leave at full salary</li> <li>( ) Half year leave at half salary</li> <li>( ) Full year leave at half sorklos</li> <li>( ) Full year leave at half worklos</li> <li>( ) Full year leave at half worklos</li> </ul>	ed at full salary sad at half salary
Date on which proposed sabbatical would be	egin
Use the rest of this form and/or a separate	te sheet appended to this form to answer the

following questions:

- A. What activities will you do during the proposed sabbatical leave and what goals are these intended to achieve?
- B. How will the proposed sabbatical meet the following criteria listed in section 9.01, 1, 2, a of the collective bargaining agreement?

"The following criteria shall be considered in determining who shall be granted for sabbatical leave:

- (1) That the objectives of the sabbatical leave, if attained, would substantially contribute to the professional growth of the unit member.
- That the objectives of the sabbatical leave, if attained, would assist the unit member in substantially contributing to institutional needs and attainment of institutional purposes.
- (3) That the unit member has the ability to achieve the goals of the project or plan based on as/her past experience and formal educational background.
- (4) That the attainment of objectives of sabbatical leave as proposed are realistic in terms of time, costs, and other related variables.
- (5) There exists independent financial support from other funding sources concerned with the proposed plan or project where College funding sources are otherwise unavailable."



Name

# STEP ONE COMPLAINT

			For Board Use		]
			Year: Board No.:		] 
<b>50</b> 555		,	DOGIG NO		j
TO PRE	SIDEPT				
GRIEVA	(last)	103	<del></del>		
		(first		middle)	
	(rank/job t	itle and cl	assification)		
DEPARTI	ENT/DIVISION/W	ORK AREA		•	
Date(S)	OF ALLEGED CO	NTRACT VIOL	ATION		
support space : caption	is needed, pleaded.):	must be ase attach	attached heret additional pa	sed. All evide to If addition ges, appropriate	mal
Specifi	Contract Prov	visions Alle	eged to Have Be	en Violated:	
Remedy	Requested:		<del></del>		
		Sig	nature	Date	
		- •		2400	
		Hom	e Address	Tel. N	<u>. o.</u>
c: MT/ Pla	Consultant for Consul	r Higher Ed 02108	ucation, MCCC/	MTA, 20 Ashburto	n
Der Ros	nis Fitzgerald d, Plymouth, M	, MCCC G ie A 02360	vance Coordina	tor, Rocky Pond	
.B. Th	ais complaint m	ust be file	d within 30 ca	lendar days.	



ij

### **ASSOCIATION EVIDENCE**

For Board Use Year: Board No.:

- 1. List on this cover sheet all documentary evidence you intend to use to support your grievance.
- 2. Attach copies of all evidence to this sheet and identify each document with the number assigned below.

# <u>Description of Evidence</u>

(Include Dates of Correspondence)

1.

2.

3.

4.

5.

6.

7.

8.

9.

10.



# MANAGEMENT EVIDENCE

For Board Use Year: Board No.:

- 1. List on this cover sheet all documentary evidence you intend to use to support your finding.
- 2. Attach copies of all evidence to this sheet and identify each document with the number assigned below.

# Description of Evidence

(Include Dates of Correspondence)

1.

2.

3.

4.

5.

6.

7.

8.

9.

10.



# STEP ONE DECISION

	For Board Use	
	Year: Board No.:	
	<del></del>	
GR1	LEVANT	
COI	COLLEGE	
res	er reviewing the complaint and supporting evidence attached ereto and after meeting with the grievant for the purpose of solving the grievance on	
1.	Statement of Facts:	
2.	Issue(s) presented by the grievant, including specific contract provisions alleged to have been breached:	
3.	Decision and Reason(s) for Decision:	
4.	Remedy offered, if appropriate:	
Pre	sident or Designee Date (must be issued within thirty (30) days after receipt of grievance)	
cc:	MTA Consultant for Higher Education, MCCC/MTA, 20 Ashburton Place, Boston, MA 02108	
	Dennis Fitzgerald, MCCC Grievance Coordinator, Rocky Pond	

N.B. You have the right to appeal this Decision to Step Two py filing an appeal on Form G5 within ten (10) calendar days after receipt of this Decision.

Road, Plymouth, MA 02360



# STEP ONE APPEAL

	Year	Board Use : d No.:	
то	Co: Chancellor's Designee Community College Counsel's 75 Pearl Street Reading, MA 1867	Office	
FROM	M: Grievant		
	Grievance Issue(s)		<u>:</u>
	I hereby appeal the Step One	Decision of the Property Community Community Community	
	Home	Address	
	Tele	phone Number	
cc:	MTA Consultant for Higher Educ Place, Boston, MA 02108	cation, MCCC/MTA,	20 Ashburton
	Dennis Fitzgerald, MCCC Grieva Road, Plymouth, MA 02360	ance Coordinator,	Rocky Pond
	College President		
N.B.	. This appeal must be filed with after receipt of the Step One		endar days



N

#### STEP TWO

# NOTICE TO MEET WITH THE CHANCELLOR OR HIS/HER DESIGNEE

		Year:	No.:		
Grie	(last) (fi			(middle	
Home	Address			•	<u> </u>
Date	Grievance Received & Iss	:ue(s) _			
_	This is to acknowledg grievance and to adviso made for you to meet follows to discuss this is invited to attend.	e you to	that an a the Chance	ppointm llor's	ent has been designee as
	Designee				
	Place				
	Date				
	Time				
	Would you arrange your order to attend this munder the contract requivithin forty (40) days f	eeting re a de	as the ti cision by	me con the Ch	straints ancellor
		Chance	llor or De	esignee	
		Date			<u> </u>
cc:	MTA Consultant for Highe Place, Boston, MA 02108	r Educa	tion, MCC	C/MTA,	20 Ashburton
	Dennis Fitzgerald, MCCC Road, Plymouth, MA 0236	Grievan O	ce Coordin	nator,	Rocky Pond
	College President				



# STEP TWO DECISION

		For Board	Use	
		Year: Board No.		
		Board No.		
Griev	(last)	(first)	(middle)	
Home	Address			—
Presi	ident of the College,	new evidence	e record submitted by a submitted at Step Two, the meeting held on my designee, I make	
1. 8	Statement of Facts:		:	
2.	Issue(s) presented by contract provisions al	the grievant, leged to have	including specific been breached:	
3. 1	Decision and Reason(s)	for Decision	1:	
<b>4.</b> i	Remedy offered, if app	propriate:		
Chan	cellor's Designee	Date	<u> </u>	
cc:	MTA Consultant for Hi Place, Boston, MA 02	igher Educatio 2108	on, MCCC/MTA, 20 Ashburt	on
	Dennis Fitzgerald, MC Road, Plymouth, MA C	CCC Grievance 02360	Coordinator, Rocky Pond	i.
	College President			
N.B.	arbitration. An ark	bitration appl the MCCC Gri	to appeal this Decision roval request (Form G8) levance Coordinator with receive this Decision.	



viii

#### ARBITRATION APPROVAL REQUEST

To be completed by the grievant and forwarded to the Grievance Coordinator within ten (10) calendar days after receipt of the Step Two Decision.

For Board Use

		DOBIG N	<del>].i</del>	<del></del>
GRIEV	(last)	(first)	(middle)	
COLLE	GE			
DATE	OF DECISION OF	CHANCELLOR	<u>;</u>	
elect am re	ion to proceed	that I am hereb to Step Three of my grievance be de committee.	the grievance p	rocedure. I
REASO	NS FOR THE DECIS	SION		
				-
		Signatu	re	Date
	MTA Consultant i	or Higher Educat: A 02108	ion, MCCC/MTA, 2	0 Ashburton



ix 112

Dennis Fitzgerald, MCCC Grievance Coordinator, Rocky Pond

Road, Plymouth, MA 02360

UNITS OF INSTRUCTION			INSTRUCTIONAL HOURS					
COURSE	CONTACT HOURS	MULT. FACTOR	UNITS	COURSE	MILT. FACTOR	PREP. Time	CONTACT TIME	INSTRUCT.
	the units nar course		ction and	instruction	nal hours	for first	sections o	f didactic
				Į)				•
Compute				14				ons of above.
		x 1 = x 1 =			X 2/3 = X 2/3 = X 2/3 =		<u>+</u>	
	the units		ction and	instructio	nal hours	for first	sections o	of lab-like
<u>=</u>	<u> </u>	X 2/3 = X 2/3 = X 2/3 =	=		X 2 X 2 X 2		<u>+</u> <u></u>	===
Compute	the units	of instru	ction and	instructio	nal hours	for subse	quent secti	ons of above.
		X 2/3 = X 2/3 = X 2/3 =	=		X 1		<u>+</u> ===	=
-				11			teaching co	
<u>=</u>	=	<u>x</u> = :			<u>x</u> = :		÷==	= ==
				instructional mode			vidualized i	instruction,
=		X 1/2 = X 1/2 = X 1/2 =	$\equiv$		X 1/3 : X 1/3 : X 1/3 :		<u>+</u>	===
	ent chairp						se reduction tions for a	
		$\frac{x}{x} = \frac{1}{x}$			X 2 X 2 X 2			
Total U	nits of Ir	nstruction				al Instru	ctional Hou	rs

The first part of the workload computation is subject to a standard of 24 units of instruction per academic year. If an assignment of over 12 units of instruction is made in the fall semester, the projected instructional workload for the spring semester may be requested. Second, an assignment of more than 24 instructional hours per semester will require an adjustment in the 13 hours of non-instructional workload.



	WORKLOAD M	A I ERIAIS ACRIUNLEDGEMEN I
o:		
M:		
B:		
	rill acknolwedge that, 19 semester rignee and returned to	the following materials for the have been received by the President or the unit member
	Course Materials	
	Student Advisemen	nt Log
	List of College S	Service Activities Performed
		Signature President or Designee
		Signature Unit Member



		Campus

#### STUDENT EVALUATION OF INSTRUCTOR

Your cooperation is requested by the Faculty and the Administration on providing information regarding the instructor of the course(s) you are now taking. Evaluation is fundamental for the maintenance of academic and professional standards of excellence, and for the promotion of professional growth. The information you give will be used to assist the College in making decisions regarding your instructor's status in areas such as promotion, tenure, multiple-year contract, reappointment and/or termination.

For purposes of student evaluation, faculty members shall not be entitled to the identity of the student respondent unless such evaluation is utilized as a basis for dismissal or other disciplinary action.

<u>DIRECTIONS</u>: In completing this evaluation form, first fill in the three-digit instructor code number (as given by the proctor) in the space provided on the left. Then, using the scale listed below, indicate the number which seems in your honest opinion, most appropriate for this instructor. Place each answer in the space provided at the left.

### RATING SCALE

- 5 = Excellent Performance
- 4 = Very Good Performance
- 3 = Fair Performance
- 2 = Poor Performance
- 1 = Unsatisfactory Performance

Finally, please sign this form in the space at the bottom. Unsigned forms cannot be used. When you have completed this form and signed it, tear the signature section from the sheet along the dotted line and hand the two parts separately to the proctor. The proctor will make sure that one of the matching sequence numbers is on each part you hand in.



# MASSACHUSETTS COMMUNITY COLLEGE SYSTEM EVALUATION OF INSTRUCTION

Course Number:		Response Choices				
Instructor:		}				
Please Read First: The purpose of this form is to evaluate your instructor's performance. Please read each statement carefully and then indicate your rating by placing a check mark under the response you have chosen.		Very Good Performance (4)	Fair Performance (3)	Poor Performance (2)	Unsatisfactory Performance (1)	
1. How well did the course meet the published course description?						
2. How well were the instructional ob-						
jectives of the course explained?  3. To what extent were the instructional	<u> </u>					
Objectives accomplished?	ŀ					
4. How well was the course organized?						
5. How well prepared was the instructor?		<del>                                     </del>				
6. How effective was the instructor's presentation?						
7. How well do you think the instructor had a grasp of his/her subject matter and related fields?						
8. To what degree do you think the	[ [	<del> </del>	1			
method of instruction was appropriate to the course objectives?						
9. How well did the instructor respond to student questions?						
10. To what degree were students encour-			<u> </u>	<del>                                     </del>		
aged and given the opportunity to participate in class?						
11. How fair was the instructor's method		<del> </del>	<u> </u>			
of evaluation of student performance?	<u> </u>		1			
12. Did the instructor meet with and help you when requested? (answer if applicable)						
<pre>13. How effective overall was the assign- ed text as a learning aid? (if applicable)</pre>	1					
14. How effective overall were the supplementary course materials as learning aids? (if applicable)						
	<u> </u>	·	Date:	· · · · · · · · · · · · · · · · · · ·	·	
			Chuda			

ERIC

#### STUDENT EVALUATION SUMMARY

or:	Title:
d Semester:	Title:
Student Evaluation Sc	
Class and Section	Score
	, 1 <b>.</b>
	2.
	3.
	4.
_	5.
Score for all Classes	3:
or's Comments:	
Member's Comments (ii	f any):
Member's Comments (ii	f any):
y Member's Comments (i	f any):
y Member's Comments (in	f any):
y Member's Comments (ii	£ any):
y Member's Comments (ii	f any):
/ Member's Comments (if	
y Member's Comments (in	I have read and received a
y Member's Comments (in	
y Member's Comments (in	I have read and received a
	I have read and received a copy of this evaluation.



117

### COURSE/INSTRUCTIONAL MATERIALS EVALUATION FORM

Fac	ulty Member:
EVO	
νqν	et client:
DIV	rision:rand Semester:
160	it did salester:
1.	Are the materials submitted for course evaluation complete in accordance with Article XVII, Section 13.02B2 (see attached checklist)?
2.	Is the course outline consistent with the College's catalogue description of the course?
3.	Does the course outline evidence teaching procedures appropriate to the course?
4.	Are the instructional objectives appropriate to the course and to the requirements students must meet to complete the course?
5.	Are the procedures for evaluating students appropriate to the course?
<del>6.</del>	Evaluator's summary of course materials evaluation.



I have read and received a copy of this evaluation.

Evaluator

Faculty Member

Date:

Date:

If the faculty member wishes to respond to this evaluation, he/she must do so within seven (7) working days.



#### COURSE/INSTRUCTIONAL MATERIALS EVALUATION FORM

Eval	uator	Title:
Depar	rtmer	nt:
DIVI	510N	
Year	and	Semester:
		CHECKLIST FOR COURSE MATERIALS
	1.	Instructor's Name/Office Location
	2.	Course Title/Number
	3.	General course description (according to College catalogue)
	4.	All required texts and paperbacks, including information on publisher and edition used
	5.	Course topics and/or assignments and/or required and/or supplemental reading
	6.	Teaching procedures (briefly describe)
	7.	Instructional objectives (list)
	8.	Basis for student grading
	9.	Procedure (criteria) for evaluating student performance
	10.	Tentative Test Schedule/Assignment(s) Schedule
	11.	Attendance Policy

# PROCESS FOR CLASSROOM/INSTRUCTIONAL OBSERVATION

Instructor			<del></del>
Evaluator:		Title:	<del></del>
Campus:			
Department:			
Class to D	e observed:		
Course:		Pre-Conference (if	appropriate)
Date:			Time:
Time:		Post-Conference (i	f appropriate)
Room:		Date:	_ Time:
1. Relati	onship of class content	to instructional object	ives of course:
	riateness of instructor' ctional objectives:	s teaching methods to a	ttainment of the stated
3. Effect	iveness of the instructo	r's teaching methods:	
4. Instru	ctor's ability to develo	op and maintain appropri	ate student interest:
5. Instru	ector's ability to organi	ze and present course o	content and materials:
6. Instru	ector's ability to respon	nd to student questions:	;
	<del></del>		



7.	Special areas of performance n performance evaluation):	o', included above (not to be included in
8.	Evaluator's summary of instruc	tional performance:
Fac	ulty Member's Comments (if any)	:
		I have read and received a copy of this evaluation.
Eva	luator	Faculty Mamber
Dat	e:	Date:
If wit	the faculty member wishes to re hin seven (7) days.	spond to this evaluation, he/she must do so

122



()

# STUDENT ADVISEMENT LOG

Advisor:	Semester:	Year:
Evaluator:	Title:	Date Submitted:
Department/Program:	Division:	

tudent's Name	Program	Date of Conference	Recommendations/Purpose
		İ	
		i	
Į	•	1	

124

Evaluator's any):	Summary	of	Student	Advisement	or	Recruitment	Activities	(if
Unit Member	's Comme	nts	(if any	):				
			• • • •	•			-	
		_						
				I have evalua		ad and recei n.	ved a copy	of this
Evaluator				Unit N	1emb	er		<del></del>
Date:		_		Date:				
	member	wis	hes to r	respond to t		evaluation,		st do so



## COLLEGE SERVICE EVALUATION

Uni	t Memher:	
Dep	artment/Program:	
Eva Eva	luator:	Title:
1.	List the college service ac Section 13.02B4 and 13.03B2	ctivities assigned pursuant to Article XIII, 2.
_		Date (a) of Daylininghian
2.	Activities Completed	Date(s) of Participation (if applicable)
3.	Attach any documentation wi service activities set for	which evidences participation in the college th above (if requested).
I i	nereby certify that I have po	participated in the college service activities as
		Unit Member
		Date:
3		
IC.		126

Evaluator's Summary of College	e Service:
Unit Member's Comments (if any	<i>y</i> ):
	I have read and received a copy of this evaluation.
Evaluator	

If the faculty member wishes to respond to this evaluation, he/she must do so within seven (7) working days.



## FACULTY SUMMARY EVALUATION

Nint - I	
Division:	
Waluator:	Title:
raluste the overall re	formance of the instructor:
wattate the overall per	Tomaire of the histractor:
	•
professional leave, ten notified in writing con	unsatisfactory and ineligible for sabbatical leave, ure, promotion or performance-based awards shall be current with receipt of this evaluation.
professional leave, ten	ure, promotion or performance-based awards shall be current with receipt of this evaluation.
professional leave, ten notified in writing con	ure, promotion or performance-based awards shall be current with receipt of this evaluation.
professional leave, ten notified in writing con	ure, promotion or performance-based awards shall be current with receipt of this evaluation.
professional leave, ten notified in writing con	ure, promotion or performance-based awards shall be current with receipt of this evaluation.
professional leave, ten notified in writing con	ure, promotion or performance-based awards shall be current with receipt of this evaluation.
professional leave, ten notified in writing con	ure, promotion or performance-based awards shall be current with receipt of this evaluation.
professional leave, ten notified in writing con	ure, promotion or performance-based awards shall be current with receipt of this evaluation.
professional leave, ten notified in writing con	ts (if any):
professional leave, ten notified in writing con	I have read and received a copy of this
professional leave, ten notified in writing con	ts (if any):
professional leave, ten notified in writing con	I have read and received a copy of this
professional leave, ten notified in writing con Faculty Member's Commen	I have read and received a copy of this evaluation.
professional leave, ten notified in writing con	I have read and received a copy of this
professional leave, ten notified in writing con Faculty Member's Commen	I have read and received a copy of this evaluation.



Professional Staff Position Descr	ription
	Page of
Period Covered by this E7	July 1, 198 to June 30, 198
Due July 1)	
ob Title: Department/Work Area:	
format the E7 as follows (use ad	ditional pages as necessary):
Tob Description Item (Goal) [use	I, II, III,]
Objective(s) (if appropriate C,]	e and mutually agreed) Item(s) [use A, B,
Activities/Methods Ite	m(s) [use 1, 2, 3,]
President or Designee	Professional Staff Member
Date	Date



Professional Staff Work Perform	mance Evaluation
	Page of
Period Covered by this E8	July 1, 198 to June 30, 198
(Due: January 15)	
Professional Staff Member:	
Job Title:	
Department/Work Area:	
President or Designee:	
Format the E8 as follows (use	additional pages as nécessary):
Job Description Item (Goal),	
Objective(s) Item(s), or	
Activities/Methods Item(s)	
Number (refers to E7)	Evaluator's Comments



	Page of
eriod Covered by this E8	, 198 to, 198 July 1, 198 to June 30, 198
	July 1, 198_ to June 30, 198
June 1)	
aluator's Summary of Work P	erformance Evaluation:
•	
rofessional Staff Member's C	Comments (if any):
rofessional Staff Member's C	Comments (if any):
rofessional Staff Member's C	Comments (if any):
rofessional Staff Member's C	Comments (if any):
rofessional Staff Member's C	I have read and received a copy of this
rofessional Staff Member's C	
	I have read and received a copy of this evaluation.
rofessional Staff Member's C	I have read and received a copy of this



rofessional Staff Summary Eval	<u>Lua Elon</u>
	Page of
riod Covered by this E9	July 1, 198 to June 30, 198
ue: February 1	
cofessional Staff Member:	
spartment/Work Area: resident or Designee:	
valuation - Overall Performan	ce of the Professional Staff Member
	•
abbatical leave, professional performance—based awards shall the receipt of this evaluation	be notified in writing concurrent with
abbatical leave, professional performance-based awards shall	leave, tenure, promotion or be notified in writing concurrent with
abbatical leave, professional performance—based awards shall the receipt of this evaluation	leave, tenure, promotion or be notified in writing concurrent with
abbatical leave, professional erformance-based awards shall he receipt of this evaluation	leave, tenure, promotion or be notified in writing concurrent with
abbatical leave, professional performance—based awards shall the receipt of this evaluation	leave, tenure, promotion or be notified in writing concurrent with
abbatical leave, professional performance—based awards shall the receipt of this evaluation	leave, tenure, promotion or be notified in writing concurrent with
abbatical leave, professional performance-based awards shall the receipt of this evaluation Professional Staff Member's Co	I leave, tenure, promotion or the notified in writing concurrent with the notified in
abbatical leave, professional performance—based awards shall the receipt of this evaluation	I have read and received a copy of this evaluation.



Faculty/Professional Staff Promotion Application
Community College
Application for a Promotion for Academic Year
1. To be completed by Unit Member:
Name
Department/Work Area
Division
Date of Last Promotion
Step/Salary / /
Number of Pronotions Since Date of Hire
Rank at Time of Hire
2. To be completed by the immediate supervisor:
A. Eligibility
•
1. Candidate for promotion is:
<ul> <li>( ) eligible for promotion         (meets minimum time since last promotion,         evaluation requirements, and does not exceed         maximum number of promotions)</li> </ul>
( ) ineligible
2. Reason(s) for ineligibility:
B. Performance
C. Seniority
Divisional Rank for Promotion (up to the number of promotional opportunities)



Faculty/Professional Staff Promotion Application Cont'd Page 2

. Reasons for Rank (if request	ced):
Chairperson Signature	Date
3. To be completed by appropria	
Institutional Rank (up to the nu opportunities):	<pre>imber of promotional</pre>
Reasons(s) if Dean's rank revenunit member:	erses any divisional rank of an
Dean(s) Signature(s)	Date
4. To be completed by the Colle	age President:
Recommendation for promotion (	( ) yes ( ) no
Reasons if President's rank remember:	everses Dean's rank of any uni
President	Date
5. To be completed by the Board	of Trustees:
Promotion Decision: ( ) yes	( ) no
Chairperson	Date



#### Evaluation Form

Name	of	Department	Chair/Curriculum	Coordinator	Being	Evaluated:
------	----	------------	------------------	-------------	-------	------------

Date:	

<u>Directions</u>: The evaluations by unit members shall be used for the sole purpose of the recommendation to appoint/non-reappoint the Department Chair. The evaluation of the Department Chair/Program Coordinator shall include the evaluation by each unit member within the work area/program. If the question does not apply, write "not applicable."

- 1. Does the Department Chair (Program) assist in the recruitment and orientation of new instructional staff and if yes, how effectively?
- 2. Does the Department Chair (Program) advise in the instructional competency of all applicants for vacant positions after consultation with members of the program unit and if yes, how effectively?
- 3. Does the Department Chair (Program) submit the preferred subject matter preparation and class schedule of unit members within his/her program consistent with Article XXI and if yes, how effectively?
- Department the Chair (Program) supervise Does the in development of instructional materials and assist conducting of the research on the effectiveness instructional program consistent with the philosophy and objectives of the College, the requirements of external and/or regulatory agencies and Article VII and if yes, how effectively?

- 5. Does th€ Department Chair (Program) assist in the implementation of the evaluation process as stated in Article XIII relative to:
  - a. The process of evaluation of course materials and if yes, how effectively?
  - b. The process for classroom observation in a clinical or laboratory-like setting wherein the student is developing a variety of occupational and/or preprofessional skills fundamental to the students' performance after completion of designated program and if yes, how effectively?
- 6. Does the Department Chair (Program) encourage faculty to develop new methods of instruction and if yes, how effectively?
- 7. Does the Department Chair (Program) cooperate and facilitate cooperation with other program areas and/or departments, learning resources, student services and administrative services within the College and if yes, how effectively?
- 8. Does the Department Chair (Program) assist in the development, dissemination and implementation of Board/College policies, regulations and procedures which affect the department/program and if yes, how effectively?
- 9. Are liaisons with external agencies that are essential to the implementation of the program effective and if yes, how effective?



10.	meetings wh	nen needed	chair ( and if yes	Progra C	onvene depa tively?	artment
Facu]	ty Member's	Comments	(if any):			
Tear	Off					
Unit	Member will	sign befor	re returnir	ng to Divisi	on Chair.	
Evalu	ator:					
Date:						
Pleas	e sign this	form in the	he space at	the bottom	. Unsigned	forms

Please sign this form in the space at the bottom. Unsigned forms cannot be used. When you have completed this form and signed it, tear the signature section from the sheet along the dotted line and hand the two parts separately to the President or his/her designee who will make sure that one of the matching sequence numbers is on each part you hand in.

# Evaluation Form

Name of Department Chair/Work Area Being Evaluated:		
Date	·	
the the Area work	ctions: The evaluations by unit members shall be used for sole purpose of the recommendation to appoint/non-reappoint Department Chair. The evaluation of Department Chair/Work shall include the evaluation by each unit member within the area/program. If the question does not apply, write "not icable."	
1.	Does the Department Chair (Work Area) assist in the recruitment and orientation of new instructional staff and if yes, how effectively?	
2.	Does the Department Chair (Work Area) advise in the instructional competency of all applicants for vacant positions after consultation with members of the program unit and if yes, how effectively?	
3.	Does the Department Chair (Work Area) submit the preferred subject matter preparation and class schedule of unit members within his/her program consistent with Article XXI and if yes, how effectively?	
4.	Does the Lepartment Chair (Work Area) assist in the implementation of the evaluation process as stated in Article XIII relative to the process of evaluation of course materials and if yes, how effectively?	

Does the Department Chair (Work Area) assist in the development, dissemination and implementation of Board/College policies, regulations and procedures which affect the department/program and if yes, how effectively?
 Does the Department Chair (Work Area) convene department meetings when needed and if yes, how effective?

Faculty Member's Comments (if any):

Tear Off
Unit member will sign before returning to Division Chair.

Please sign this form in the space at the bottom. Unsigned forms cannot be used. When you have completed this form and signed it, tear the signature section from the sheet along the dotted line and hand the two parts separately to the President or his/her designee who will make sure that one of the matching sequence numbers is on each part you hand in.

Evaluator: \_\_\_\_\_

Date: \_\_\_\_



IAME					
OLLEGE					
IVISION	DEPARTMENT				
ATE					
ATURE OF REQUEST:					

AMOUNT REQUESTED:



ERIC Clearinghouse for Junior Colleges